



Health Education East of England

Does health coaching work?

A rapid review of empirical evidence

April 2014

Key themes

What is health coaching?

If people are confident in managing their health, they may feel better, be more satisfied with health services and have less need of services. This is important given the significant staffing and financial challenges facing the NHS and the burden of long-term conditions which accounts for about 70% of NHS spending. Health coaching has been proposed as a method to help reduce this burden by supporting people to take more responsibility for their health and wellbeing.

The term 'health coaching' refers to a diverse set of interventions, but all with a shared aim of helping people set goals and take action to improve their health or lifestyle. Health coaching has been defined as "a patient-centred process that... entails goal setting determined by the patient, encourages self-discovery in addition to content education, and incorporates mechanisms for developing accountability in health behaviours."¹

There are many types of health coaching. Health coaching may be facilitated by people with long-term conditions (peers), nurses, doctors, pharmacists, therapists or other allied health professionals. It can take place in person, by telephone, by email or online, or using a combination of approaches. It may be a one-off occurrence or occur regularly, such as every week or month for several months. Health coaching can be used to support people with long-term conditions or those who wish to make changes to their lifestyle. Table A illustrates the range of variables within health coaching initiatives. This is important because it shows that 'health coaching' is not a single, well-defined intervention.

Health coaching is popular in North America and Australia and there is an increasing interest in rolling out this approach in the UK. To be sure that UK initiatives are based upon the most up-todate evidence about whether health coaching is effective and who it may work best for, **this rapid review searched ten bibliographic databases for studies available as at March 2014. A total of 275 studies about health coaching were included, plus 67 studies about training professionals to support behaviour change more generally.**

The review aimed to support Health Education East of England make decisions about further embedding its programme in health coaching training. Health coaching has been piloted in the East of England since 2010. From April 2013 to October 2014, Health Education East of England has been building on this further, with a two-day training programme rolled out to almost 800 clinicians from 31 organisations, including nurses, allied health professionals and doctors. Twentyfour trainers completed a six-day accredited programme to train people in how to use health coaching skills when delivering usual care (rather than as a standalone intervention).

Focus	Attributes	Examples
Characteristics of the participant	<u>Who</u> : Who is the participant?	 Someone with a long-term physical condition Someone with a specific short-term condition or need (eg pregnancy) Someone with mental health issues Someone who may seek to improve healthy behaviours (diet, exercise, alcohol, smoking) An employee taking part in a general workplace health promotion programme
Characteristics of the coach	Who: Who is the coach?	 Peer Nurse Doctor Healthcare assistant Pharmacist Other
Characteristics of the health coaching	<u>What</u> : Is health coaching delivered alone or as part of a broader intervention?	Standalone interventionOne component of broader intervention
	<u>How</u> : How is health coaching delivered?	 Face-to-face Telephone Online / email / smartphone app
	<u>How many</u> : How many people are coached simultaneously?	One-to-one supportGroup support
	How long: What is the duration of health coaching support?	 One-off One month or less Less than six months Six months or longer
	<u>How many</u> : How many health coaching sessions are included?	 One session Two to five sessions Six to ten sessions 11 to 20 sessions 21+ sessions
	How often: How often are sessions run?	 Every week Every fortnight Every month Every two months, and so on

Table A: Taxonomy of health coaching characteristics based on research

Is health coaching effective?

There is promising evidence about health coaching, particularly for supporting behaviour change (see Table B). However, there are also a number of caveats to bear in mind when interpreting this evidence, as outlined on page five.

Table B: Impact of health coaching

Impact	Review	Random trial	Other study
Improved	0%	75%	92%
self-efficacy	of 1	of 16	of 26
	study	studies	studies
Improved	0%	59%	89%
self-care	of 6	of 36	of 37
behaviours	studies	studies	studies
Improved	33%	37%	84%
health	of 6	of 60	of 43
outcomes	studies	studies	studies
Reduced	25%	30%	70%
service use or	of 4	of 12	of 10
cost	studies	studies	studies

Improving attitudes and self-efficacy

One systematic review, 16 randomised trials and 26 other studies examined the impact of health coaching on people's attitudes and self-efficacy. 0% of the reviews, 75% of randomised trials and 92% of other studies found a benefit. In other words, **there is some evidence that health coaching can support people's motivation to self-manage** or to change their behaviours, and their confidence in their ability to do so.

Improving behaviour

Six systematic reviews, 36 randomised trials and 37 other studies examined the impact of health coaching on people's behaviours. 0% of reviews, 59% of trials and 89% of other studies found a benefit. In other words, **there is some evidence that health coaching can support people to adopt healthy behaviours** and lifestyle choices. Research has most commonly cited benefits in increasing physical activity, eating more healthily and reducing smoking.

Improving health outcomes

Six systematic reviews, 60 randomised trials and 43 other studies examined the impact of health coaching on health outcomes. 33% of reviews, 37% of trials and 84% of other studies found a benefit. In other words, **there is mixed evidence about the impact of health coaching on physical outcomes** such as cholesterol, blood pressure, blood sugar control and weight loss. This may be because it takes time to demonstrate changes in clinical indicators and many studies do not include long follow-up periods or large sample sizes.

Improving service use and costs

Four systematic reviews, 12 randomised trials and 10 other studies examined service use and costs. 25% of reviews, 33% of trials and 87% of other studies found a benefit. **There is insufficient evidence to conclude whether health coaching reduces healthcare use or costs.** Most studies are from outside the UK, making generalisation difficult.

Who does health coaching work for?

Targeting appropriately

One systematic review, nine randomised trials and 23 other studies commented about whether health coaching works better for some demographic groups than others. These studies did not tend to compare health coaching versus another initiative, but rather explored who gained most from a specific health coaching programme.

The evidence suggests that health coaching may be most effective for people who are highly motivated to change from the outset and who have the most severe conditions or unhealthy lifestyles. There is also some evidence that women may take to health coaching more readily than men.

Less advantaged groups

The review did not identify studies that sought to explore how to make health coaching more accessible or effective for less advantaged groups. There are examples of health coaching being used with people from lower socio-economic groups, minority ethnic groups and the homeless, but these studies did not explore how health coaching may need to be adapted for those groups.

The small amount of evidence available suggests that **health coaching may be just as effective with less advantaged groups**, but that a greater number of health coaching sessions may be needed before an impact is demonstrated.

Who makes the best coaches?

Different types of coaches

Two systematic reviews, six randomised trials and nine other studies examined different types of coaches. These studies suggest that people with long-term conditions who have received training in health coaching can be just as effective at health coaching as health professionals. Nurses, doctors and allied health professionals may be equally effective, though there are few direct comparisons made.

There are many other studies about health coaching provided by peers, nurses, doctors, pharmacists, allied health professionals and students, but these do not compare different types of coaches. **There is not enough evidence to suggest that one type of coach is more effective than others.** Many people may make good coaches, with appropriate training.

Training coaches

One randomised trial and 12 other studies examined approaches for providing training in health coaching skills and 67 additional studies examined training in behaviour change techniques, which may form a component of health coaching.

There is not enough evidence to conclude that one type of training is more effective than others. However, trends from the literature suggest that **training could usefully be at least two to five days in duration, allow for practice and observation, include role play and incorporate refreshers** and follow-ups to assess competency and motivate professionals to use their skills in practice.

Caveats with the evidence

Despite these positive trends, there are some caveats.

1. Variation in interventions

There has been no clear definition of what comprises health coaching and studies have used this or similar terms to represent widely varying interventions.

As described above, health coaching can take a variety of forms so it is difficult to say that health coaching as a concept works because some interventions may be several months long and some might be one-off, some might be delivered by peers and others by professionals, some might be telephone-based and others delivered face to face, and so on. The role that coaches take and the competencies they have can also vary markedly. In short, it is difficult to compare studies because the interventions included are vastly different, much like comparing apples and pears.

This means that where studies have found that health coaching does not work, it is important to consider the exact health coaching methods, providers, duration and frequency used rather than assuming the concept as a whole may be flawed.

2. Variations in quality of evidence

Furthermore, as illustrated in Table B, the quality of the evidence varies widely. The review has weighted the quality of evidence based on study design, whereby randomised trials and systematic reviews are potentially more robust than other studies. Many of the studies that have found positive outcomes from health coaching are not systematic reviews or randomised trials.

3. Lack of comparative evidence

Most studies do not compare health coaching with other alternatives. This means that even where studies suggest that health coaching has improved attitudes or behaviours over time, it is not possible to say whether health coaching has done this more quickly or effectively than usual care or other types of support.

Furthermore, there is little evidence about the cost-effectiveness of health coaching. Thus studies may have found positive benefits, but they do not explore at what cost or the opportunity costs involved.

4. Generalisability issues

Much of the evidence available about health coaching comes from a US context, where health systems, attitudes, commissioning and payments and are very different from the UK. Whilst studies from other countries can provide useful insights into potential trends, it is usually not possible to transfer interventions from one health ecosystem to another without adaptation.

Importantly, most of the evidence relates to health coaching set up as a separate intervention rather than used within routine practice, such as GP consultations so it may be difficult to extrapolate the findings to the effects of using health coaching within routine practice. It remains unclear whether health coaching from a clinician with whom people have an established relationship has greater benefits.

Planning for the future

Bearing in mind these caveats, teams in the UK are recognising the potential of health coaching to empower people, support healthier lifestyles and reduce the burden of long-term conditions for individuals and for health and social services. Health coaching is not a panacea and will not automatically lead to improvements in health outcomes or service use, but the evidence suggests 'top tips' to make the most of health coaching.

Top tips for selecting participants

- Health coaching appears to work best for people who are **motivated to change** and those who have the most severe disease. Health coaching also has a role in supporting a change in motivation, though this takes time.
- Health coaching may be just as effective for less advantaged groups, but more time may be needed to show an impact.

Top tips for selecting coaches

- Nurses, doctors, allied health professionals and peers can all provide health coaching if appropriately trained.
- Training of two to five days duration, with practical activities, role plays, observation and follow-up has been found to work well. A brief intervention is unlikely to disseminate health coaching skills well. More in-depth training may be required which focuses on shifting the mind-set of coaches towards being 'partners' helping to support change.

Top tips for providing health coaching

- There are many opportunities to embed health coaching as part of routine care. Health coaching can be provided in any care settings and in people's homes.
- Health coaching using face-to-face or telephone sessions may be more effective than relying solely on automated or electronic health coaching. There is no clear evidence that in-person health coaching is more effective than telephone health coaching.
- Health coaching programmes could combine a variety of methods, such as an initial meeting with a coach followed by monthly calls.
- There is no clear evidence about how frequently health coaching should occur but health coaching every one to two weeks for several months has often been found to work well to build relationships and support behaviour change.

Top tips for demonstrating impact

- Local and regional teams in the UK can add to the evidence base by building in robust evaluation which focuses on longer-term outcomes and health service use. Comparing those who receive and do not receive health coaching is essential.
- There is a need to gather and publish more robust information about the costs and cost-effectiveness of health coaching.

Contents

Part 1:	What is health coaching?	8
	Setting the scene What is health coaching?	9 12
Part 2:	Does health coaching work?	16
	Impact on attitudes and self-efficacy Impact on behaviour Impact on health status Impact service use and costs Impact of different types of health coaching	17 22 27 32 36
Part 3:	Who benefits most?	40
	Who benefits from health coaching? Targeting less advantaged groups	41 44
Part 4:	Who makes the best coaches?	46
	Different types of coaches Training coaches	47 50
Part 5:	Moving forward	58
	Implications for policy and practice	59
Annex 1:	Identifying evidence	64
Reference	25	66

This review was completed over a three-week period by The Evidence Centre for Health Education East of England (HEEOE). It complements the East of England Health Coaching Development Skills programme and evaluation, rolled out to almost 800 clinicians between April 2013 and October 2014. The review is based on studies principally relating to standalone health coaching services in the US whereas the approach in the East of England supports health coaching as part of a person's usual care. For more information about the Health Coaching Programme see: www.eoeleadership.nhs.uk/healthcoaching

Part 1: What is health coaching?

Setting the scene

The increasing impact of long-term conditions on health, quality of life and costs is well known. New approaches are required to improve patient activation, confidence, and healthy behaviours. Health coaching may be one of a suite of methods for enhancing patient selfmanagement.

From 2010, the East of England began piloting health coaching. From April 2013 to October 2014, Health Education East of England has embedded the initiative further, with a two-day training programme rolled out to almost 800 clinicians from 31 organisations, including nurses, allied health professionals and doctors. Twenty-four trainers completed a six-day accredited programme to train people in how to use health coaching skills when delivering usual care.

Health Education East of England is evaluating the impacts of the programme and compiling other evidence. To feed into this, this review describes empirical evidence about the impacts of health coaching.

Part 1 explores what health coaching is and why it may be important. Part 2 describes research about the impacts of health coaching. Part 3 explores whether health coaching works better for some people than others and Part 4 describes the variety of professionals using health coaching skills and how they can learn those skills. Part 5 summarises the implications of the empirical evidence for policy makers and practitioners.

The challenge

Engaging people in keeping themselves well is an essential component of reducing ill health and the demand for health services.² Supporting people to be active participants in their care has important impacts on satisfaction, the extent to which people adhere to treatment, relationships between patients and professionals and long-term health outcomes. ^{3,4,5,6,7,8,9,10} How people think and feel about their health impacts on what they do.^{11,12,13,14,15,16} If people feel more confident about managing their own health, they may have better health outcomes.^{17,18,19,20,21,22,23,24,25}

Research from around the world suggests that people want to be more involved in their care, ^{26,27,28,29} but often this does not happen in practice.^{30,31}

Patient activation is increasingly important given financial challenges and the need to make best use of limited staffing and service capacity.³² In the UK, health services are facing significant challenges. The population continues to grow in size and people are living longer, but not always in good health.^{33,34} More than 17 million people are living with one or more long-term conditions in Britain. These conditions have major impacts on people's health and wellbeing. The impacts are not only physical, such as reduced mobility, but also psychological, impacting on people's resilience and feelings of independence and control. This may increase people's need for support from healthcare and social services.

People with long-term conditions use the highest proportion of GP consultations and days in hospital. In fact, people with long-term conditions account 70% of the overall NHS spend.³⁵ The demand for health services outstrips what can be provided and staffing and financial resources are at a premium.

The number of people with three or more long-term conditions is predicted to rise by a million to 2.9 million by 2018 and it is estimated that by 2020 three quarters of all deaths will be from long-term conditions.³⁶

All of these factors have increased the focus on supporting people to be involved in decisions about their care and taking more responsibility for their wellbeing. The health White Paper *Equity and Excellence: Liberating the NHS* and the 2011 Health and Social Care Bill both emphasise strengthening people's voice.^{37,38} There is a move away from a paternalistic healthcare model where clinicians 'do things to' and make decisions for people towards helping people take more control of their health and wellbeing.

New approaches are required to:

- provide people with the knowledge, skills and confidence to self-manage and thrive;
- address the challenges associated with lifestyle change whilst reducing service use and cost;
- help all groups feel 'activated,' especially those with fewer skills and poorer quality of life who may require more intensive support;
- improve patient experience.

Part of the solution

Health coaching may be part of the solution to address these challenges. **'Health coaching' is an umbrella term used to describe many different interventions** that 'coach' or actively support people to self-care and a move away from a dependent model to one that is empowering and shared, based around a person's own aspirations and goals.

Health coaching has been used to support shared decision-making, to empower people with long-term conditions to selfmanage, to support health improvement and to optimise physical and mental health. It may be used as an adjunct to clinical skills in routine consultations, be offered in dedicated health coaching appointments or be part of wider wellness programmes.

Whilst health coaching is common in the US and Australia, it is somewhat novel in the UK. Some parts of the UK are testing the value of health coaching. For example, over the past year NHS Health Education East of England ran a series of two day health coaching training programmes for clinicians from acute trusts, primary care and community service providers and clinical commissioning groups. The aim is to provide clinicians with additional consultation skills and a mind-set that helps them transform the patient-clinician relationship and move from the role of 'expert' to 'enabler.' The programme is being evaluated through case studies with five organisations. In addition to the qualitative evaluation, NHS Health Education East of England wanted to compile empirical evidence about the potential value of health coaching and thus commissioned this rapid review.

Reviewing the evidence

The aims of the review were to compile readily available empirical evidence to:

- assess the impact of health coaching on outcomes for patients and the health service;
- examine whether some ways of delivering health coaching are more effective than others;
- understand which patients health coaching may be most effective for;
- consider how health coaching can be adapted for people who are less activated or from lower socioeconomic groups;
- understand whether some professionals are more able to adopt and use health coaching skills or whether some types of coaches are associated with better outcomes;
- examine any evidence of the impact of health coaching training for clinicians.

In other words, the focus was on exploring what, how and who questions: what are the impacts of health coaching, how can health coaching work best in practice, and who does health coaching work best and least for? Annex 1 describes the methods used to identify and synthesise the evidence. In brief, two reviewers independently searched ten bibliographic databases plus websites and reference lists to identify published or grey literature about the impacts of health coaching. Studies specifically labelling themselves as 'health coaching' or 'health coaching' were eligible for inclusion. The review was completed over a three-week timeframe.

More than 7,000 studies were screened and 275 studies were included in the review. Seven percent of these were systematic reviews, 40% were randomised controlled trials and 53% were studies of other designs (see Table 1).

In total 6% were from the UK, 18% were from other parts of Europe, 61% were from North America and 15% were from other countries.

Table 1: Studies included in the review

Country	Review	Random trial	Other studies	Total number
UK	5	4	7	16
Europe	3	25	22	50
North America	7	66	94	167
Other countries	3	14	25	42
Total	18	109	148	275

In addition, 67 studies were included about training clinicians to support behaviour change more broadly: five systematic reviews, 13 randomised trials and 49 other studies.

What is health coaching?

Towards a definition

Health coaching helps people consider their goals and how to make changes to improve their health.³⁹ From the 1990s onwards this approach gained popularity in North America, particularly as a way of supporting people with alcohol and substance use issues. Over the past ten years it has also gained momentum in many other parts of the world for supporting people with long-term conditions, lifestyle issues and behaviour change.⁴⁰

There is no one universally accepted definition of health coaching and there are many different models or

frameworks that can be thought of as health coaching or used within a health coaching approach. However, most conceptions of health coaching have some common characteristics, including:^{41,42}

- empowering people to take ownership of their health
- focusing on people's goals rather than what professionals want to achieve
- developing a collaborative relationship between the participant and coach
- assuming that people are resourceful and have potential
- helping people assess where they are and what they would like to achieve
- helping people plan how to achieve their goals in easy steps and do things they may have struggled to do in the past
- challenging habits and beliefs that inhibit people or are barriers to positive change

Health coaching has been described as "a goal-oriented, client-centred partnership that is health-focused and occurs through a process of client enlightenment and empowerment."⁴³

Health coaching differs from traditional approaches which tend to direct information 'at' people and ask people to do the things that health professionals instruct them to do. In the traditional model, professionals are seen as having expert knowledge and are tasked with imparting this to people and their families. In contrast, health coaching strives to help people and professionals work in partnership. People themselves are seen as having important knowledge and as being experts in their own wellbeing.⁴⁴ Using questioning and supportive techniques, health coaches help people talk about what they want to achieve, what is troubling them, what they want to change, what support they have to help make changes and what difficulties need to be addressed or minimised.45 The health coach's main role is not to teach, advise or counsel people but rather to support people to plan and reach their own goals.

A systematic review of 284 theoretical and empirical articles about health and wellbeing health coaching aimed to draw out the characteristics of health coaching in order to develop a robust definition. The reviewers identified an emerging consensus in what is referred to as health coaching, namely:⁴⁶

"a patient-centred process that is based upon behaviour change theory and is delivered by health professionals with diverse backgrounds. The actual health coaching process entails goal setting determined by the patient, encourages self-discovery in addition to content education, and incorporates mechanisms for developing accountability in health behaviours."

"The literature operationalised health and wellness health coaching as a process that is fully or partially patient-centred (86% of articles), included patient determined goals (71%), incorporated self-discovery and active learning processes (63%) (vs more passive receipt of advice), encouraged accountability for behaviours (86%), and provided some type of education to patients along with using health coaching processes (91%). Additionally, 78% of articles indicated that the health coaching occurs in the context of a consistent, ongoing relationship with a human coach who is trained in specific behaviour change, communication, and motivational skills."

Like sports coaching, health coaching aims to help people achieve their potential and goals, but health coaching is not focused on teaching a particular skill or technique. Rather, it encourages people to explore what is important to them and provides encouragement and support to take steps towards achieving those goals.

The term health coaching is often confused with or used interchangeably with methods such as motivational interviewing, counselling or consulting skills but it is not the same thing. Whilst motivational interviewing may be used as one technique within health coaching,⁴⁷ health coaching is a wider framework, mind-set or approach rather than a specific technique.⁴⁸ A key distinguishing feature of health coaching is how professionals interact with and view people as resourceful, empowered and activated partners in care. The professional does not take responsibility for advice and outcomes, only for managing the process. Health coaching is also different from other approaches because it focuses mainly on solutions rather than problems, and the solutions are defined by people themselves.

There are a wide range of studies about motivational interviewing and behavioural counselling, but these techniques are not reviewed here. Other reviews have found that these techniques can support behaviour change, particularly regarding alcohol and drug misuse.^{49,50,51,52} Some effect sizes may be modest.^{53,54} The impacts on health service use and costs are mixed.⁵⁵ These are amongst many techniques sometimes used within health coaching.

Taxonomy of characteristics

Health coaching can be delivered in many formats including in person, by telephone and online.^{56,57} It may be offered to individuals or groups.⁵⁸

Many different people can facilitate health coaching, including people with long-term conditions themselves, nurses, doctors, medical and nursing students, health educators, psychologists, physical therapists, pharmacists, health assistants, social workers and occupational therapists, amongst others.^{59,60}

The 275 studies included in this review were used to develop a basic taxonomy of the characteristics of health coaching to illustrate the widely varying forms that health coaching may take (see Table 2). This is not an exhaustive list of all the possible attributes of health coaching, but aims to show that there is no 'standard' manner of facilitating health coaching.

Focus	Attributes	Examples
Characteristics of the participant	<u>Who</u> : Who is the participant?	 Someone with a long-term physical condition Someone with a specific short-term condition or need (eg pregnancy) Someone with mental health issues Someone who may seek to improve healthy behaviours (diet, exercise, alcohol, smoking) An employee taking part in a general workplace health promotion programme
Characteristics of the coach	<u>Who</u> : Who is the coach?	 Peer Nurse Doctor Healthcare assistant Pharmacist Other
Characteristics of the health coaching	<u>What</u> : Is health coaching delivered alone or as part of a broader intervention?	Standalone interventionOne component of broader intervention
	<u>How</u> : How is health coaching delivered?	 Face-to-face Telephone Online / email / smartphone app
	<u>How many</u> : How many people are coached simultaneously?	One-to-one supportGroup support
	<u>How long</u> : What is the duration of health coaching support?	 One-off One month or less Less than six months Six months or longer
	<u>How many</u> : How many health coaching sessions are included?	 One session Two to five sessions Six to ten sessions 11 to 20 sessions 21+ sessions
	How often: How often are sessions run?	 Every week Every fortnight Every month Every two months, and so on

Table 2: Taxonomy of health coaching characteristics based on research

Part 2: Does health coaching work?

Impact on attitudes and self-efficacy

Available evidence

The review identified one systematic review, 16 randomised trials and 26 other studies focused on the effects of health coaching on attitudes, self-efficacy (confidence to self-manage or change behaviours) or satisfaction. Other studies may have included self-efficacy as a secondary outcome, but the studies in this section included self-efficacy or satisfaction as a major focus.

Patient activation, which combines someone's knowledge, skill, and confidence about managing their health and care is an important component of current policy. Tools such as the Patient Activation Measure (PAM) are being increasingly promoted to measure empowerment and activation.⁶¹ However this concept has not been well measured within empirical studies about health coaching. In fact, there are few studies that explore this concept in relation to health coaching.

Key trends

- There is some evidence that health coaching can support people's motivation to selfmanage or to change their behaviours, and their confidence in their ability to do so.
- 0% of the reviews, 75% of randomised trials and 92% of other studies suggested that health coaching may have a positive effect on people's attitudes towards changing their behaviour, their motivation or their self-confidence to manage their health (see Table 3).

Table 3: Direction of findings about attitudes and self-efficacy

Findings	% reviews	% trials	% other studies
Largely positive	0	75%	92%
Largely negative	0	19%	4%
Very mixed	100%	6%	4%
Total studies	1	16	26

Whilst the findings are very positive about the effects of health coaching on selfefficacy, it is important to note that most of the evidence is based on small, beforeand-after or other observational studies. This means it is difficult to state that it was the health coaching approach that made a difference (versus merely having regular contact with a professional, for example).

Furthermore, whilst it may be tempting to assume that the increased confidence to make change associated with health coaching (self-efficacy) will lead to behavioural changes or improved clinical outcomes, the evidence about this is mixed. There is little research directly related to health coaching showing a link between increased self-efficacy and improved clinical or service use outcomes. In the wider literature, a number of authors have extolled the importance of improving self-efficacy or empoerment, but there is very mixed empirical evidence about whether self-efficacy and activation leads to improved outcomes. A number of studies about topics other than health coaching suggest that improving selfefficacy is associated with other improved outcomes, but a large number of other studies suggest that the relationship is more complex or have not found a direct link. 62,63,64,,65,66,67,68,69,70

Examples

Impacts on satisfaction

The feasibility and acceptance of health coaching is generally reported as high.^{71,72} Some research has explored the impact of health coaching on people's satisfaction with services.⁷³ For example, 241 people with long-term conditions were randomly assigned to health coaching through an internet portal or use of the portal without health coaching. After three months, people taking part in health coaching were more satisfied with their healthcare and thought their clinician was more likely to give them useful advice.⁷⁴

Elsewhere in the US, an analysis of one years' worth of data about a nurse telephone health coaching initiative found that having nurse health coaches available 24 hours a day was associated with increased patient satisfaction.⁷⁵

Impacts on attitudes

Health coaching may affect people's attitudes towards the medicines they are taking, their expectations about whether they can recover and live healthy lives or their level of 'activation' or readiness to change.⁷⁶

A randomised trial in the US assigned 26 people with acute low back pain and low to moderate recovery expectations to usual physiotherapy alone or with the addition of five sessions of telephone health coaching by a physiotherapist trained in health coaching. After 12 weeks the health coaching group improved significantly more than the control group in terms of recovery expectation.⁷⁷

In the Netherlands, 168 people prescribed antidepressants were randomly assigned to three in-person health coaching sessions from a pharmacist plus a 25minute take-home video or usual care. Three months after the intervention ended, health coaching was associated with a better attitude towards their medication.⁷⁸

Not all studies find a change in attitudes, however, even if there are other positive impacts.⁷⁹

Impacts on self-efficacy

Studies from many parts of the world have suggested improvements in people's confidence in making changes following health coaching.^{80,81,82,83,84,85}

For instance, in Denmark 186 people with diabetes were randomly assigned to receive health coaching or education about oral health. Health coaching was associated with greater improvements in self-efficacy in tooth brushing. People with low or moderate self-efficacy at baseline were particularly likely to benefit.^{86,87}

In Japan, 24 people took part in 10 telehealth coaching sessions. Interviews with three coaches and nine patients plus analysis of logs kept by coaches suggested that health coaching helped patients to tell their own stories, encouraged them to experience and adopt fresh points of view and helped them to start working towards attainable goals.⁸⁸

However some trials have found that changes in self-efficacy and feelings of anxiety are mixed and may be shortterm.⁸⁹

Examples in primary care

Improvements in self-efficacy have been noted when GPs, nurses, medical assistants and non-clinicians use health coaching skills in primary care.⁹⁰ For instance, in the US 250 people with diabetes were provided with a literacyappropriate education guide and brief behaviour change health coaching which comprised a goal-setting meeting and nurse telephone follow-up at two and four weeks. At three-month follow-up, there were improvements in participants' knowledge, activation, self-efficacy, diabetes-related distress and self-reported behaviours.⁹¹

In the Netherlands, 133 people recently diagnosed with type 2 diabetes from 54 general practices were randomly assigned to usual care or three home visits by a peer (expert patient) who adhered to recommended treatment and lifestyle guidelines. The peer coach helped people set goals and these were evaluated during the next visit. Participants in the control group received care as usual. Health coaching was associated with improved self-efficacy, coping and saturated fat intake, especially for those with low selfefficacy at baseline.⁹²

Impacts for carers

Health coaching has also been found to improve the self-efficacy of people caring for someone with a long-term condition or other health issue.^{93,94,95} For instance, ten family caregivers of people with heart failure took part in three months of nurse telephone health coaching in the US. Health coaching was associated with reduced caregiver burden scores and increased self-reported confidence and preparedness for home management.⁹⁶

But not all evidence about self-efficacy is positive. Researchers in England provided 16 carers of people with eating disorders a dvd and telephone health coaching for a few months. Carers were satisfied with most aspects of the training and found dvds useful, but there were no recorded changes in confidence, distress or depression.⁹⁷

Health coaching may work best when one component of a wider support package. A randomised trial in the Netherlands assigned the carers of people with dementia to telephone health coaching alone, telephone health coaching plus respite day-care or day-care alone. Health coaching occurred every two to three weeks over a 20-week period. Caregivers who took part in health coaching in combination with respite care reported significantly less burden and more confidence compared to those who took part in health coaching only.⁹⁸

Types of health coaching

Studies of **health coaching provided by peers or novices** and those of health professional coaches have found equally favourable impacts on self-efficacy and satisfaction,⁹⁹ though studies did not compare different coaches directly.¹⁰⁰

For example, in the US layperson coaches worked with 120 parents of children with asthma. Having more coach contacts and more discussion was associated with improved confidence and readiness to adopt asthma management behaviours.¹⁰¹

There is also some evidence that virtual or electronic health coaching may have merit. A US study described a virtual coach implemented as a conversational character, providing health coaching via text-based and interactive dialogue with the participant. Health coaching was more effective in helping participants establish regular mindfulness meditation practice than self-administered training using written and audio materials. The coached group also felt more likely to change and had greater self-efficacy after health coaching.¹⁰²

Even a one-off health coaching session may improve short-term self-efficacy, though this does not usually last in the longer term. For example, a randomised trial in the US assigned 265 people with cancer and moderate or severe pain to usual care or tailored education and health coaching before a clinic visit. Immediately after the clinic visit, health coaching was associated with increased pain communication self-efficacy, though impacts did not last at six or 12-week follow-up.¹⁰³

Impact on behaviour

Available evidence

The review identified six systematic reviews, 36 randomised trials and 37 other studies focused on the effects of health coaching on people's behaviour.

Key trends

- There is some evidence that health coaching can support people to adopt healthy behaviours and lifestyle choices. Research has most commonly cited benefits in increasing physical activity, eating more healthily and reducing smoking.
- 0% of the reviews, 59% of randomised trials and 89% of other studies suggested that health coaching may have a positive effect on people's behaviour, including reducing the use of alcohol and tobacco, eating more fruit and vegetables and exercising more regularly.
- Whilst some of these studies are single case reports or small samples, randomised trials are also well represented, meaning we may be able to be more confident about the quality of the evidence (see Table 4).

Table 4: Direction of findings about behaviour change

Findings	% reviews	% trials	% other studies
Largely positive	0%	59%	89%
Largely negative	50%	25%	0%
Very mixed	50%	16%	11%
Total studies	6	32	37

Examples

Findings from systematic reviews

One of the systematic reviews explicitly examined health coaching and the others assessed a range of different interventions, of which health coaching was one.

A review of 15 studies about health coaching reported that 40% found significant improvements in nutrition, physical activity, weight management or medication adherence. Features of effective programmes included goal setting (73% of studies), motivational interviewing (27%) and collaboration with healthcare providers (20%).¹⁰⁴

The other reviews, which focused on interventions to improve diet or physical activity, found either insufficient evidence to draw conclusions,^{105,106} mixed evidence about the benefits of health coaching¹⁰⁷ or no benefits of health coaching on behaviour change.¹⁰⁸

For example, a review of 38 randomised trials involving 9,445 participants explored strategies to improve adherence to dietary advice. Thirty-two out of 123 (26%) diet adherence outcomes favoured the intervention group, however, studies investigating behaviour change health coaching reported no improvement in diet adherence.¹⁰⁹

Supporting behaviour change

Some randomised trials and other studies report that **health coaching can support people to change a wide range of behaviours**, including increasing physical activity,^{110,111,112,113,114} improving diet,^{115,116,117,118} improving lifestyle,¹¹⁹ reducing smoking,^{120,121,122} seeing health professionals more regularly or appropriately,¹²³ communicating with professionals¹²⁴ or family members,¹²⁵ medication adherence^{126,127} and undertaking other self-care behaviours.^{128,129,130,131,132,133,134}

The fact that the results of individual studies differ from combined systematic reviews may suggest that health coaching can have an impact on behaviour change, but is not necessarily more effective than other proactive interventions to support behaviour change.

For example, in the US 30 people who had a stroke were assigned to usual care or medication health coaching via telephone after hospital discharge. The coach discussed risk factors, reviewed medications and triaged people's questions to a stroke nurse or pharmacist. Initial calls lasted half an hour, with any follow-up calls to answer questions lasting about 10 minutes if needed. At three months after discharge, the coached group were more likely to have seen their primary care provider. The researchers suggested health coaching could improve appointment-keeping with primary care.¹³⁵ Elsewhere in the US, more than 3,500 smokers who called a helpline to seek assistance quitting were assigned to receive either self-help booklets by post or booklets and up to five sessions of telephone health coaching. Health coaching nearly doubled the odds of quitting and maintaining cessation for one year, at a cost of about US\$1,300 per person.¹³⁶

In France, 1,013 children and their parents were randomly assigned to dietary health coaching monthly by telephone for eight months or a control group given no support. Children and parents receiving health coaching were more likely than controls to achieve their nutritional targets for fat intake and to a lesser extent sugar and complex carbohydrate intake.¹³⁷

In Scotland, a non-randomised comparison investigated whether groupbased health coaching from peers improved breastfeeding initiation and duration. Health coaching was set up in four regions. Data were analysed for 598 women with live births in the nine months before the initiative began and 557 women during the nine-month intervention. Data from ten other regions were compared. Breastfeeding rates increased compared with baseline rates and compared to other regions.¹³⁸ However, not all findings are positive.¹³⁹ In the US, pregnant women who smoked were randomly assigned to receive structured advice from their doctor (who had received training in motivating behaviour change) and referral to individual health coaching or to receive brief advice to stop smoking and a quit smoking booklet (usual care). There were no differences between groups in smoking during pregnancy or one year after giving birth.¹⁴⁰

In Sweden, 228 people with early rheumatoid arthritis were randomly assigned to one-year of health coaching by a physical therapist or no intervention. All were regularly seen by rheumatologists and took part in rehabilitation. There was no difference between groups in the proportion undertaking healthy physical activity for 30 minutes or more each day.¹⁴¹

Some studies have found that health coaching impacts on some behaviours but not others or that the changes are not sustained once health coaching ends.142,143 For instance, a randomised trial in the US compared improved usual care alone versus adding six-sessions of health coaching during the three months after hospitalisation for acute coronary syndromes. Data from 525 people admitted to one of five hospitals were analysed after eight months. The health coaching group had higher self-reported physical activity during the first three months and this declined after the intervention was withdrawn. There were no differences between groups in smoking cessation, medication use, functional status or quality of life.¹⁴⁴

Impacts for carers

Research suggests that **health coaching carers** can also improve the achievement of self-selected goals.¹⁴⁵

However, not all studies with carers are universally positive. In the Netherlands, 97 parents of young adults with recent-onset schizophrenia were randomly assigned to usual care or training in motivational interviewing and interaction skills. At three-month follow-up, families in the intervention group had lower frequency and quantity of cannabis use, but there was no difference between groups regarding other substance use, general level of functioning, parental distress or sense of burden.¹⁴⁶

Types of health coaching

Studies of health coaching provided by peers or novices and those of health professional coaches have found equally favourable impacts on behaviour change, though studies generally did not compare different coaches directly.¹⁴⁷

For example, a randomised trial in China allocated 62 older people living in rural areas to usual care or health coaching from a volunteer. All participants received routine medication safety instructions about their long-term conditions. In addition, the health coaching group received reminders by trained volunteers, instructions with pictorial aids, three home visits and five telephone calls. Volunteer health coaching was associated with improved knowledge of medication safety and improved medication safety behaviours.¹⁴⁸ In the US, people who were successfully managing their diabetes were paired with people who were struggling with behaviour change associated with managing diabetes. The pairs were matched according to age, sex and physical appearance. Peer coaches initially met with participants face-to-face for one hour then talked by telephone once a week for 10 to 15 minutes for the next eight weeks. Participants reported that health coaching was personalised and useful in disease management. They reported making progress towards changing behaviour.¹⁴⁹

Most research has not directly compared whether some formats for delivering health coaching are more effective for supporting behaviour change than others. There are examples of positive findings in studies about face-to-face, telephone and online or virtual health coaching.^{150,151}

Health coaching may be useful when incorporated as part of a wider intervention.^{152,153} In England an online health coaching programme to encourage teeth brushing included role-modelling cartoons for children, a guide for parents on using rewards, a personalised plan with clear steps, tips to follow and a weekly 10minute review of progress. Children aged five to nine years from 44 families took part in a randomised trial. Over a threeweek period, children in the health coaching group brushed their teeth 38% more often than those in the control group.¹⁵⁴

Health coaching duration

Most studies do not directly compare the effect of health coaching for greater or fewer numbers of sessions. The findings of individual studies suggest that even a oneoff health coaching session can be useful in some instances, though a greater number of sessions may be more likely to sustain behaviour change or embed a wider range of changes.

For example, 473 young people aged between 12 and 20 years old being treated for an injury in a US hospital emergency department were randomly assigned to no intervention or a brief session of behaviour change health coaching to reduce risky behaviours associated with the injury. At three and six months, the intervention was associated with selfreported improvements in seatbelt use and bicycle helmet use. There was no impact on other target behaviours or the risk of re-injury requiring medical attention.¹⁵⁵ In Wales, 53 GPs and practice nurses from 27 general practices were trained in behaviour change motivational interviewing-based health coaching. 1,827 patients who screened positive for at least one risky behaviour were randomly assigned to receive a one-off health coaching session or usual care. Health coaching was associated with more reported attempts to change and selfreported improvements in healthy eating at three and 12 months and improved activity at 12 months, but there were no changes in clinical outcomes. Training cost £1,597 per practice.¹⁵⁶

Impact on health status

Available evidence

The review identified six systematic reviews, 60 randomised trials and 43 other studies focused on the effects of health coaching on people's health status and clinical outcomes.

Key trends

- There is mixed evidence about the impact of health coaching on physical outcomes such as cholesterol, blood pressure, blood sugar control and weight loss. This may be because it takes time to demonstrate changes in clinical indicators and many studies do not include long follow-up periods or large sample sizes.
- 33% of the reviews, 37% of randomised trials and 84% of other studies suggested that health coaching may have a positive effect on people's health status, including improving blood pressure, blood sugar control in people with diabetes, cholesterol and cardiovascular risk factors (see Table 5). Thus, the evidence about improved health status is very mixed, with similar numbers of trials finding improvements or no improvements.

Table 5: Direction of findings about health status and clinical indicators

Findings	% reviews	% trials	% other studies
Largely positive	33%	37%	84%
Largely negative	0%	38%	5%
Very mixed	67%	25%	11%
Total studies	6	60	43

27

Examples

Findings from systematic reviews

Six systematic reviews have examined the potential of health coaching. Three of these reviews specifically examined health coaching and the other three explored health coaching as one of many behavioural change interventions.

A review of 41 articles about 34 telephone health coaching initiatives to support people with long-term conditions found that 94% of studies reported positive outcomes from health coaching.¹⁵⁷

Another review of 30 studies about telephone health coaching for people with long-term conditions found that several studies reported improvements in health behaviour, self-efficacy and health status and satisfaction. There was less evidence about improvements in quality of life and patient satisfaction with the service.¹⁵⁸

A review of 50 studies of face-to-face communication-related behaviour change techniques provided in primary care by doctors or nurses found favourable health outcomes in 56% of studies. The most successful techniques were behavioural counselling / health coaching, motivational interviewing, education and advice.¹⁵⁹ A review of 17 randomised trials exploring broad strategies to increase physical activity in people with type 2 diabetes found that behavioural interventions were associated with increased blood sugar control. Interventions underpinned by behaviour change theories and those using a large variety of behaviour change techniques were most effective.¹⁶⁰

Another review of 17 studies about behaviour change strategies for preventing and managing childhood obesity found that the six most effective strategies were providing information about the consequences of behaviour to the individual, environmental restructuring, prompting changes in behaviour, role modelling / advocate, stress management / emotional control training and general communication skills training.¹⁶¹

These broader reviews suggest that some of the components common within health coaching have been associated with positive health outcomes for children and adults. However, a review of five studies about whether life health coaching could influence health outcomes found mixed results.¹⁶²

Impact on physical outcomes

A number of individual studies have reported improved clinical indicators^{163,164,165,166} such as better blood sugar control for people with diabetes,^{167,168,169,170,,171,172,173,174} improved blood pressure,^{175,176,177,178} reduced cholesterol,^{179,180,181,182} reduced cardiovascular or lifestyle risks,^{183,184,185} reduced pain¹⁸⁶ and weight loss or reduced body mass index.^{187,188,189,190,191,192,193,194,195,196,197,198,199, ²⁰⁰ Although some of these studies objectively measure outcomes, others rely on self-report.²⁰¹ It is also true that some of the improvements are modest.²⁰²}

In the US, 56 people with diabetes were allocated to six months of health coaching (14 30-minute sessions) or usual care. Health coaching was associated with increased patient activation, perceived social support, self-reported adherence, exercise frequency and perceived health status and reduced stress. Those with poorly controlled blood sugar had improved blood sugar control.²⁰³

Another trial in the US examined whether health coaching children aged eight to 13 vears reduced their second-hand smoke exposure. Two hundred and one families were assigned to eight in-home health coaching sessions or no intervention. During health coaching sessions, children were shown charts of cotinine assay results and provided with praise and incentives for reductions. At the end of the programme, health coaching was associated with a greater decrease in urine cotinine concentration and child reported exposure to cigarette smoke per day. However, this was not sustained at 12month follow-up.²⁰⁴

Elsewhere in the US, 525 people with longterm conditions at high-risk were randomly assigned to usual care or aggressive pharmacological management plus behavioural health coaching from a nurse practitioner or community health worker. At 12 months, the intervention group had greater improvements in cholesterol, blood pressure, blood sugar control and perceptions of the quality of care.²⁰⁵

But **not all findings are positive**. Some trials and other studies have either found no evidence of improvements in health status or clinical indicators^{206,207,208,209,210,211,212,213,214,215,216,217} or mixed findings, with improvements in some things but not others.^{218,219,220,221,222,223,224}

For example, in Japan 134 people with diabetes were randomly assigned to oneto-one monthly nurse health coaching by telephone or usual care. There were no significant differences in blood sugar control, body mass index (BMI), blood pressure, cholesterol or health related quality of life.²²⁵

In Australia, 151 overweight and obese 13 to 16 year olds were randomly assigned to a community-based weight management programme alone or with additional telephone health coaching or text messages or email communication once per fortnight. The weight management programme involved young people and parents separately attending seven weekly group sessions, followed by quarterly sessions for young people. At two-year follow-up, health coaching was not associated with any additional improvements in outcomes.²²⁶ In another part of Australia, 473 people with diabetes from 59 primary care practices were randomly assigned to receive health coaching from a practice nurse or usual care. Nurses received two days of training in telephone health coaching and aimed to provide eight telephone and one face-to-face session per patient. The median number of health coaching sessions was three per person. At 18-month follow-up there was no difference between groups in blood sugar control or other outcomes.²²⁷

In Canada, 45 university students with a body mass index greater than 30 were assigned to work with a coach to achieve personal goals or to learn from a specialist who provided scripted, education-based lessons about lifestyle, exercise, attitudes, relationships and nutrition. Follow-up occurred immediately after the programme and at three and six months. Structured education was more effective than health coaching for reducing weight, but health coaching resulted in greater decreases in calorie intake. The health coaching group said that selfunderstanding and self-responsibility were primary outcomes of their experience whereas the education group focused on the value of the practical knowledge gained.228

Impact on quality of life

Health coaching may impact on healthrelated quality of life.^{229,230} In Korea, 48 women who survived breast cancer were randomly assigned to receive usual care or face-to-face education, telephone health coaching and small group meetings. The intervention was associated with better reported quality of life and emotional wellbeing and lower psychological symptom distress.²³¹

On the other hand, in the US 17 people with chronic obstructive pulmonary disease (COPD) were randomly assigned to a mobile phone health coaching intervention or a mobile phone selfmonitoring intervention. All participants met with a nurse to develop an individualised exercise plan, were issued a pedometer and exercise booklet and asked to log their daily exercise and symptoms. The health coaching group also received weekly reinforcement text messages and reports of worsening symptoms were automatically flagged for follow-up. After six months, there were no differences between groups in functioning or health-related quality of life.²³² Other studies have also failed to find an improvement in quality of life following health coaching.^{233,234}

Impact on mental health outcomes

Research has found some benefits of health coaching for improving perceived mental health^{235,236} or reducing depression and anxiety,²³⁷ especially amongst people with moderate or severe symptoms initially. ^{238,239}

For example, 30 cancer survivors took part in six wellness health coaching sessions in the US. During the three-month intervention, there were improvements in depression, anxiety and quality of life. The extent of improvements decreased over 12 months of follow-up, but remained higher than baseline. The perceived most helpful aspect of working with a coach was motivation and feedback.²⁴⁰

In another part of the US, 111 older people were randomly assigned motivational interviewing-based health coaching or usual care. The health coaching group chose the health behaviours they wanted to change and took part in one in-person session followed by nurse telephone calls and email contact for six months. Health coaching was associated with less illness intrusiveness and health distress at six months but it was uncertain whether these mental health outcomes resulted from behaviour changes.²⁴¹

However, not all studies exploring the impact of health coaching on mental health have found positive changes.^{242,243,244}

Impact for carers

There are also a small number of studies available about the impact of health coaching on the **health outcomes of carers**. For example, in Australia 22 relatives of colorectal cancer survivors took part in six telephone health coaching sessions and received a handbook and pedometer. A before-and-after study reported that at six weeks there were improvements in physical activity, diet, alcohol intake, body mass index, waist circumference and physical and mental health-related quality of life. Retention was 100%.²⁴⁵

In Israel, 169 parents of children with learning disabilities received either individual health coaching, group counselling or no intervention. Those receiving health coaching or group sessions had more favourable outcomes then the comparison group. Group counselling was more effective than individual health coaching for reducing stress. Bonding was the most consistent predictor of outcomes.²⁴⁶

Impact on service use and costs

Available evidence

The review identified four systematic reviews, 12 randomised trials and 10 other studies focused on the effects of health coaching on health service use and costs.

Key trends

- There is insufficient evidence to conclude whether health coaching reduces healthcare use or costs. Most studies are from outside the UK, making generalisation difficult.
- 25% of the reviews, 30% of randomised trials and 70% of other studies suggested that health coaching may help to reduce the use of health services or be cost-effective (see Table 6).

Table 6: Direction of findings about service use and costs

Findings	% reviews	% trials	% other studies
Largely positive	25%	33%	70%
Largely negative	50%	67%	20%
Very mixed	25%	0%	10%
Total studies	4	12	10

Examples

Findings from systematic reviews

Systematic reviews that included health coaching alongside other behaviour change interventions suggest that there is insufficient evidence to draw conclusions about cost-benefits or that health coaching is not associated with cost savings. However other, more specific, systematic reviews have suggested some potential. For example, a review of 98 randomised trials of patient decision aids found that 11% included health coaching. Compared to usual care, health coaching provided alongside a decision aid improved knowledge and decreased average costs. Impacts on decisionmaking were variable.²⁴⁷

A review of 24 randomised trials about lay health advisors found little evidence of effectiveness regarding improving diet or exercise. Incremental cost effectiveness ratios were £6,000 for smoking cessation; £14,000 for telephone-based diabetes management, and £250,000 or more for promoting mammography attendance and HIV prevention amongst drug users. There was insufficient evidence to estimate costeffectiveness for breastfeeding or mental health promotion.²⁴⁸

Potential cost savings

The quality of studies exploring the costs or cost-effectiveness of health coaching is widely variable, but some randomised trials are available.

For instance, 174,120 people in the US were assigned to a usual-support group or an enhanced-support group. Both groups were provided with usual care or health coaching based on whether they had high predicted medical costs. The same telephone intervention was delivered to the two groups, but more people in the enhanced-support group were eligible for health coaching due to lowering cut-off points for predicted future costs and expanding the number of qualifying health conditions. After 12 months, 10% of the enhanced-support group and 4% of the usual-support group received health coaching. Average monthly medical and pharmacy costs per person in the enhanced-support group were about 4% lower than those in the usual-support group, largely due to reduced hospital admissions. The cost of the intervention was less than US\$2 per person per month.²⁴⁹

Another randomised trial in the US examined health coaching to support shared decision making in people with back pain or joint pain. 24,167 adults were assigned to usual care or to automated voice prompts that allowed people to connect to a health coach. The automated system led to 11 times more use of health coaches. There was no difference in the rate of surgery, but a trend towards reductions in costs.²⁵⁰

Non-randomised studies have also suggested a trend towards reduced costs from health coaching.^{251,252,253}

Worksite wellness programmes

Many studies about costs focus on worksite wellness programmes or large health promotion programmes run by health maintenance organisations.

In the US, analysis of a worksite health programme that included health coaching estimated healthcare expenses per person per year to be US\$176 lower for participants. Inpatient expenses were lower by US\$182 per person. There was a return on investment of US\$1.65 for every dollar spent on the programme.²⁵⁴

Another US worksite health improvement initiative included health risk assessments, health education and health coaching. Analysis of four years' worth of data suggested improvements in smoking status, dietary fat and fibre intake, exercise, mental health, readiness to change, perceptions of overall health and health risks. There were no changes in body mass index. Return on investment, including incentive costs and fees, was 2.87:1.²⁵⁵

Another US programme for 1,282 people included consultations with wellness coaches and personal trainers, a 24-hour company fitness centre, personalised health assessments and preventive screening. People taking part reported improvements in physical activity, healthy food consumption, weight loss and blood pressure. Absenteeism decreased by approximately 25% and the overall return on the investment was calculated to be 2.43.²⁵⁶ In Lebanon, six rural employers were divided into three groups: a control group and two worksite wellness interventions with varying activities. Over a four year period, the health coaching and referral group had the greatest degree of participant engagement and greater improvement in health and wellness indicators compared to the control and lower-intensity intervention groups. However, the lower-intensity educationfocused intervention had more favourable cost-effectiveness ratios.²⁵⁷

Changes in service use

A small number of studies have found reductions in healthcare service use. This is not necessarily a good indicator of the effectiveness of health coaching because by empowering people health coaching may actually increase the use of services in the short-term if people attend regular checks or seek help when they notice exacerbations. Nevertheless, some studies have found positive outcomes in this regard.

For example, in the US people aged 65 years or older admitted to hospital were assigned to receive usual care or encouragement from a 'transition coach' to take a more active role in their care and to assert their preferences. The intervention was associated with fewer readmissions at 30, 90 and 180 days.²⁵⁸ A similar study by the same research group found reduced readmission rates a 30 and 90 days, but not 180 days. At 180 days, average hospital costs were lower for the intervention group (US\$2,058 versus \$2,546).²⁵⁹

Non-randomised studies have also suggested some reductions in hospital admissions.²⁶⁰

Conflicting findings

Not all cost analyses are in favour of health coaching. For example, in the UK 2,698 people with heart failure, coronary heart disease, diabetes or chronic obstructive pulmonary disease and a history of inpatient or outpatient hospital use were invited to take part in telephone health coaching and were matched oneto-one with a comparison group. Health coaching involved monthly telephone support from a nurse and a personalised care plan over a median duration of 26 months. Emergency admissions and outpatient attendances increased more quickly among those receiving health coaching than matched controls. Health coaching was associated with an increase in secondary care costs of £175 per person.²⁶¹

Another UK study randomly assigned 208 people with ocular hypertension or glaucoma to behaviour change health coaching or usual care. There was no difference between groups in adherence to therapy or intraocular pressure reduction. The average cost per patient was £10.35, but the intervention was not cost-effective.²⁶²

In Australia, 430 adults who had suffered a heart attack were randomised to telephone health coaching or usual care for six months. At six and 12-month follow-up, there were no changes in improved health status between groups. People in the health coaching group were more likely to be admitted to hospital due to causes unrelated to cardiovascular disease. The overall cost for the health coaching group was higher (US\$10,574 versus \$8,534 usual care), mainly due to greater hospitalisation costs. The incremental cost-effectiveness ratio was US\$85,423 per QALY.²⁶³ In the Netherlands, 730 older workers were randomly assigned to exercise, health coaching and fruit or no intervention. There was no difference between groups in costs and benefits. Per point improvement, the incremental costeffectiveness ratio was 280 Euro for general vitality and 7506 Euro for workrelated vitality. 2.21 Euro was lost per Euro invested.²⁶⁴

Elsewhere in the Netherlands, people with heart failure were randomised to receive basic support by a specialist nurse, intensive health coaching support by a nurse or routine follow-up by a cardiologist. Basic nurse support was best in terms of costs per life year gained and costs per quality-adjusted life year gained. There was no difference in the incremental cost-effectiveness ratio between intensive health coaching and basic support (532,762 Euro per life-year). The researchers concluded that basic support may be useful for those with mild to moderate heart failure, with more intensive health coaching reserved for those with severe disease.²⁶⁵

Another trial in the Netherlands randomised 88 people newly prescribed an antidepressant to education and health coaching by a pharmacist or usual care. Health coaching comprised three contacts with the pharmacist, with an average duration of between 13 and 20 minutes, and a take-home video. There were no differences between groups in medication adherence, improvements in depression scores or costs. The incremental costeffectiveness ratio associated with pharmacist health coaching was 149 Euro per 1% improvement in adherence and 2,550 Euro per point improvement in the depression average item score. The intervention was not likely to be costeffective compared with usual care.²⁶⁶

In the US, 1,755 overweight people were randomly assigned to i) written materials and access to a basic online weight management programme, ii) materials plus an interactive website or iii) materials, interactive website and brief telephone and e-mail health coaching. All groups had weight loss, increased physical activity and reduced blood pressure which was maintained after 18-months. The costeffectiveness ratio was US\$900 to \$1100 per quality-adjusted life year (QALY) for the basic and interactive websites and US\$1900 per QALY for health coaching. It would take three years to recover costs for the basic and interactive website and six years for the health coaching intervention.²⁶⁷

Similarly, a number of studies have found that health coaching does not impact on healthcare service use. For example, in the US 241 parents of children aged between two and ten years were randomly assigned to 18-months of health coaching or usual care. More children of coached parents had at least one asthma monitoring visit after enrolment, but the proportion with at least four monitoring visits in two years was low. There were no differences between groups in emergency department visits or admissions.²⁶⁸

A non-random comparison in the US invited people with long-term conditions who had two admissions to hospital or visits to the emergency department over the past twelve months to take part in health coaching. At two-year follow-up, there were no differences in admissions, primary care visits or healthcare costs compared to a matched group.²⁶⁹
Impact of different types of health coaching

Available evidence

The review identified four reviews, 14 randomised trials and 26 other studies describing or comparing different types of health coaching styles or formats.

Key trends

- Health coaching may take many different forms.^{270,271} It is difficult to suggest that one form is more effective than others because studies do not tend to compare.
- Face-to-face and telephone approaches may be more interactive than online or virtual health coaching, but it would be premature to conclude that this is associated with better outcomes. Both one-toone and group sessions have been found to have benefits.
- Health coaching can be provided as part of routine consultations or as a standalone initiative. Most research considers health coaching as a specific initiative rather than exploring incorporating health coaching skills into usual consultations.
- One clear message is that the more health coaching sessions people take part in, the more likely they may be to change.

Examples

Delivery style

As described in the taxonomy in Part 1, health coaching can be delivered in a range of ways including in-person, by telephone, online or as part of automated tools.^{272,273,274,275,276,277,278,279,280}

People may sometimes prefer seeing a coach face-to-face²⁸¹ but there is not strong evidence that in-person health coaching is more effective than telephone calls. This may be due to a lack of studies directly comparing these approaches.

A randomised trial in the US included 415 obese people with at least one cardiovascular risk factor. One group received usual care, one group received health coaching support by telephone, a website and e-mail and the final group received the same remote support plus inperson contact during group and individual sessions. Health coaching was associated with weight loss, but there was no difference between remote or inperson approaches.²⁸²

Another US study compared combining telephone and face-to-face health coaching versus telephone health coaching alone for exercise performance in a home-based low-intensity programme for high-risk ethnically diverse older people. The combined group had better exercise performance.²⁸³ Another study found that those taking part in telephone health coaching were more likely to be retained within the programme than those taking part in faceto-face sessions.²⁸⁴

A randomised trial in the Netherlands assigned 1,386 overweight people to sixmonths of telephone health coaching, sixmonths of email health coaching or selfdirected materials (control). After two years, there was no difference between groups in weight loss.²⁸⁵

In the US, seven people with COPD took part in an online meeting (webinar) to coach them in how to engage in conversations about end-of-life care with their clinicians. After three months, all participants had taken further action about end-of-life planning. Five (71%) thought webinars were acceptable option if face-to-face contact was not possible but all said that adding a video stream would have promoted more interaction. Although this intervention was labelled as health coaching, it focused mainly on education.²⁸⁶

People from specific demographic groups may prefer some styles of health coaching more than others. In the US 6,055 people from 10 large employers self-selected into postal or telephone health coaching. Those who selected telephone health coaching were more likely to be older, female, salaried and more ready, confident and motivated to make a behaviour change compared to those in the postal programme. Both strategies helped to reduce health risk status, with the telephone approach being slightly more effective.²⁸⁷

Delivery context: individual or group

In addition to one-to-one health coaching, there are many studies about group health coaching.²⁸⁸ For instance, in Canada a randomised trial assigned 680 people at high risk or with coronary heart disease to either risk factor feedback, brief advice and handouts (control) or health coaching which involved the same advice and handouts plus six weekly one-hour teleconferenced sessions with groups of four to eight people. At six-month followup, group health coaching was associated with improved exercise, diet, blood pressure, cholesterol and 10-year absolute risk of cardiovascular events.²⁸⁹

Evidence about any differences in the effectiveness of individual versus groupbased health coaching is lacking. Most studies have not found differences between group or individual approaches. For instance, a case-control study in Canada compared health coaching delivered in groups versus one-to-one for young people with disabilities. Health coaching improved goal attainment. There was no difference between group and one-to-one formats.²⁹⁰

However, some studies suggest that group health coaching may have additional benefits. A US trial randomised 34 women from rural areas to a behavioural weight loss health coaching programme delivered one-to-one or in a group. Group health coaching was more effective and more cost-effective.²⁹¹ A non-random comparison from Israel found that group health coaching was associated with better stress reduction in carers, largely due to being able to bond with others.²⁹² The importance of bonding was also mentioned in a Scottish study examining breastfeeding health coaching.²⁹³

Content of health coaching sessions

Most empirical studies do not describe in detail the content and topics covered within health coaching sessions, but it is possible to make some general statements. A systematic review of 284 empirical and theoretical articles examined the components of health coaching. The reviewers found that in the published peer reviewed literature, health coaching incorporates a process that is fully or partially patient-centred (86% of articles), includes patient-determined goals (71%), incorporates self-discovery and active learning processes (63%), encourages accountability for behaviours (86%), and provides education alongside health coaching (91%).294

There is insufficient evidence to be able to state that the inclusion of some content within health coaching sessions leads to better outcomes, but some studies suggest that **health coaching which is integrated into routine care and signposts people to other community resources is well received**.^{295,296,297} Using reflective and change-inducing questions has been found to spark conversations about change, though these may not be well-used in routine practice, even by practitioners who have been trained in health coaching or motivational skills.^{298,299}

Psychological support and building a strong trusting relationship may be an important component.^{300,301} Interviews with 32 people who took part in health coaching in the US reported that health coaching involved a caring relationship, leading to greater patient investment in making changes.³⁰² Other interviews in Norway found that if people had a good relationship with their GP, this created motivation to change and meant health coaching was interpreted as 'good care.'³⁰³

A randomised trial in the US examined whether sequential stage of change-based health coaching to stop smoking, reduce dietary sodium level and increase physical activity would be more effective than simultaneous counselling in primary care. Two hundred and thirty African Americans with hypertension were assigned to one in-clinic health coaching session on all three behaviours every six months, supplemented by motivational interviewing by telephone for 18 months; or a similar protocol that addressed a new behaviour every six months; or a single referral to existing group classes (usual care). At 18 months, there were no differences between groups.³⁰⁴

Duration of health coaching

As with other aspects of health coaching there is little comparative evidence about the benefits of greater or fewer numbers of health coaching sessions.

In one of the only randomised trials available about this topic, 50 people in the US were allocated to receive nutrition health coaching about eating behaviours at baseline only (control group) or an additional eight times over a six-month period. Multiple health coaching contacts were associated with a 31% decrease in the intake of energy, fat and carbohydrate compared to 20% in the control group. Multiple health coaching sessions were associated with weight loss but one-off health coaching was not.³⁰⁵

Some non-randomised studies also suggest that the more health coaching sessions people take part in, the more likely they are to achieve their goals or improve health outcomes.^{306,307,308,309}

Combining with other interventions

Some research suggests that health coaching may be feasible and effective when combined with other support.³¹⁰

For example, in Belgium a 15-minute selfefficacy health coaching session was added to the start of a 12-week physical activity programme. The researchers randomly assigned 227 people to the usual programme or to additional selfefficacy health coaching. After the physical activity programme both groups had improved fitness and reduced body mass index. The change was more marked in those who took part in health coaching. ³¹¹

A randomised trial in the US assigned 320 young people with type 1 diabetes transitioning to adolescence to one of two online interventions: a coping skills health coaching programme or diabetes education. Neither intervention improved blood sugar control. Young people were invited to cross over and try the other intervention after one year. After 18 months, youth who completed both interventions had better blood sugar control, quality of life and self-efficacy and lower perceived stress compared with those who completed only one programme.³¹² Another randomised trial in the US assigned 120 overweight or obese adults to one of four groups: a weight-loss supplement alone, a placebo, health coaching alone or the supplement plus 11 30-minute health coaching sessions for 17 weeks. The initial health coaching conversation lasted 60 to 90 minutes. The combined intervention was associated with greater reductions in weight than the supplement, health coaching or usual care alone.³¹³

A before-and-after study in the US analysed data from 12,984 people taking part in a worksite health programme. The programme comprised a posted health risk assessment and a free nurse helpline. A pilot sub-programme added screening, resources, and for high risk participants, health coaching and vouchers for medical office visits. Those eligible for health coaching and vouchers had a greater reduction in health risks.³¹⁴ Part 3: Who benefits most?

Who benefits from health coaching?

Available evidence

The review identified one systematic review, nine randomised trials and 23 other studies focused on the participants in health coaching relationships (patients). Most of the other studies included in the review also included details about participants but these studies were particularly focused on participant characteristics or whether health coaching may be more effective for some groups than others.

Key trends

Studies have found that health coaching can be effective in younger and older people, in men and women and in people from a wide range of ethnic groups.³¹⁵ There is no evidence that health coaching is more effective for people with some conditions than others.

Studies tend to undertake post-hoc analyses to explore whether health coaching is more effective for some groups than others. These analyses may be based on small samples with other methodological weaknesses, so the findings should be treated with caution. However, the overall trends are that health coaching has been found to work best for:

- people with low levels of self-efficacy prior to health coaching^{316,317}
- people with high readiness / motivation to change^{318,319,320,321,322,323}
- people at highest risk^{324,325}
- people with the most severe symptoms^{326,327}
- people with low levels of selfmanagement or medication adherence³²⁸
- women^{329,330,331}
- young people (for telephone or virtual health coaching)³³² or much older people³³³
- minority ethnic groups³³⁴
- vulnerable groups³³⁵
- people with the lowest levels of education³³⁶

Examples

People most in need of change

Many studies have identified that **the people who appear to benefit most from health coaching are those who have the most to improve**, such as the greatest amount of weight to lose, the most severe symptoms or the most poorly controlled conditions.³³⁷ For example, a random trial in Sweden assigned 228 people with rheumatoid arthritis to usual care or one year of health coaching to support increased physical activity. Health coaching was most effective for those who were more severely affected by their disease at the outset.³³⁸

An analysis of data about 6,129 people enrolled in a US health coaching programme found that people who made positive changes as a result of health coaching had the highest body mass index, lowest levels of exercise and the poorest overall health at baseline. These people were also most likely to continue with health coaching through to 12 months, whereas there was an overall attrition rate of 36%.³³⁹

Despite people most in need of change sometimes benefitting the most from health coaching, some research suggests that these people may not always be targeted for health coaching or that clinicians may not use health coaching skills with them in routine consultations.³⁴⁰

Demographics

Although some studies suggest that women and people from minority ethnic groups may be more likely to take part in health coaching if it is offered to them, these findings do need to be treated with caution because they are based on retrospective analyses.

For instance, a primary care clinic in the US ran nurse-facilitated group health coaching sessions to support behaviour change. The sessions included education plus an opportunity for brief one-to-one health coaching with motivational interviewing. An average of 13 people attended each session. Those from minority ethnic groups were more likely to attend and more likely to achieve their goals.³⁴¹

Another US study examined the demographic and clinical characteristics of adults who accessed self-help health coaching online for binge eating. Data from 4,051 men and women who registered for the programme over a 13month period were analysed. Women were more likely to use the initiative. Most participants reported binge eating below the frequency required for a clinical diagnosis, but reported high motivation to overcome their eating binges. Few had received prior treatment for an eating disorder. The researchers concluded that digital health coaching may be useful for individuals who might otherwise not seek or receive treatment.³⁴²

A survey of 34,291 employees from 52 US companies providing worksite health programmes examined factors influencing participation in health coaching. People were more likely to enrol in health coaching if they were older, female and in poorer health. Programme completion was greater amongst those who were older, did not use tobacco and worked at a company with strong communications.³⁴³

In Australia, an automated online health coaching programme was set up to support smoking cessation. The initiative was designed to be used repeatedly over time, guiding the user through the process of smoking cessation in the manner of a 'life coach.' Email reminders were sent to prompt repeated use. Over a four-year period, 28,247 people completed an initial assessment and 84% registered. 62% of users were women and the median age was 34 years. 37% were people who had recently begun smoking again following a quit attempt. 27% of those who registered returned for a second visit, a median of nine days after their first visit. 11% completed a third visit. Women and older smokers were more likely to stay engaged.344

Another Australian study described how 4,828 people from the general population were offered six months of health coaching or detailed self-help information. Those using the health coaching service were representative of the adult population regarding education, employment status, fruit and vegetable consumption and alcohol use. Proportionately more female, middleaged, English-speaking, rural and socially disadvantaged adults took part. Participants were more likely to be overweight and ex-smokers than the general population.³⁴⁵

Stage of change

Studies of the potential barriers to making change within health coaching relationships or the reasons why people do not complete health coaching programmes suggest that health coaching may be most acceptable to those who are motivated to change their behaviour or lifestyle.^{346,347} In other words, **health coaching may work best when people have already decided that they want to do something differently** and health coaching provides a tool to help them achieve this.

There is some evidence that professionals using health coaching skills within routine consultations adapt the techniques according to people's stage of change, but some research suggests that health coaching skills are most commonly applied with people in the 'preparation' stage.³⁴⁸

The overall message from empirical evidence might be that health coaching needs to be targeted towards those ready and wanting to make a change, regardless of their demographic background. People who have the most severe symptoms or who smoke or are overweight and want to do something about this might be good candidates. Whilst women may be easier to engage, health coaching has also been found to work well for men so there may be a need to target messages about health coaching appropriately for a male audience. Media advertising has been found to have some success in promoting health coaching, particular among men, those of working age and lower socioeconomic groups.^{349,350}

Targeting less advantaged groups

Available evidence

The review identified two randomised trials and six other studies focused on targeting health coaching to less advantaged groups. Other studies included in the review may also have included a diverse range of participants, but these studies were particularly focused on the applicability of health coaching to specific groups.

Key trends

- As outlined in the previous section, people from minority ethnic groups and those from lower socio-economic groups may be more likely to participate in and continue to engage with health coaching if offered an opportunity to do so. Therefore there are questions about whether special targeting is really needed, or whether instead it is important to focus on making health coaching available to a wide range of groups.
- Research has not explored the best ways to increase the uptake of health coaching amongst specific groups.
 Studies of financial incentives for a broad range of people suggest that incentives can encourage people to take part in standalone health coaching programmes, but this is not specific to vulnerable groups.³⁵¹

Examples

Supporting minority ethnic groups

Most studies do not specifically target minority ethnic groups, but there are some favourable findings. For example, in the US 67 people with cancer and moderate pain, including 15 people from minority ethnic groups, were assigned to standardised information about controlling pain (control group) or a 20-minute individualised education and health coaching session to increase knowledge of pain self-management and rehearse an individually scripted conversation with clinicians about pain control. At enrolment, people from minority ethnic groups reported more pain than others. At two-week follow-up, people from minority ethnic groups in the control group continued to have more pain but those in the intervention group did not.³⁵²

In another US study, 1,051 South Asian people took part in health coaching for one year. Culturally specific health coaching was found to be appealing and feasible for reducing cardiovascular disease risk. There was low attrition.³⁵³ Elsewhere in the US, a non-random study compared a 27-week automated telephone diabetes self-management support / health coaching initiative for low income people from diverse ethnic groups versus waitlist controls. At six months, the health coaching group had greater improvements in diabetes self-care behaviours and perceived physical wellbeing. There were no differences between groups in perceived patientcentred processes of care or cardio metabolic outcomes.³⁵⁴

But not all findings are favourable. A randomised trial with 201 low income and largely minority ethnic group people with poorly controlled diabetes compared a 24minute video and workbook about behaviour change and five sessions of health coaching versus a 20-page educational booklet. There were no differences between groups in blood sugar control, blood pressure, cholesterol, diabetes knowledge or self-care behaviours. The researchers concluded that a larger number of sessions may be needed for less advantaged people.³⁵⁵

A non-randomised comparison in the US explored outcomes from five Chinese-American medical assistants who took part in five one-hour health coaching training sessions and then worked with Chinese-American patients. Ninety-two people with diabetes were assigned to receive health coaching or usual care. At six months, there was no difference in blood sugar control.³⁵⁶

Supporting other vulnerable groups

A small number of studies have explored targeting health coaching towards other particularly vulnerable groups.³⁵⁷

In the US, four homeless, two formerly homeless and six low-income people with long-term conditions took part in 12weeks of free face-to-face health coaching, which would usually be paid for. Health coaching was found to help motivate people and be modestly useful for breaking through social isolation and loneliness. It helped people develop capacity for accomplishing short-term self-identified goals and provided an opportunity for people to become more active in initiating contact with health services.³⁵⁸

In Oman, 98 largely illiterate people with diabetes were provided with dietary and lifestyle health coaching. After three years, there were improvements in blood sugar control and cholesterol. Improvements in weight and body mass index were seen in women only.³⁵⁹

These studies suggest that health coaching can be as effective for particularly vulnerable groups as for others, but do not provide insights into the best ways to target health coaching to reach such groups. Part 4: Who makes the best health coaches?

Types of health coaches

Available evidence

The review identified two systematic reviews, six randomised trials and nine other studies exploring the role of different professionals or peers in facilitating health coaching. Other studies in the review included information about the type of professionals applying health coaching skills, but these studies were particularly focused on assessing how certain professionals could use health coaching or comparing different coaches.

Key trends

- There is research about nurses, doctors, pharmacists and allied health professionals providing health coaching. Health coaching provided by peers and laypeople has also been researched. Studies do not tend to compare one type of coach with another so it is not possible to say that certain professionals are more likely to engage with or be effective at using health coaching skills.
- Although professionals sometimes express doubts about the feasibility of using health coaching skills within routine practice or their ability to motivate people towards behaviour change,³⁶⁰ there is evidence that a wide range of professionals and peers can successfully apply health coaching skills.

Examples

Peers

In recent years, there has been an increasing focus on testing whether peers (people with similar conditions) can feasibly provide health coaching.³⁶¹ A number of studies have found favourable results.³⁶² For instance, in the US 23 people with diabetes who completed 36 hours of health coaching training acted as peer coaches. 299 patients with low incomes were randomly assigned to peer health coaching or usual care. At six months, peer health coaching was associated with a greater improvement in blood sugar control.³⁶³

A focus group and 17 interviews with community-based peer coaches supporting people with diabetes in the US found that peer coaches played three roles: advisor, supporter and role model. Peer coaches had varying styles and approaches to setting emotional boundaries and allocating responsibility for implementing health behaviour changes, but all felt they were more empowered themselves to better manage their own diabetes as a result of facilitating health coaching.³⁶⁴ Another US study explored which characteristics of peer coaches were most associated with improvement in diabetes control among low-income people. Twenty people provided peer health coaching to 109 patients over a six-month period. Patient improvement in blood sugar control was associated with having a coach with a lower sense of self-efficacy in diabetes management, higher level of diabetes-related distress and lower depression score. In other words, peer coaches do not need to be the most knowledgeable, confident or well.³⁶⁵

This is supported by a randomised trial comparing professional and different types of peer group health coaching in the US. Forty-four people were assigned to weight loss support plus either a professional or peer coach. Peer coaches were either people in the same situation (termed peers) or those who had successfully lost weight (termed mentors). Groups met weekly for six weeks, then biweekly for six weeks and monthly thereafter, for a total of 12 meetings. Between meetings participants emailed weight loss information to their coach and received feedback. Retention was 95%. Professional and peer coaches were associated with greater weight loss than mentors.³⁶⁶

Nurses and doctors

Many studies of health coaching involve nurses. A systematic review of 50 studies of various behaviour change support strategies in primary care found that nurses and doctors were equally capable of facilitating such initiatives. ³⁶⁷

There are many examples of GPs or hospital doctors providing health coaching,³⁶⁸ and some studies suggest it is useful when doctors work alongside nurses.³⁶⁹

Allied health professionals

A wide variety of allied professionals have been trained as health coaches, including pharmacists, physical therapists, dieticians and occupational therapists.³⁷⁰

A randomised trial in Australia examined whether dieticians or nurses who did not prescribe medications could coach patients with coronary heart disease to work with their doctors to achieve target levels for total cholesterol and other risk factors. 792 patients from six hospitals were assigned usual care or health coaching via telephone and mailings. At six months, health coaching was associated with improvements in total cholesterol and quality of life compared to usual care.³⁷¹ Focus groups in the US examined whether doctors accepted non-clinicians trained as health coaches for people with high blood pressure. Doctors supported the delegation of functions to health coaches. They liked the high frequency of interactions between coaches and patients and thought that health coaching could improve medication adherence. They believed that good communication between coaches and clinicians was important.³⁷²

Teams

Ensuring whole teams have health coaching skills has been found useful. In Wales, 26 hospital paediatric diabetes teams were randomly assigned to be trained in health coaching-style consultation skills or not. Recording and analysis of consultations found that trained staff were more capable than controls in guiding and agenda setting, though skills waned over time. Training had no effect on patients' blood sugar control. ³⁷³

Training coaches

Available evidence

The review identified one randomised trial and 12 other studies focused on training professionals or peers to use health coaching skills. Other studies sometimes also mentioned the training undertaken, but these studies had a special emphasis on training.

Key trends

- There is insufficient evidence to draw conclusions about the most effective ways of training professionals or peers in health coaching skills. Some studies suggest that active role plays and observation can work well rather than solely theoretical content. Well evaluated training usually lasts at least two to five days and includes follow-up support.
- Time, commitment and motivation may be key things needed by coaches.³⁷⁴ Empirical studies were not identified comparing specific health coaching competencies.
- There is some research to suggest that the context in which health coaching is implemented is important. In other words, professionals need to be supported to provide health coaching practically and emotionally and there needs to be a receptive organisational context.^{375,376}

Examples

Satisfaction with training

Training in health coaching is often wellreceived by professionals and usually results in an increase in knowledge or health coaching skills, at least in the short term.^{377,378}

An online survey with 150 professionals that took part in health coaching skills development programmes in the UK found that clinicians thought that health coaching skills were highly relevant and that health coaching could help clinicians with improved problem-solving and interpersonal skills. They also thought that health coaching could improve patient motivation, readiness to change and selfmanagement.³⁷⁹

In a random trial in the US, first-year medical students acted as health coaches for people with diabetes, offering at least six sessions per patient over a year-long period. Students attended eight 90-minute small group teaching sessions about communication and behavioural change planning. Initially some students complained about increased workloads compared with those completing traditional education or had reservations about their ability to support patients, but these complaints diminished as students worked with patients.³⁸⁰ Not all training is well received, however. In Wales, interviews with 27 exercise professionals, 10 area coordinators and a trainer in motivational interviewing examined the value of a two-day workshop for delivering motivational interviewing-based health coaching within an exercise referral programme. Most saw a potential role for motivational interviewing skills but some said it was difficult to incorporate this into routine practice, especially given the need for structured data gathering activities. There were no overall improvements in practice observed.³⁸¹

Training approach

There are a number of descriptive studies outlining the broad approach to training professionals in health coaching skills, but these are not sufficiently detailed to extract information about the competencies suggested for coaches or the content of training programmes.

In the UK, 14 GP trainees were invited to attend a health coaching skills training course spanning five day-long sessions. Each trainee was asked to apply health coaching skills to up to four people with long-term conditions over a one to twomonth period. The study found that it is feasible and useful to train GPs in health coaching. Health coaching had selfreported benefits for GP trainees and their patients. Training of at least two days for GPs was suggested, with opportunities for follow-up and observed practice.³⁸²

A US study described training peers to provide telephone behavioural change health coaching using motivational interviewing. Fifty-six volunteers were trained using a dvd followed with two or three booster sessions. 89% of participants reported learning about peer counselling and 87% reported learning communication skills. There was a small improvement in self-efficacy in motivational interviewing skills.³⁸³ Elsewhere in the US, a lay health coach provided follow-up support via face-toface visits, e-mail, telephone contacts, or a combination thereof every two to four weeks for up to six months for obese people in primary care. The initial fourweek training programme for the health coach consisted of review of videotapes, reading assignments, a self-change project and practising with two patients. Half (48%) of the 92 patients initiated a behaviour change in eating habits, physical activity or both.³⁸⁴

Building in refresher sessions may be useful.³⁸⁵ In South Africa, 39 lay counsellors took part in 35 hours of training in motivational interviewingbased health coaching to help people adhere to antiretroviral treatment. The laypeople struggled to achieve proficiency so 18 hours of refresher training and one year of supervision were implemented. Audio-recordings of 22 counselling sessions were analysed. Refresher training and supervision was associated with some improvement in behaviour change skills.³⁸⁶

Training content

A study in the UK examined the essential competencies (attitudes, knowledge and skills) needed to train a diabetes team in behaviour change counselling and appropriate training methods. Interviews with staff suggested that training should cover the Stages of Change model, motivational interviewing and behavioural change techniques. Acquiring the competencies was more difficult than anticipated, though most competencies were achieved after one year of training. Individual supervision and video examples were the most valued training methods.³⁸⁷

A number of studies have suggested that role-playing with observation may be a useful component of the content of health coaching training.³⁸⁸

Involving professionals in translating theory to practice may also have benefits. A cohort study in the US described how nurses used health coaching skills during home visits to support behaviour change in low-income mothers. The researchers suggested that a success factor was orientating nurses to self-efficacy theories and involving them in developing programme materials to translate theoretical concepts into nursing interventions.³⁸⁹

Studies about training in behaviour change skills

As the empirical evidence about training in health coaching is sparse, the review also considered trends in additional studies about more general training to support behaviour change. This included studies about training in motivational interviewing and in behaviour change counselling.

The review did not seek to include an exhaustive range of studies of this nature, but rather to draw out key trends in readily available literature. Five systematic reviews, 13 randomised trials and 49 other studies were included.

Studies have examined the extent to which behaviour change support is provided in routine care and concluded this is often lacking. This may be due a wide range of barriers, including a lack of knowledge, confidence and support.^{390, 391, 392} For instance, a UK study interviewed 33 nurses, doctors, allied health professionals and clinical support staff from inpatient and outpatient departments at a children's hospital about their attitudes and beliefs towards supporting lifestyle behaviour change. Professionals identified a range of barriers, including constraints associated with the hospital environment, the perceived inappropriateness of providing advice given the patient's condition, job role priorities and perceived limits of the advice given. Providing lifestyle behaviour change support was often seen as an educational activity, rather than a behaviour change activity.³⁹³ Studies such as this suggest that professionals may need further support and training to successfully provide behaviour change counselling and motivational interviewing. A number of studies have explored the most appropriate and useful forms for such training.

Professional participants

Training in behaviour change skills has taken place with doctors, nurses and allied health professionals working in hospital, in primary care and in other community settings.^{394,395,396,397,398,399} Teachers, dentists and pharmacists have also been trained to support health-related behaviour change,^{400,401,402,403} as have peers.^{404,405}

A systematic review of ten studies about motivational interviewing training for GPs found that the average length of the training was nine hours. Common elements included basic motivational interviewing skills, the motivational interviewing mind-set or spirit, recognising and reinforcing change talk and addressing resistance. Most studies included follow-up sessions. The training was generally associated with improvements in skills.⁴⁰⁶

Training may not be equally effective for all professionals. Just as health coaching may be most effective for people who are motivated and ready to change, so too may training in behaviour change skills be most worthwhile for practitioners who have an appropriate mind-set.⁴⁰⁷

Short workshops

Several studies have described short workshops to provide professionals with basic skills in behaviour change techniques such as motivational interviewing.⁴⁰⁸ For example, a UK hospital evaluated a fourhour workshop in motivational interviewing for cystic fibrosis teams, followed six months later by a workshop on applying motivational interviewing during brief consultations. Feedback from interviews and surveys with 50 health professionals suggested that all participants had used motivational interviewing skills after taking part in the training and rated their skills as 'moderate.'409

In Canada, ten dieticians, pharmacists, nurses and social workers took part in a one-day workshop about motivational interviewing. Interviews conducted prior to and at one and four weeks after training found that initially professionals had low confidence in their motivational skills and thought that this approach would be 'hard work.' After the training, professionals reported feeling more inspired and confident. They said they were focusing more on partnering with patients and less on giving advice.⁴¹⁰

Workshops as short as two hours in duration have been associated with improvements in motivational interviewing and behaviour change knowledge and skills, though most workshops evaluated are half or full days.^{411,412,413,414,415,416} Single short workshops or a series of short workshops have been associated with clearer communication, decreased use of closed-ended questions, increased use of reflections, increased use of open-ended questions and increased use of motivational interviewing techniques.⁴¹⁷

A meta-analysis of the effect of motivational interviewing training included 15 studies with 715 clinicians. Training was associated with medium effects on behaviour, though the reviewers emphasised that this might be due to publication bias whereby studies with positive findings are more likely to be reported.⁴¹⁸

Sometimes a short workshop is followed by one or two refresher sessions or telephone calls, and this has been found to work well.^{419,420}

However, a study in Sweden suggested that short training may not help practitioners improve their competence. Thirty-six nurses took part in a three-anda-half day workshop in motivational interviewing followed by four sessions of supervision and observation. Participants did not reach proficiency.⁴²¹ This is supported by other studies that suggest that short workshops may not help professionals achieve objectively assessed proficiency in behaviour change skills.^{422,423,424}

Longer courses

Many slightly longer courses in behaviour change techniques are undertaken as part of the training of medical, nursing or allied health professional trainees.^{425,426,427,428,429,430} These longer courses tend to be associated with improved confidence and skills amongst professionals.⁴³¹

A systematic review of 109 behaviour change counselling courses for medical students found that most studies came from the US (90%) and took place in primary care settings (74%). Curricular topics included techniques to support smoking cessation (61% of studies), nutrition (28%), reduced alcohol or drug use (24%) and exercise (20%). Most studies did not include theoretical frameworks, but one third used the Transtheoretical Stage of Change model (36%). Half of the studies involved eight or fewer hours of curricular time (56%) and spanned four or fewer weeks (47%). Training that employed multiple techniques and included practice in either simulated or actual clinical settings had the best outcomes.432

Whilst most studies find some improvement in (often self-reported) knowledge and skills immediately after training, this may not always be sustained. A randomised trial in the Netherlands found that primary care nurses who took part in workshops about motivational interviewing had no better skills one year later than nurses who had not taken part.⁴³³

Role play

Practice and rehearsal is often beneficial in helping professionals to acquire communication skills. Some studies have explored the best ways to do this to support motivational interviewing skills.⁴³⁴ Role plays have emerged as an important component of many programmes.^{435,436}

In the US, following a lecture about motivational interviewing, first-year pharmacy students were randomised to practice using written dialogue, peer roleplay or mock-patient counselling activities. Those taking part in simulated patient encounters demonstrated better motivational interviewing skills and knowledge.⁴³⁷

But some studies suggest that role plays with others taking part in training programmes may be just as effective as role plays with simulated patients.⁴³⁸ For example, health professionals from Wales who attended a two-day workshop in motivational interviewing were randomly assigned to conduct practice sessions with either a simulated patient or a another trainee during the training. Their competence was assessed before and after training using a validated scale. There was no difference in skill levels between groups following training.⁴³⁹

Observation

Researchers from The Netherlands examined the effects of filming and providing feedback on the communication skills and motivational interviewing skills of experienced practice nurses working in primary care. Consultations between 17 nurses and 325 patients were filmed at two time-points, three to six months apart. Nurses were randomly assigned to receive feedback about their videos or not. The group who received feedback later communicated more clearly and paid more attention to the agenda setting and permission seeking aspects of motivational interviewing during their consultations. The researchers concluded that videoing and providing feedback may be an important component of training in motivational interviewing skills.440

Online resources

Online modules have been tested to train professionals to support behaviour change,⁴⁴¹ but these omit the crucial role of practice and feedback. In the US, five three-hour long live video workshops about motivational interviewing were delivered a month apart. The video signal was transmitted through telephone lines to staff at substance abuse treatment organisations. The first workshop had the largest participation rates. Some participants were frustrated by interrupted audio or video signals. Handouts and videotaped examples of motivational interviewing were thought to be helpful aspects of the training. There were selfreported improvements in knowledge and skills, however audiotapes of actual counselling sessions suggested only minimal improvement in skills.442

To mitigate the lack of feedback, blended styles of learning have been tested whereby online materials are used alongside group gatherings or structured practice sessions.^{443,444}

In the US, primary care staff were trained in motivational interviewing using a halfday in-person workshop followed by an hour-long virtual training session followed by another workshop. Training improved knowledge and confidence, but there were no impacts on perceived comfort level and skill in lifestyle counselling or job-related burnout.⁴⁴⁵

'Virtual worlds' have also been tested as a way to give professionals an opportunity to practice their behaviour change skills.⁴⁴⁶

Follow-up sessions

Providing some form of supervision, monitoring or follow-up training may help professionals to maintain and apply their behaviour change skills after training ends.^{447,448,449,450,451,452,453}

A meta-analysis of 21 studies about training in motivational interviewing examined ways to sustain people's skills after training finished. Training was initially associated with improved skills, but programmes that did not include feedback or ongoing supervision or support reported that skills eroded over a sixmonth period. Having three to four feedback or supervision sessions during the six months following training helped to sustain skills.⁴⁵⁴

Supervision via teleconferencing has been found to be feasible.^{455,456}

A randomised trial in the UK assessed whether a workplace-based educational intervention helped 54 professionals who had trained in motivational interviewing become more competent in these skills. One group took part in an update day about motivational interviewing (control), another group took part in the update day plus received 12 weekly worksheets to stimulate reflection on their practice of motivational interviewing, feedback about audiotaped sessions with patients and three 30-minute sessions of telephone support. The group receiving ongoing support had greater increases in competence.457

To summarise, due to a lack of comparative research, it is not possible to say that a certain approach to training in behavioural change skills is more effective than others. However **training that includes observation, role play and other practical activities seems to be associated with positive outcomes**. Shorter and longer courses have both been tested, but their relative effectiveness has not been documented.

Researchers from the US tested three strategies for training clinicians in motivational interviewing: self-study, expert-led workshops and train-the-trainer workshops. The workshop groups also took part in three monthly supervision sessions. The number of clinicians meeting motivational interviewing performance standards was measured at 12-week follow-up. The study found that self-study was likely to be the most cost-effective training strategy if the threshold value is less than about US\$2870 per person meeting the criteria. Expert-led workshops were likely to be most cost-effective when the threshold value is greater than about US\$2870.458

Part 5: Moving forward

Implications for policy and practice

This review of research about health coaching is one of many tools that commissioners can use when considering whether health coaching is effective, for whom and how it could be offered in the NHS. Used alongside local evaluations and expertise about the local context and priorities, the empirical evidence suggests some implications for policy and practice.

Should health coaching be rolled out?

- There is evidence to suggest that health coaching has potential, but it cannot be assumed to be a panacea for all the challenges facing the NHS. Health coaching has been associated with improvements in self-efficacy and behaviour change, suggesting that this approach can help people to feel more motivated and empowered to make healthy lifestyle changes.
- The impacts on health outcomes are more mixed, possibly because many studies may be too small to show a difference or too short to capture any changes over time. There is some positive evidence about the potential of health coaching for improving clinical outcomes in people with diabetes, high blood pressure, high cholesterol and high risks of heart attack or stroke.

- There is very little evidence about the impact of health coaching on health service use or resource use. Whilst there are some positive studies, it would be inappropriate to use these to suggest that health coaching has a definite return on investment. Most of the positive studies are small or methodologically weak, and draw on data from health systems that are very different from the UK.
- A number of studies have found that health coaching is not cost-effective or does not reduce the use of healthcare resources. These too need to be treated with caution because they may have been of insufficient duration to show an impact. Changes to behaviour made as a result of health coaching may take a while to impact on physical outcomes and health service use.
- The overall message from the evidence base is that there are many benefits likely associated with health coaching, but in order to be effective health coaching may need to be implemented as part of a wider programme supporting education and behaviour change.

When does health coaching work best?

- There is less evidence about where, how and for whom health coaching is most effective. The evidence base is not strong enough to conclude that face-to-face health coaching is more effective than telephone health coaching, for example, or that individual health coaching is more effective than group health coaching. This is due to a lack of studies comparing different approaches.
- Similarly, it is difficult to generalise about who health coaching may be most effective for. Studies that find benefits from health coaching often suggest that the benefits are most marked for people who are motivated to change from the outset and those who have the most severe symptoms or exacerbations. Women have been found in some studies to be more amenable to health coaching than men, but other studies have found that men can benefit equally.
- There is no evidence to suggest that health coaching is most effective for people with certain conditions.
- Few studies have explored strategies to ensure that health coaching is well targeted for the most vulnerable groups, though research has found that health coaching can work well for people from minority ethnic groups, lower socio-economic groups and the homeless.

How should coaches be trained?

- Nurses, doctors, pharmacists and physical therapists are the most commonly researched health coaches. There is also evidence emerging about the role of people with long-term conditions themselves acting as coaches for others.
- The evidence base is insufficient to conclude that one type of professional or peer is more effective than others in applying health coaching skills. There are few studies that compare one type of coach with another, but the findings of individual studies suggest that health coaching skills can be applied by a wide range of professionals, either in routine practice or as part of bespoke health coaching consultations.
- Almost all of the studies included in the review state that the professionals facilitating health coaching have received specialised training, however the details of this training are sparse. Very few studies explicitly examine the training offered to health coaches. The studies that do exist suggest that the most effective training may be at least two to five days in duration, incorporate observation, practical activities and role plays and allow for follow-up and refresher training.

Adding to the evidence base

- This review suggests that health coaching has potential for the NHS, but there are gaps in knowledge about how best to train coaches and the longer-term impacts of health coaching, particularly on service use and costs. Box 2 lists caveats with the evidence available.
- As teams within the NHS consider rolling out health coaching, it may be important to also consider more robust evaluation that compares health coaching with other options and compares different types of health coaching. Analysis of the costs and cost-effectiveness of health coaching should be built into any evaluation.
- There is an opportunity for NHS teams to contribute significantly to knowledge in this area, potentially making an empowering and worthwhile form of support available to a wide range of people.

All of the implications listed in this section are important for both national audiences as well as the East of England when considering how, when and who health coaching may work best for, how to select and train people as health coaches and gaps in current knowledge.

To summarise, Box 1 lists some of the key implications of the empirical evidence for the East of England programme.

Box 1: Implications for East of England

- There is evidence to support the East of England's testing of health coaching, but this initiative should be seen as one component of wider programmes to support selfmanagement rather than a standalone entity.
- The East of England model of having at least two days' worth of training follows good practice. There may be scope to continue to refine the model. For example, refresher or follow-up sessions may be beneficial for participants.
- There may be a need to build in content about how to use health coaching effectively in routine practice. Most of the evidence available is about standalone coaching initiatives.
- More guidance may be needed about how to select professionals to take part in training and how to select patients that health coaching may be most effective with. A more targeted approach may be useful.
- Evaluation needs to focus on clinical and cost outcomes, rather than solely impacts on professionals' skills. Given the extent of investment in the programme, a robust evaluation strategy is needed.
- There is scope to share the lessons learned widely, both in terms of the story so far and the evidence base.
 Given the paucity of evidence, it may be useful to publish key trends from the evidence review as a journal article.

1. Variation in interventions

There are some issues that readers should keep in mind when interpreting studies about health coaching.

There has been no clear definition of what comprises health coaching and studies have used this or similar terms to represent widely varying interventions. Health coaching can take a variety of forms so it is difficult to say that health coaching as a concept works because some interventions may be several months long and some might be one-off, some might be delivered by peers and others by professionals, some might be telephonebased and others delivered face to face, and so on. The role that coaches take and the competencies they have can also vary markedly. In short, it is difficult to compare studies because the interventions included are vastly different, much like comparing apples and pears.

This means that where studies have found that health coaching does not work, it is important to consider the exact health coaching methods, providers, duration and frequency used rather than assuming the concept as a whole may be flawed.

2. Variations in quality of evidence

Furthermore, the quality of the evidence varies widely. The review has weighted the quality of evidence based on study design, whereby randomised trials and systematic reviews are potentially more robust than other studies. Many of the studies that have found positive outcomes from health coaching are not systematic reviews or randomised trials.

3. Lack of comparative evidence

Most studies do not compare health coaching with other alternatives. This means that even where studies suggest that health coaching has improved attitudes or behaviours over time, it is not possible to say whether health coaching has done this more quickly or effectively than usual care or other types of support.

Furthermore, there is little evidence about the cost-effectiveness of health coaching. Thus studies may have found positive benefits, but they do not explore at what cost or the opportunity costs involved.

4. Generalisability issues

Much of the evidence available about health coaching comes from the US, where health systems, attitudes, commissioning and payments and are very different from the UK. Whilst studies from other countries can provide useful insights into potential trends, it is usually not possible to transfer interventions from one health ecosystem to another without adaptation.

Importantly, most of the evidence relates to health coaching set up as a separate intervention rather than used within routine practice, such as GP consultations so it may be difficult to extrapolate the findings to the effects of using health coaching within routine practice. It remains unclear whether health coaching from a clinician with an established relationship has greater benefits.

These caveats emphasise the importance of East of England evaluating its programme robustly and adding to the evidence base.

Annex and references

Annex 1: Identifying evidence

This annex describes the approach used to identify evidence for inclusion in the review.

Inclusion and exclusion criteria

The review was completed over a threeweek period. It included readily available empirical published and grey literature from the UK and other countries.

To be eligible for inclusion in the review, material had to:

- be empirical research of any methodological design;
- be published in a print or online journal or available via the grey literature;
- include information about outcomes from health coaching or factors influencing effectiveness;
- and be published in the English language.

Only studies that stated the intervention was 'health coaching' or 'health coaching' were eligible. Studies of related topics such as behaviour change and motivational interviewing were not included (except in terms of training).

Studies published in any year were eligible for inclusion. There were no geographic restrictions.

Search strategy

To identify studies for inclusion in the review, two reviewers independently searched ten bibliographic databases, comprising Medline / Pubmed, Web of Science, Cumulative Index to Nursing and Allied Health Literature (CINAHL), Google Scholar, Applied Social Science Information and Abstracts (ASSIA), PsycInfo, ProQuest, Social Care Online, Social Services Abstracts and GreyLit.

In addition, the bibliographies of identified articles and the websites of relevant organisations such as Age Concern, the Department of Health, the US Institute for Healthcare Improvement, WHO and other think tanks and policy groups were searched. Experts in the field were contacted for further empirical work.

The search terms included combinations of health coaching, health coach, wellness health coaching, wellbeing health coaching, health coaching for health, health coaching, coach, activated consumer, behaviour change, behaviour change counselling, lifestyle modification, navigator, lifestyle counselling, motivational interviewing, self-efficacy, training, cost and similes.

Study selection

A total of 7,294 unique potentially relevant citations were identified. Two reviewers independently screened the titles and abstracts, rating the studies as 'relevant', 'not relevant' or 'uncertain.' The full text of all 'relevant' and 'uncertain' studies was retrieved and screened for eligibility. The findings of those that met the inclusion criteria were summarised for incorporation in the review (see Figure 1).

Figure 1: Selecting studies for the review



Appraising quality

Studies of any quality were included. No formal weighting of quality was undertaken because the aim was to illustrate the diversity of the material available. The review also sought to address questions about who is best placed to facilitate health coaching and who health coaching is most effective with which may be unlikely to be answered by trials alone. However, the results of systematic reviews and randomised trials were reported separately from the results of other studies to make it clear where more robust evidence was available.

Synthesising evidence

In total, 275 studies about health coaching were included in the review, with an additional 67 about training in behavioural change or motivational interviewing.

The review aimed to report high level themes. To synthesise material, the broad findings of each study were extracted independently by two reviewers using a template and then compiled. The studies were heterogeneous in terms of their focus, definitions, research design, size and geographic contexts so quantitative synthesis was not appropriate. A narrative synthesis was undertaken, grouping the literature according to key themes and specific review questions.

The review was funded by Health Education East of England. All selection and analysis was undertaken independently of the funder.

References

- Wolever RQ, Simmons LA, Sforzo GA, Dill D, Kaye M, Bechard EM, Southard ME, Kennedy M, Vosloo J, Yang N. <u>A systematic review of the literature on health and</u> wellness health coaching: defining a key behavioral intervention in healthcare. Glob Adv Health Med 2013;2(4):38-57.
- Bennett HD, Coleman EA, Parry C, Bodenheimer T, Chen EH. <u>Health coaching for patients with chronic illness.</u> Fam Pract Manag 2010;17(5):24-29.
- 3 Edwards A, Elwyn G. Inside the black box of shared decision making: distinguishing between the process of involvement and who makes the decision. Health Expect 2006;9(4):307-320.
- 4 Janz NK, Wren PA, Copeland LA, Lowery JC, Goldfarb SL, Wilkins EG. <u>Patient-physician concordance: preferences</u>, <u>perceptions, and factors influencing the breast cancer</u> <u>surgical decision</u>, J Clin Oncol 2004;22(15):3091-3098.
- 5 Moumjid N, Carrère MO, Charavel M, Brémond A. Clinical issues in shared decision-making applied to breast cancer, Health Expect 2003;6(3):222-227.
- 6 Whelan T, Levine M, Willan A, Gafni A, Sanders K, Mirsky D, Chambers S, O'Brien MA, Reid S, Dubois S. <u>Effect of a decision aid on knowledge and treatment decision making for breast cancer surgery: a randomized trial.</u> JAMA 2004;292(4):435-441.
- 7 van Roosmalen MS, Stalmeier PF, Verhoef LC, Hoekstra-Weebers JE, Oosterwijk JC, Hoogerbrugge N, Moog U, van Daal WA. <u>Randomised trial of a decision aid and its</u> <u>timing for women being tested for a BRCA1/2 mutation.</u> *Br J Cancer* 2004;90(2):333-342.
- 8 Mandelblatt J, Kreling B, Figeuriedo M, Feng S. <u>What is</u> the impact of shared decision making on treatment and outcomes for older women with breast cancer? J Clin Oncol 2006;24(30):4908-4913.
- 9 Merenstein D, Diener-West M, Krist A, Pinneger M, Cooper LA. <u>An assessment of the shared-decision model</u> in parents of children with acute otitis media. *Pediatrics* 2005;116(6):1267-1275.
- 10 Penticuff JH, Arheart KL. <u>Effectiveness of an intervention</u> to improve parent-professional collaboration in <u>neonatal intensive care</u> J Perinat Neonatal Nurs 2005;19(2):187-202.
- 11 Padilha JM. <u>Influence of perception of chronic</u> <u>obstructive pulmonary disease in promotion of self-</u> <u>management of disease.</u> *Rev Port Pneumol* 2010;16(4):641-648.
- 12 Lubetkin EI, Lu WH, Gold MR. <u>Levels and correlates of</u> patient activation in health center settings: building <u>strategies for improving health outcomes</u>. J Health Care Poor Underserved 2010;21(3):796-808.
- 13 Shigaki C, Kruse RL, Mehr D, Sheldon KM, Bin Ge, Moore C, Lemaster J. <u>Motivation and diabetes self-</u> <u>management.</u> Chronic Illn 2010;6(3):202-214.
- 14 Kaptein AA, Klok T, Moss-Morris R, Brand PL. <u>Illness</u> perceptions: impact on self-management and control in asthma. Curr Opin Allergy Clin Immunol 2010;10(3):194-199.
- 15 Gallagher R. <u>Self management, symptom monitoring</u> and associated factors in people with heart failure living in the community. *Eur J Cardiovasc Nurs* 2010;9(3):153-160.

- 16 Coulter A, Ellins J. *Patient-focused Interventions: A review* of the Evidence. London: The Health Foundation, 2006.
- 17 Remmers C, Hibbard J, Mosen DM, Wagenfield M, Hoye RE, Jones C. <u>Is patient activation associated with future</u> <u>health outcomes and healthcare utilization among</u> <u>patients with diabetes?</u> J Ambul Care Manage 2009;32(4):320-327.
- 18 Shon KH, Park SS. <u>Medication and symptom</u> <u>management education program for the rehabilitation</u> <u>of psychiatric patients in Korea: the effects of promoting</u> <u>schedule on self-efficacy theory.</u> *Yonsei Med J* 2002;43(5):579-589.
- 19 Blissmer B, Prochaska JO, Velicer WF, Redding CA, Rossi JS, Greene GW, Paiva A, Robbins M. <u>Common factors predicting long-term changes in multiple health behaviors</u>. J Health Psychol 2010;15(2):205-214.
- 20 Song M. <u>Diabetes mellitus and the importance of selfcare</u>. J Cardiovasc Nurs 2010;25(2):93-98.
- 21 Fishman L, Barendse RM, Hait E, Burdick C, Arnold J. Self-management of older adolescents with inflammatory bowel disease: a pilot study of behavior and knowledge as prelude to transition. *Clin Pediatr* 2010:49(12):1129-1133.
- 22 Mancuso CA, Sayles W, Allegrante JP. <u>Knowledge</u>, <u>attitude</u>, and self-efficacy in asthma self-management and quality of life. J Asthma 2010:47(8):883-888.
- 23 Sol BG, van der Graaf Y, van Petersen R, Visseren FL. <u>The effect of self-efficacy on cardiovascular lifestyle.</u> Eur J Cardiovasc Nurs 2011;10(3):180-186.
- 24 Chih AH, Jan CF, Shu SG, Lue BH. <u>Self-efficacy affects</u> <u>blood sugar control among adolescents with type I</u> <u>diabetes mellitus</u>. J Formos Med Assoc 2010;109(7): 503-510.
- 25 Du S, Yuan C. <u>Evaluation of patient self-management</u> outcomes in health care: a systematic review. *Int Nurs Rev* 2010;57(2): 159-167.
- 26 Campos DC, Graveto JM. <u>The role of nurses and</u> <u>patients' involvement in the clinical decision-making</u> <u>process</u>, *Rev Lat Am Enfermagem* 2009;17(6):1065-1070.
- 27 Baars JE, Markus T, Kuipers EJ, van der Woude CJ. Patients' preferences regarding shared decision-making in the treatment of inflammatory bowel disease: results from a patient-empowerment study. Digestion 2010;81(2):113-119.
- 28 Schattner A, Bronstein A, Jellin N. <u>Information and shared decision-making are top patients' priorities</u>. BMC Health Serv Res 2006;6:21.
- 29 Loh A, Kremer N, Giersdorf N, Jahn H, Hänselmann S, Bermejo I, Härter M. Information and participation interests of patients with depression in clinical decision making in primary care. Z Arztl Fortbild Qualitatssich 2004;98(2):101-107.
- 30 Charles C, Gafni A, Whelan T. <u>Self-reported use of shared decision-making among breast cancer specialists and perceived barriers and facilitators to implementing this approach.</u> *Health Expect* 2004;7(4):338-348.
- 31 Elwyn G, Gwyn R, Edwards A, Grol R. <u>Is 'shared decision-making' feasible in consultations for upper respiratory tract infections? Assessing the influence of antibiotic expectations using discourse analysis. *Health Expect* 1999;2(2):105-117.</u>

- 32 Elwyn G, Edwards A. Shared decision-making in health care: Achieving evidence-based patient choice (2nd ed). Oxford: Oxford University Press, 2009.
- 33 World Health Organisation. Innovative Care for Chronic Conditions. Building Blocks for Action. Geneva: WHO, 2002.
- 34 Department of Health. Improving Chronic Disease Management. London: Department of Health, 2004.
- 35 Department of Health. *Improving Chronic Disease Management*. London: Department of Health, 2004.
- 36 Department of Health. *Improving quality of life for people with long term conditions.* London: Department of Health, 2013.
- 37 Department of Health. *Equity and Excellence: Liberating the NHS*. London: Stationery Office, 2010.
- 38 http://services.parliament.uk/bills/2010-11/healthandsocialcare.html
- 39 Huffman MH. <u>Health coaching: a fresh, new approach to improve quality outcomes and compliance for patients with chronic conditions.</u> Home Healthc Nurse 2009;27(8):490-496;
- 40 Kreitzer MJ, Sierpina VS, Lawson K. <u>Health coaching:</u> <u>innovative education and clinical programs emerging.</u> *Explore* 2008;4(2):154-155.
- 41 Wolever RQ, Eisenberg DM. <u>What is health coaching</u> anyway? Standards needed to enable rigorous research. *Arch Intern Med* (Published online October 2011).
- 42 Palmer S. <u>Health coaching to facilitate the promotion of healthy behaviour and achievement of health-related goals</u>. *Int J Health Prom Educ* 2003;41;91-93.
- 43 Olsen JM. <u>Health coaching: a concept analysis.</u> *Nurs Forum* 2014;49(1):18-29.
- 44 Licht M, Davis JM, Scripps A, Cone J. <u>Whole person</u> <u>health for the whole population: one-year evaluation of</u> <u>health coaching.</u> *Perm J* 2007;11(4):41-44.
- 45 Wetzel TG. <u>Health coaching.</u> Hosp Health Netw 2011;85(5):20-24.
- 46 Wolever RQ, Simmons LA, Sforzo GA, Dill D, Kaye M, Bechard EM, Southard ME, Kennedy M, Vosloo J, Yang N. <u>A systematic review of the literature on health and</u> wellness health coaching: defining a key behavioral intervention in healthcare. Glob Adv Health Med 2013;2(4):38-57.
- 47 Linden A, Butterworth S, Prochaska J. <u>Motivational interviewing-based health coaching as a chronic care intervention</u>. J Eval Clin Prac 2009;16(1);166-174.
- 48 Lawson K. <u>Could health coaching build a bridge to a new system of healthcare?</u> Altern Ther Health Med 2009;15(5):16-18.
- 49 Appiah-Brempong E, Okyere P, Owusu-Addo E, Cross R. <u>Motivational interviewing interventions and alcohol</u> <u>abuse among college students: a systematic review.</u> *Am J Health Promot* (Published online March 2014).
- 50 Schwalbe CS, Oh HY, Zweben A. <u>Sustaining motivational interviewing: a meta-analysis of training studies.</u> Addiction (Published online March 2014).
- 51 Gayes LA, Steele RG. <u>A meta-analysis of motivational interviewing interventions for pediatric health behavior change</u>, *J Consult Clin Psychol* (Published online February 2014).
- 52 Riper H, Andersson G, Hunter SB, de Wit J, Berking M, Cuijpers P. <u>Treatment of comorbid alcohol use disorders</u> and depression with cognitive-behavioural therapy and <u>motivational interviewing: a meta-analysis</u>. *Addiction* 2014;109(3):394-406.

- 53 Lundahl B, Moleni T, Burke BL, Butters R, Tollefson D, Butler C, Rollnick S. <u>Motivational interviewing in medical</u> <u>care settings: a systematic review and meta-analysis of</u> <u>randomized controlled trials.</u> *Patient Educ Couns* 2013;93(2):157-168
- 54 Vanbuskirk KA, Wetherell JL. <u>Motivational interviewing</u> with primary care populations: a systematic review and <u>meta-analysis</u>. J Behav Med (Published online August 2013).
- 55 Purath J, Keck A, Fitzgerald CE. <u>Motivational interviewing</u> for older adults in primary care: a systematic review. *Geriatr Nurs* (Published online March 2014).
- 56 Leveille SG, Huang A, Tsai SB, Allen M, Weingart SN, Iezzoni LI. <u>Health coaching via an internet portal for</u> primary care patients with chronic conditions: a randomized controlled trial. *Med Care* 2009;47(1):41-47.
- 57 Allen M, Iezzoni LI, Huang A, Huang L, Leveille SG. <u>Improving patient-clinician communication about</u> <u>chronic conditions: description of an internet-based</u> <u>nurse E-coach intervention.</u> Nurs Res 2008;57(2):107-112.
- 58 Armstrong C, Wolever RQ, Manning L, Elam R 3rd, Moore M, Frates EP, Duskey H, Anderson C, Curtis RL, Masemer S, Lawson K. <u>Group health coaching: strengths,</u> <u>challenges, and next steps.</u> *Glob Adv Health Med* 2013;2(3):95-102.
- 59 Lipscomb R. <u>Health coaching: a new opportunity for dietetics professionals.</u> J Am Diet Assoc 2006;106(6):801-803.
- 60 Wagner PJ, Jester DM, Moseley GC. <u>Medical students as</u> health coaches. Acad Med 2002;77(11):1164-1165.
- 61 <u>www.kingsfund.org.uk/publications/supporting-people-</u> manage-their-health
- 62 Bodenheimer T et al. <u>Helping Patients Manage Their</u> <u>Chronic Conditions</u>. California Health Care Foundation, 2005.
- 63 Brownson CA et al. A quality improvement tool to assess self-management support in primary care. <u>Jt Comm J</u> <u>Qual Pat Saf</u> 2007; 33:7.
- 64 Coulter A, Ellins J. <u>How engaged are people in their</u> <u>healthcare2</u> The Health Foundation, 2005.
- 65 Thoolen B, De Ridder D, Bensing J, Maas C, Griffin S, Gorter K, Rutten G. Effectiveness of a self-management intervention in patients with screen-detected type 2 diabetes. <u>Diabetes Care</u> 2007; 30(11): 2832-7.
- 66 Jordan JE, Briggs AM, Brand CA, Osborne RH. Enhancing patient engagement in chronic disease selfmanagement support initiatives in Australia: the need for an integrated approach. *Med J Aust* 2008; 189 (10 Suppl): S9-S13.
- 67 Glasgow NJ, Jeon YH, Kraus SG, Pearce-Brown CL. Chronic disease self-management support: the way forward for Australia. *MJA* 2008; 189 (10 Suppl): S14-S16.
- 68 Wagner EH, Davis C, Schaefer J et al. A survey of leading chronic disease management programs: are they consistent with the literature? *Manag Care Q* 1999; 7: 56-66.
- 69 Nolte E, McKee M (eds). *Caring for people with chronic conditions: a health system perspective*. Maidenhead, UK: McGraw Hill Open University Press, 2008.
- 70 Harris MF, Williams AM, Dennis SM, et al. Chronic disease self-management: implementation with and within Australian general practice. *Med J Aust* 2008; 189 (10 Suppl): S17-S20.
- 71 Neuner-Jehle S, Schmid M, Grüninger U. <u>The "health</u> <u>coaching" programme: a new patient-centred and</u> <u>visually supported approach for health behaviour</u> <u>change in primary care.</u> *BMC Fam Pract* 2013;14:100.

- 72 Karapinar-Carkit F, Borgsteede SD, Zoer J, Siegert C, van Tulder M, Egberts AC, van den Bemt PM. <u>The effect of</u> <u>the COACH program (Continuity Of Appropriate</u> <u>pharmacotherapy, patient Counselling and information</u> <u>transfer in Healthcare) on readmission rates in a</u> <u>multicultural population of internal medicine patients.</u> *BMC Health Serv Res* 2010;10:39.
- 73 Swerczek LM, Banister C, Bloomberg GR, Bruns JM, Epstein J, Highstein GR, Jamerson PA, Sterkel R, Wells S, Garbutt JM. <u>A telephone health coaching intervention to</u> <u>improve asthma self-management behaviors</u>. *Pediatr Nurs* 2013;39(3):125-130, 145.
- 74 Leveille SG, Huang A, Tsai SB, Allen M, Weingart SN, Iezzoni LI. <u>Health coaching via an internet portal for</u> primary care patients with chronic conditions: a randomized controlled trial, Med Care 2009;47(1):41-47.
- 75 Licht M, Davis JM, Scripps A, Cone J. <u>Whole person</u> health for the whole population: one-year evaluation of health coaching. *Perm J* 2007;11(4):41-44.
- 76 Wolever RQ, Webber DM, Meunier JP, Greeson JM, Lausier ER, Gaudet TW. <u>Modifiable disease risk, readiness to change, and psychosocial functioning improve with integrative medicine immersion model.</u> *Altern Ther Health Med* 2011;17(4):38-47.
- 77 Iles R, Taylor NF, Davidson M, O'Halloran P. <u>Telephone</u> health coaching can increase activity levels for people with non-chronic low back pain: a randomised trial. J Physiother 2011;57(4):231-238.
- 78 Brook O, van Hout H, Nieuwenhuyse H, Heerdink E. Impact of health coaching by community pharmacists on drug attitude of depressive primary care patients and acceptability to patients; a randomized controlled trial. *Eur Neuropsychopharmacol* 2003;13(1):1-9.
- 79 Wang CJ, Fetzer SJ, Yang YC, Wang JJ. <u>The impacts of using community health volunteers to coach medication safety behaviors among rural elders with chronic illnesses</u>, *Geriatr Nurs* 2013;34(2):138-145.
- 80 Jordan M. <u>Health coaching for the underserved</u>. *Glob Adv Health Med* 2013;2(3):75-82.
- 81 Howard LM, Hagen BF. Experiences of persons with type II diabetes receiving health coaching: an exploratory gualitative study. Educ Health 2012;25(1):66-69.
- 82 Stretton C, Mudge S, Kayes NM, Taylor D, McPherson KM. Activity health coaching to improve walking is liked by rehabilitation patients but physiotherapists have concerns: a qualitative study. J Physiother 2013;59(3):199-206.
- 83 Walters JA, Cameron-Tucker H, Courtney-Pratt H, Nelson M, Robinson A, Scott J, Turner P, Walters EH, Wood-Baker R. <u>Supporting health behaviour change in</u> <u>chronic obstructive pulmonary disease with telephone</u> <u>health-mentoring: insights from a qualitative study.</u> BMC Fam Pract 2012;13:55.
- 84 Linden A, Butterworth SW, Prochaska JO. <u>Motivational interviewing-based health coaching as a chronic care intervention</u>. J Eval Clin Pract 2010;16(1):166-174.
- 85 Newman P, McDowell A. Exploring the benefits of health coaching for clinicians and patients. HSJ 2011.
- 86 Basak Cinar A, Schou L. <u>Health promotion for patients</u> with diabetes: health coaching or formal health education? Int Dent J 2014;64(1):20-28.
- 87 Basak Cinar A, Schou L. <u>The role of self-efficacy in health coaching and health education for patients with type 2 diabetes</u>. *Int Dent J* (Published online February 2014).

- 88 Hayashi A, Kayama M, Ando K, Ono M, Suzukamo Y, Michimata A, Akiyama MO, Fukuhara S, Izumi SI. <u>Analysis of subjective evaluations of the functions of tele-health coaching intervention in patients with spinocerebellar degeneration</u>. *NeuroRehabilitation* 2008;23(2):159-169.
- 89 O'Neil A, Hawkes AL, Atherton JJ, Patrao TA, Sanderson K, Wolfe R, Taylor CB, Oldenburg B. <u>Telephone-delivered health coaching improves anxiety outcomes after myocardial infarction: the 'proactive heart' trial.</u> Eur J Prev Cardiol 2014;21(1):30-38.
- 90 Ngo V, Hammer H, Bodenheimer T. <u>Health coaching in the teamlet model: a case study.</u> J Gen Intern Med 2010;25(12):1375-1378.
- 91 Wallace AS, Seligman HK, Davis TC, Schillinger D, Arnold CL, Bryant-Shilliday B, Freburger JK, DeWalt DA. <u>Literacy-appropriate educational materials and brief</u> <u>counseling improve diabetes self-management</u>. *Patient Educ Couns* 2009;75(3):328-333.
- 92 van der Wulp I, de Leeuw JR, Gorter KJ, Rutten GE. Effectiveness of peer-led self-management health coaching for patients recently diagnosed with type 2 diabetes mellitus in primary care: a randomized controlled trial. Diabet Med 2012;29(10):e390-e397.
- 93 Taylor TK, Webster-Stratton C, Feil EG, Broadbent B, Widdop CS, Severson HH. <u>Computer-based intervention</u> with health coaching: an example using the incredible years program, Cogn Behav Ther 2008;37(4):233-246.
- 94 Foster L, Dunn W, Lawson LM. <u>Health coaching mothers</u> of children with autism: a qualitative study for occupational therapy practice. *Phys Occup Ther Pediatr* 2013;33(2):253-263.
- 95 Graham F, Rodger S, Ziviani J. <u>Effectiveness of</u> occupational performance health coaching in improving children's and mothers' performance and mothers' selfcompetence. *Am J Occup Ther* 2013;67(1):10-18.
- 96 Piamjariyakul U, Smith CE, Russell C, Werkowitch M, Elyachar A. <u>The feasibility of a telephone health</u> <u>coaching program on heart failure home management</u> <u>for family caregivers.</u> *Heart Lung* 2013;42(1):32-39.
- 97 Sepulveda AR, Lopez C, Macdonald P, Treasure J. <u>Feasibility and acceptability of DVD and telephone</u> <u>health coaching-based skills training for carers of</u> <u>people with an eating disorder.</u> Int J Eat Disord 2008;41(4):318-325.
- 98 van Mierlo LD, Meiland FJ, Dröes RM. <u>Dementelcoach:</u> <u>effect of telephone health coaching on carers of</u> <u>community-dwelling people with dementia. Int</u> *Psychogeriatr* 2012;24(2):212-222.
- 99 Prevatt F, Yelland S. <u>An empirical evaluation of ADHD</u> <u>health coaching in college students.</u> J Atten Disord (Published online March 2013).
- 100 Caldwell KL, Gray J, Wolever RQ. <u>The process of patient</u> empowerment in integrative health coaching: how does <u>it happen?</u> *Glob Adv Health Med* 2013;2(3):48-57.
- 101 Nelson KA, Highstein G, Garbutt J, Trinkaus K, Smith SR, Strunk RC. <u>Factors associated with attaining health</u> coaching goals during an intervention to improve child <u>asthma care.</u> Contemp Clin Trials 2012;33(5):912-919.
- 102 Hudlicka E. <u>Virtual training and health coaching of</u> health behavior: example from mindfulness meditation training. Patient Educ Couns 2013;92(2):160-166.
- 103 Kravitz RL, Tancredi DJ, Grennan T, Kalauokalani D, Street RL Jr, Slee CK, Wun T, Oliver JW, Lorig K, Franks P. Cancer Health Empowerment for Living without Pain (Ca-HELP): effects of a tailored education and health coaching intervention on pain and impairment. Pain 2011;152(7):1572-1582.

- 104 Olsen JM, Nesbitt BJ. <u>Health coaching to improve</u> <u>healthy lifestyle behaviors: an integrative review.</u> Am J Health Promot 2010;25(1):e1-e12.
- 105 Michie S, Abraham C, Whittington C, McAteer J, Gupta S. <u>Effective techniques in healthy eating and physical</u> <u>activity interventions: a meta-regression</u>. *Health Psychol* 2009;28(6):690-701.
- 106 Cox NS, Alison JA, Holland AE. <u>Interventions for</u> promoting physical activity in people with cystic fibrosis. *Cochrane Database Syst Rev* 2013;12:CD009448.
- 107 Currie S, Sinclair M, Murphy MH, Madden E, Dunwoody L, Liddle D. <u>Reducing the decline in physical activity</u> <u>during pregnancy: a systematic review of behaviour</u> <u>change interventions.</u> *PLoS One* 2013;8(6):e66385.
- 108 Desroches S, Lapointe A, Ratté S, Gravel K, Légaré F, Turcotte S. <u>Interventions to enhance adherence to</u> <u>dietary advice for preventing and managing chronic</u> <u>diseases in adults.</u> *Cochrane Database Syst Rev* 2013;2:CD008722.
- 109 Desroches S, Lapointe A, Ratté S, Gravel K, Légaré F, Turcotte S. <u>Interventions to enhance adherence to</u> <u>dietary advice for preventing and managing chronic</u> <u>diseases in adults.</u> *Cochrane Database Syst Rev* 2013;2:CD008722.
- 110 Holland SK, Greenberg J, Tidwell L, Malone J, Mullan J, Newcomer R. <u>Community-based health coaching,</u> <u>exercise, and health service utilization.</u> J Aging Health 2005;17(6):697-716.
- 111 Van Hoecke AS, Delecluse C, Opdenacker J, Lipkens L, Martien S, Boen F. Long-term effectiveness and mediators of a need-supportive physical activity health coaching among Flemish sedentary employees. Health Promot Int 2013;28(3):407-417.
- 112 Iles RA, Taylor NF, Davidson M, O'Halloran P. <u>An</u> effective health coaching intervention for people with low recovery expectations and low back pain: a content analysis. J Back Musculoskelet Rehabil 2014;27(1):93-101
- 113 Tidwell L, Holland SK, Greenberg J, Malone J, Mullan J, Newcomer R. <u>Community-based nurse health coaching</u> and its effect on fitness participation. *Lippincotts Case Manag* 2004;9(6):267-279.
- 114 Hawkes AL, Patrao TA, Green A, Aitken JF. <u>CanPrevent: a</u> telephone-delivered intervention to reduce multiple behavioural risk factors for colorectal cancer. *BMC Cancer* 2012;12:560.
- 115 Heimendinger J, Uyeki T, Andhara A, Marshall JA, Scarbro S, Belansky E, Crane L. <u>Health coaching process</u> outcomes of a family visit nutrition and physical activity intervention. *Health Educ Behav* 2007;34(1):71-89.
- 116 Yang NY, Wroth S, Parham C, Strait M, Simmons LA. Personalized health planning with integrative health coaching to reduce obesity risk among women gaining excess weight during pregnancy. Glob Adv Health Med 2013;2(4):72-77.
- 117 O'Hara BJ, Phongsavan P, Venugopal K, Eakin EG, Eggins D, Caterson H, King L, Allman-Farinelli M, Haas M, Bauman AE. <u>Effectiveness of Australia's Get Healthy</u> information and health coaching service: translational research with population wide impact. *Prev Med* 2012;55(4):292-298.
- 118 LeCheminant JD, Merrill RM. <u>Improved health behaviors</u> persist over two years for employees in a worksite wellness program. Popul Health Manag 2012;15(5):261-266.

- 119 van de Glind IM, Heinen MM, Evers AW, Wensing M, van Achterberg T. Factors influencing the implementation of a lifestyle counseling program in patients with venous leg ulcers: a multiple case study. Implement Sci 2012;7:104.
- 120 Rabius V, McAlister AL, Geiger A, Huang P, Todd R. <u>Telephone counseling increases cessation rates among</u> <u>young adult smokers.</u> *Health Psychol* 2004;23(5):539-541.
- 121 Terry PE, Seaverson EL, Staufacker MJ, Tanaka A. <u>The</u> <u>effectiveness of a telephone-based tobacco cessation</u> <u>program offered as part of a worksite health promotion</u> <u>program.</u> *Popul Health Manag* 2011;14(3):117-125.
- 122 An LC, Zhu SH, Nelson DB, Arikian NJ, Nugent S, Partin MR, Joseph AM. <u>Benefits of telephone care over primary</u> <u>care for smoking cessation: a randomized trial</u>. *Arch Intern Med* 2006;166(5):536-542.
- 123 Sieber W, Newsome A, Lillie D. <u>Promoting self-</u> management in diabetes: efficacy of a collaborative care approach. Fam Syst Health 2012;30(4):322-329.
- 124 Kravitz RL, Tancredi DJ, Jerant A, Saito N, Street RL, Grennan T, Franks P. <u>Influence of patient health</u> <u>coaching on analgesic treatment adjustment: secondary</u> <u>analysis of a randomized controlled trial.</u> *J Pain Symptom Manage* 2012;43(5):874-884.
- 125 Jimison HB, Klein KA, Marcoe JL. <u>A socialization</u> intervention in remote health coaching for older adults in the home. Conf Proc IEEE Eng Med Biol Soc 2013;2013:7025-7028.
- 126 Tanio C, Chen C. <u>Innovations at Miami practice show</u> promise for treating high-risk Medicare patients. *Health Aff* 2013;32(6):1078-1082.
- 127 Melko CN, Terry PE, Camp K, Xi M, Healey ML. <u>Diabetes</u> <u>health coaching improves medication adherence: a pilot</u> <u>study.</u> *Am J Lifestyle Med* 2010;4(2):187-194.
- 128 Mohr DC, Duffecy J, Ho J, Kwasny M, Cai X, Burns MN, Begale M. <u>A randomized controlled trial evaluating a</u> <u>manualized TeleHealth coaching protocol for improving</u> <u>adherence to a web-based intervention for the</u> <u>treatment of depression</u>, *PLoS One* 2013;8(8):e70086.
- 129 Ratanawongsa N, Handley MA, Sarkar U, Quan J, Pfeifer K, Soria C, Schillinger D. <u>Diabetes health information</u> technology innovation to improve quality of life for health plan members in urban safety net. J Ambul Care Manage 2014;37(2):127-37.
- 130 Lin JJ, Mann DM. <u>Application of persuasion and health</u> <u>behavior theories for behavior change counseling:</u> <u>design of the ADAPT (Avoiding Diabetes Thru Action</u> <u>Plan Targeting) program.</u> *Patient Educ Couns* 2012;88(3):460-466.
- 131 Hanks C, Kitzman H, Milligan R. <u>Implementing the</u> <u>COACH relationship model: health promotion for</u> <u>mothers and children.</u> ANS Adv Nurs Sci 1995;18(2):57-66.
- 132 Wongpiriyayothar A, Piamjariyakul U, Williams PD. <u>Effects of patient teaching, educational materials, and</u> <u>health coaching using telephone on dyspnea and</u> <u>physical functioning among persons with heart failure.</u> *Appl Nurs Res* 2011;24(4):e59-e66.
- 133 Keenan S, King G, Curran CJ, McPherson A. Effectiveness of experiential life skills health coaching for youth with a <u>disability</u>. Phys Occup Ther Pediatr (Published online October 2013).
- 134 Wolever RQ, Webber DM, Meunier JP, Greeson JM, Lausier ER, Gaudet TW. <u>Modifiable disease risk</u>, readiness to change, and psychosocial functioning improve with integrative medicine immersion model. *Altern Ther Health Med* 2011;17(4):38-47.

- 135 Sides EG, Zimmer LO, Wilson L, Pan W, Olson DM, Peterson ED, Bushnell C. <u>Medication health coaching</u> program for patients with minor stroke or TIA: a pilot study, BMC Public Health 2012;12:549.
- 136 McAlister AL, Rabius V, Geiger A, Glynn TJ, Huang P, Todd R. <u>Telephone assistance for smoking cessation</u>: <u>one year cost effectiveness estimations</u>. *Tob Control* 2004;13(1):85-86.
- 137 Paineau DL, Beaufils F, Boulier A, Cassuto DA, Chwalow J, Combris P, Couet C, Jouret B, Lafay L, Laville M, Mahe S, Ricour C, Romon M, Simon C, Tauber M, Valensi P, Chapalain V, Zourabichvili O, Bornet F. <u>Family dietary health coaching to improve nutritional intakes and body weight control: a randomized controlled trial. Arch Pediatr Adolesc Med 2008;162(1):34-43.</u>
- 138 Hoddinott P, Lee AJ, Pill R. <u>Effectiveness of a</u> <u>breastfeeding peer health coaching intervention in rural</u> <u>Scotland</u>, *Birth* 2006;33(1):27-36.
- 139 Brook OH, van Hout H, Stalman W, Nieuwenhuyse H, Bakker B, Heerdink E, de Haan M. <u>A pharmacy-based</u> health coaching program to improve adherence to antidepressant treatment among primary care patients. *Psychiatr Serv* 2005;56(4):487-489.
- 140 Secker-Walker RH, Solomon LJ, Flynn BS, Skelly JM, Mead PB. <u>Reducing smoking during pregnancy and postpartum: physician's advice supported by individual counseling</u>. *Prev Med* 1998;27(3):422-430.
- 141 Brodin N, Eurenius E, Jensen I, Nisell R, Opava CH. <u>Health coaching patients with early rheumatoid arthritis</u> to healthy physical activity: a multicenter, randomized, <u>controlled study</u>, *Arthritis Rheum* 2008;59(3):325-331.
- 142 Hawkes AL, Chambers SK, Pakenham KI, Patrao TA, Baade PD, Lynch BM, Aitken JF, Meng X, Courneya KS. Effects of a telephone-delivered multiple health behavior change intervention (CanChange) on health and behavioral outcomes in survivors of colorectal cancer: a randomized controlled trial. J Clin Oncol 2013;31(18):2313-2321.
- 143 Adelman AM, Graybill M. <u>Integrating a health coach into</u> primary care: reflections from the Penn state ambulatory research network. Ann Fam Med 2005;3 Suppl 2:S33-S35.
- 144 Holmes-Rovner M, Stommel M, Corser WD, Olomu A, Holtrop JS, Siddiqi A, Dunn SL. <u>Does outpatient</u> telephone health coaching add to hospital quality improvement following hospitalization for acute coronary syndrome? J Gen Intern Med 2008;23(9):1464-1470.
- 145 Taylor TK, Webster-Stratton C, Feil EG, Broadbent B, Widdop CS, Severson HH. <u>Computer-based intervention</u> with health coaching: an example using the incredible years program. *Cogn Behav Ther* 2008;37(4):233-246.
- 146 Smeerdijk M, Keet R, Dekker N, van Raaij B, Krikke M, Koeter M, de Haan L, Barrowclough C, Schippers G, Linszen D. <u>Motivational interviewing and interaction</u> skills training for parents to change cannabis use in young adults with recent-onset schizophrenia: a randomized controlled trial, *Psychol Med* 2012;42(8):1627-1636.
- 147 Smith LL, Lake NH, Simmons LA, Perlman A, Wroth S, Wolever RQ. <u>Integrative health coach training: a model</u> for shifting the paradigm toward patient-centricity and <u>meeting new national prevention goals.</u> *Glob Adv Health Med* 2013;2(3):66-74.
- 148 Wang CJ, Fetzer SJ, Yang YC, Wang JJ. <u>The impacts of</u> using community health volunteers to coach medication safety behaviors among rural elders with chronic illnesses. *Geriatr Nurs* 2013;34(2):138-145.

- 149 Joseph DH, Griffin M, Hall RF, Sullivan ED. <u>Peer health</u> coaching: an intervention for individuals struggling with diabetes. *Diabetes Educ* 2001;27(5):703-710.
- 150 McDonald DD, Walsh S, Vergara C, Gifford T. <u>Effect of a virtual pain coach on pain management discussions: a pilot study</u>. *Pain Manag Nurs* 2013;14(4):200-209.
- 151 McDonald DD, Gifford T, Walsh S. <u>Effect of a virtual pain</u> coach on older adults' pain communication: a pilot <u>study</u>. *Pain Manag Nurs* 2011;12(1):50-56.
- 152 Verwey R, van der Weegen S, Spreeuwenberg M, Tange H, van der Weijden T, de Witte L. <u>A pilot study of a tool</u> to stimulate physical activity in patients with COPD or type 2 diabetes in primary care. J Telemed Telecare 2014;20(1):29-34.
- 153 Yan T, Wilber KH, Simmons WJ. <u>Motivating high-risk</u> older adults to exercise: does health coaching matter? *Home Health Care Serv Q* 2011;30(2):84-95.
- 154 Hurling R, Claessen JP, Nicholson J, Schäfer F, Tomlin CC, Lowe CF. <u>Automated health coaching to help</u> parents increase their children's brushing frequency: an <u>exploratory trial.</u> *Community Dent Health* 2013;30(2):88-93.
- 155 Johnston BD, Rivara FP, Droesch RM, Dunn C, Copass MK. <u>Behavior change counseling in the emergency</u> <u>department to reduce injury risk: a randomized,</u> <u>controlled trial, Pediatrics</u> 2002;110(2 Pt 1):267-274.
- 156 Butler CC, Simpson SA, Hood K, Cohen D, Pickles T, Spanou C, McCambridge J, Moore L, Randell E, Alam MF, Kinnersley P, Edwards A, Smith C, Rollnick S. <u>Training practitioners to deliver opportunistic multiple</u> <u>behaviour change counselling in primary care: a cluster</u> <u>randomised trial.</u> *BMJ* 2013;346:f1191.
- 157 Hutchison AJ, Breckon JD. <u>A review of telephone health</u> coaching services for people with long-term conditions. *J Telemed Telecare* 2011;17(8):451-458.
- 158 Dennis SM, Harris M, Lloyd J, Powell Davies G, Faruqi N, Zwar N. <u>Do people with existing chronic conditions</u> <u>benefit from telephone health coaching?</u> A rapid review. *Aust Health Rev* 2013;37(3):381-388.
- 159 Noordman J, van der Weijden T, van Dulmen S. <u>Communication-related behavior change techniques</u> <u>used in face-to-face lifestyle interventions in primary</u> <u>care: a systematic review of the literature</u>. *Patient Educ Couns* 2012;89(2):227-244.
- 160 Avery L, Flynn D, van Wersch A, Sniehotta FF, Trenell MI. Changing physical activity behavior in type 2 diabetes: a systematic review and meta-analysis of behavioral interventions, Diabetes Care 2012;35(12):2681-2689.
- 161 Martin J, Chater A, Lorencatto F. <u>Effective behaviour</u> <u>change techniques in the prevention and management</u> <u>of childhood obesity</u>. *Int J Obes* 2013;37(10):1287-1294.
- 162 Ammentorp J, Uhrenfeldt L, Angel F, Ehrensvärd M, Carlsen EB, Kofoed PE. <u>Can life health coaching improve</u> <u>health outcomes? A systematic review of intervention</u> <u>studies.</u> BMC Health Serv Res 2013;13:428.
- 163 Saleh SS, Alameddine MS, Hill D, Darney-Beuhler J, Morgan A. <u>The effectiveness and cost-effectiveness of a</u> <u>rural employer-based wellness program.</u> J Rural Health 2010;26(3):259-265.
- 164 DeBar LL, Ritenbaugh C, Aickin M, Orwoll E, Elliot D, Dickerson J, Vuckovic N, Stevens VJ, Moe E, Irving LM. Youth: a health plan-based lifestyle intervention increases bone mineral density in adolescent girls. Arch Pediatr Adolesc Med 2006;160(12):1269-1276.
- 165 Lanese BS, Dey A, Srivastava P, Figler R. <u>Introducing the health coach at a primary care practice: a pilot study (part 2)</u>, *Hosp Top* 2011;89(2):37-42.

- 166 Goessens BM, Visseren FL, de Nooijer J, van den Borne HW, Algra A, Wierdsma J, van der Graaf Y. <u>A pilot-study</u> to identify the feasibility of an Internet-based health coaching programme for changing the vascular risk profile of high-risk patients. Patient Educ Couns 2008;73(1):67-72.
- 167 Thom DH, Ghorob A, Hessler D, De Vore D, Chen E, Bodenheimer TA. <u>Impact of peer health coaching on</u> <u>glycemic control in low-income patients with diabetes: a</u> <u>randomized controlled trial</u>. *Ann Fam Med* 2013;11(2):137-144.
- 168 Sacco WP, Malone JI, Morrison AD, Friedman A, Wells K. Effect of a brief, regular telephone intervention by paraprofessionals for type 2 diabetes. J Behav Med 2009;32(4):349-359.
- 169 Cinar AB, Oktay I, Schou L. <u>"Smile healthy to your diabetes": health coaching-based intervention for oral health and diabetes management.</u> *Clin Oral Investig* (Published online December 2013).
- 170 Sacco WP, Morrison AD, Malone JI. <u>A brief, regular,</u> proactive telephone "health coaching" intervention for diabetes: rationale, description, and preliminary results. J Diabetes Complications 2004;18(2):113-118.
- 171 Whittemore R, Chase S, Mandle CL, Roy SC. <u>The content</u>, <u>integrity</u>, and efficacy of a nurse health coaching <u>intervention in type 2 diabetes</u>. *Diabetes Educ* 2001;27(6):887-898.
- 172 Nelson K, Pitaro M, Tzellas A, Lum A. <u>Practice profile.</u> <u>Transforming the role of medical assistants in chronic</u> <u>disease management. Health Aff</u> 2010;29(5):963-965.
- 173 Fera T, Bluml BM, Ellis WM, Schaller CW, Garrett DG. <u>The diabetes ten city challenge: interim clinical and humanistic outcomes of a multisite community pharmacy diabetes care program.</u> J Am Pharm Assoc 2008;48(2):181-190.
- 174 Ammentorp J, Thomsen J, Kofoed PE. <u>Adolescents with</u> poorly controlled type 1 diabetes can benefit from health coaching: a case report and discussion. J Clin Psychol Med Settings 2013;20(3):343-350.
- 175 Patja K, Absetz P, Auvinen A, Tokola K, Kytö J, Oksman E, Kuronen R, Ovaska T, Harno K, Nenonen M, Wiklund T, Kettunen R, Talja M. <u>Health coaching by telephony to support self-care in chronic diseases: clinical outcomes from the TERVA randomized controlled trial. BMC Health Serv Res 2012;12:147.</u>
- 176 Sebo P, Abetel G, Stalder H, Bovier PA. <u>Importance of</u> <u>lifestyle counselling by primary care physicians for</u> <u>diabetic patients.</u> *Swiss Med Wkly* 2006;136(35-36):566-573.
- 177 Tillman J. <u>Health coaching interventions with a heart-healthy Lenoir project client</u>. *Glob Adv Health Med* 2013;2(3):83-86.
- 178 Leung LB, Busch AM, Nottage SL, Arellano N, Glieberman E, Busch NJ, Smith SR. <u>Approach to</u> <u>antihypertensive adherence: a feasibility study on the</u> <u>use of student health coaches for uninsured</u> <u>hypertensive adults</u>, *Behav Med* 2012;38(1):19-27.
- 179 Vale MJ, Jelinek MV, Best JD, Dart AM, Grigg LE, Hare DL, Ho BP, Newman RW, McNeil JJ. <u>Health coaching</u> <u>patients On Achieving Cardiovascular Health (COACH): a</u> <u>multicenter randomized trial in patients with coronary</u> <u>heart disease</u>. Arch Intern Med 2003;163(22):2775-2783.
- 180 Cho JH, Kwon HS, Kim HS, Oh JA, Yoon KH. <u>Effects on diabetes management of a health-care provider mediated, remote health coaching system via a PDA-type glucometer and the Internet.</u> J Telemed Telecare 2011;17(7):365-370.

- 181 Fritsch MA, Montpellier J, Kussman C. <u>Worksite wellness:</u> <u>a cholesterol awareness program.</u> AAOHN J 2009;57(2):69-76, quiz 77-78.
- 182 Al-Sinani M, Min Y, Ghebremeskel K, Qazaq HS. Effectiveness of and adherence to dietary and lifestyle counselling: effect on metabolic control in type 2 diabetic Omani patients. Sultan Qaboos Univ Med J 2010;10(3):341-349.
- 183 Edelman D, Oddone EZ, Liebowitz RS, Yancy WS Jr, Olsen MK, Jeffreys AS, Moon SD, Harris AC, Smith LL, Quillian-Wolever RE, Gaudet TW. <u>A multidimensional integrative medicine intervention to improve</u> <u>cardiovascular risk</u>, J Gen Intern Med 2006;21(7):728-734.
- 184 Nolan RP, Upshur RE, Lynn H, Crichton T, Rukholm E, Stewart DE, Alter DA, Chessex C, Harvey PJ, Grace SL, Picard L, Michel I, Angus J, Corace K, Barry-Bianchi SM, Chen MH. <u>Therapeutic benefit of preventive telehealth</u> <u>counseling in the community outreach heart health and</u> <u>risk reduction trial.</u> Am J Cardiol 2011;107(5):690-696.
- 185 Aoun S, Osseiran-Moisson R, Shahid S, Howat P, O' Connor M. <u>Telephone lifestyle health coaching: is it</u> <u>feasible as a behavioural change intervention for men?</u> J Health Psychol 2012;17(2):227-236.
- 186 White M, Stinson JN, Lingley-Pottie P, McGrath PJ, Gill N, Vijenthira A. <u>Exploring therapeutic alliance with an</u> <u>internet-based self-management program with brief</u> <u>telephone support for youth with arthritis: a pilot study.</u> *Telemed J E Health* 2012;18(4):271-276.
- 187 Adams SR, Goler NC, Sanna RS, Boccio M, Bellamy DJ, Brown SD, Neugebauer RS, Ferrara A, Wiley DM, Schmittdiel JA. <u>Patient satisfaction and perceived</u> <u>success with a telephonic health coaching program: the</u> <u>Natural Experiments for Translation in Diabetes (NEXT-D) study, Northern California, 2011. Prev Chronic Dis</u> 2013;10:E179.
- 188 Shahnazari M, Ceresa C, Foley S, Fong A, Zidaru E, Moody S. <u>Nutrition-focused wellness health coaching</u> promotes a reduction in body weight in overweight US veterans. J Acad Nutr Diet 2013;113(7):928-935.
- 189 Seghers J, Van Hoecke AS, Schotte A, Opdenacker J, Boen F. <u>The added value of a brief self-efficacy health</u> <u>coaching on the effectiveness of a 12-week physical</u> <u>activity program.</u> J Phys Act Health 2014;11(1):18-29.
- 190 Rimmer JH, Wang E, Pellegrini CA, Lullo C, Gerber BS. <u>Telehealth weight management intervention for adults</u> with physical disabilities: a randomized controlled trial. *Am J Phys Med Rehabil* 2013;92(12):1084-1094.
- 191 Ball GD, Mackenzie-Rife KA, Newton MS, Alloway CA, Slack JM, Plotnikoff RC, Goran MI. <u>One-on-one lifestyle</u> <u>health coaching for managing adolescent obesity:</u> <u>findings from a pilot. randomized controlled trial in a</u> <u>real-world. clinical setting.</u> *Paediatr Child Health* 2011;16(6):345-350.
- 192 Moore C. <u>Case report of hemoglobin alc and weight</u> reduction in integrative health coaching. *Glob Adv Health Med* 2013;2(3):87-89.
- 193 Temmingh H, Claassen A, van Zyl S, Carrara H, Dayakalashe NM, Myer L, Stein DJ. <u>The evaluation of a</u> <u>telephonic wellness health coaching intervention for</u> <u>weight reduction and wellness improvement in a</u> <u>community-based cohort of persons with serious</u> <u>mental illness.</u> J Nerv Ment Dis 2013;201(11):977-986.
- 194 Rankin P, Morton DP, Diehl H, Gobble J, Morey P, Chang E. Effectiveness of a volunteer-delivered lifestyle modification program for reducing cardiovascular disease risk factors. Am J Cardiol 2012;109(1):82-86.

- 195 O'Hara BJ, Phongsavan P, Eakin EG, Develin E, Smith J, Greenaway M, Bauman AE. <u>Effectiveness of Australia's</u> <u>Get Healthy Information and Health coaching Service:</u> <u>maintenance of self-reported anthropometric and</u> <u>behavioural changes after program completion.</u> *BMC Public Health* 2013;13:175.
- 196 Mettler EA, Preston HR, Jenkins SM, Lackore KA, Werneburg BL, Larson BG, Bradley KL, Warren BA, Olsen KD, Hagen PT, Vickers KS, Clark MM. <u>Motivational</u> improvements for health behavior change from wellness health coaching. Am J Health Behav 2014;38(1):83-91.
- 197 Bowden DE. <u>Employee weight management through</u> <u>health coaching.</u> Eat Weight Disord 2010;15(1-2):e52e59.
- 198 Holtrop JS, Dosh SA, Torres T, Thum YM. <u>The</u> <u>community health educator referral liaison (CHERL): a</u> <u>primary care practice role for promoting healthy</u> <u>behaviors</u>, *Am J Prev Med* 2008;35(5 Suppl):S365-S372.
- 199 Aoun S, Osseiran-Moisson R, Collins F, Newton R, Newton M. <u>A self-management concept for men at the</u> <u>community level: the 'waist' disposal challenge</u>. *J Health Psychol* 2009;14(5):663-674.
- 200 Appel LJ, Clark JM, Yeh HC, Wang NY, Coughlin JW, Daumit G, Miller ER 3rd, Dalcin A, Jerome GJ, Geller S, Noronha G, Pozefsky T, Charleston J, Reynolds JB, Durkin N, Rubin RR, Louis TA, Brancati FL. <u>Comparative effectiveness of weight-loss interventions in clinical practice. N Engl J Med 2011;365(21):1959-1968.</u>
- 201 Thomas ML, Elliott JE, Rao SM, Fahey KF, Paul SM, Miaskowski C. <u>A randomized, clinical trial of education</u> or motivational-interviewing-based health coaching compared to usual care to improve cancer pain management. *Oncol Nurs Forum* 2012;39(1):39-49.
- 202 Bray K, Turpin RS, Jungkind K, Heuser G. <u>Defining</u> success in diabetes disease management: digging deeper in the data. *Dis Manag* 2008;11(2):119-128.
- 203 Wolever RQ, Dreusicke M, Fikkan J, Hawkins TV, Yeung S, Wakefield J, Duda L, Flowers P, Cook C, Skinner E. <u>Integrative health coaching for patients with type 2</u> <u>diabetes: a randomized clinical trial.</u> *Diabetes Educ* 2010;36(4):629-639.
- 204 Hovell MF, Wahlgren DR, Liles S, Jones JA, Hughes SC, Matt GE, Ji M, Lessov-Schlaggar CN, Swan GE, Chatfield D, Ding D. <u>Providing health coaching and cotinine</u> <u>results to preteens to reduce their secondhand smoke</u> <u>exposure: a randomized trial. Chest 2011;140(3):681-689.</u>
- 205 Allen JK, Dennison-Himmelfarb CR, Szanton SL, Bone L, Hill MN, Levine DM, West M, Barlow A, Lewis-Boyer L, Donnelly-Strozzo M, Curtis C, Anderson K. <u>Community</u> <u>Outreach and Cardiovascular Health (COACH) trial: a</u> <u>randomized, controlled trial of nurse</u> <u>practitioner/community health worker cardiovascular</u> <u>disease risk reduction in urban community health</u> <u>centers.</u> *Circ Cardiovasc Qual Outcomes* 2011;4(6):595-602.
- 206 Sjöquist ES, Brodin N, Lampa J, Jensen I, Opava CH. <u>Physical activity health coaching of patients with</u> <u>rheumatoid arthritis in everyday practice: a long-term</u> <u>follow-up</u>. *Musculoskeletal Care* 2011;9(2):75-85.
- 207 Robling M, McNamara R, Bennert K, Butler CC, Channon S, Cohen D, Crowne E, Hambly H, Hawthorne K, Hood K, Longo M, Lowes L, Pickles T, Playle R, Rollnick S, Thomas-Jones E, Gregory JW. <u>The effect of the Talking Diabetes consulting skills intervention on glycaemic control and quality of life in children with type 1. diabetes: cluster randomised controlled trial (DEPICTED study). BMJ 2012;344:e2359.</u>

- 208 McDonald DD, Walsh S, Vergara C, Gifford T, Weiner DK. <u>The effect of a Spanish virtual pain coach for older</u> <u>adults: a pilot study.</u> Pain Med 2012;13(11):1397-1406.
- 209 Jaser SS, Whittemore R, Chao A, Jeon S, Faulkner MS, Grey M. <u>Mediators of 12-month outcomes of two</u> <u>internet interventions for youth with type 1 diabetes.</u> J Pediatr Psychol 2014;39(3):306-315.
- 210 Jansink R, Braspenning J, Keizer E, van der Weijden T, Elwyn G, Grol R. <u>No identifiable Hb1Ac or lifestyle</u> <u>change after a comprehensive diabetes programme</u> <u>including motivational interviewing: a cluster</u> <u>randomised trial.</u> *Scand J Prim Health Care* 2013;31(2):119-127.
- 211 Heinrich E, Candel MJ, Schaper NC, de Vries NK. <u>Effect</u> <u>evaluation of a motivational interviewing based</u> <u>counselling strategy in diabetes care</u>. *Diabetes Res Clin Pract* 2010;90(3):270-278.
- 212 Frosch DL, Uy V, Ochoa S, Mangione CM. <u>Evaluation of a</u> <u>behavior support intervention for patients with poorly</u> <u>controlled diabetes.</u> *Arch Intern Med* 2011;171(22):2011-2017.
- 213 Elbers NA, Akkermans AJ, Cuijpers P, Bruinvels DJ. Effectiveness of a web-based intervention for injured claimants: a randomized controlled trial. Trials 2013;14(1):227.
- 214 Dowd T, Kolcaba K, Steiner R. <u>The addition of health</u> <u>coaching to cognitive strategies: interventions for</u> <u>persons with compromised urinary bladder syndrome.</u> J Wound Ostomy Continence Nurs 2003;30(2):90-99.
- 215 Debussche X, Rollot O, Le Pommelet C, Fianu A, Le Moullec N, Régnier C, Boyer MC, Cogne M, Bakiri F, Schwager JC, Favier F. Quarterly individual outpatients lifestyle counseling after initial inpatients education on type 2 diabetes: the REDIA prev-2 randomized controlled trial in Reunion Island. Diabetes Metab 2012;38(1):46-53.
- 216 Butz AM, Matsui EC, Breysse P, Curtin-Brosnan J, Eggleston P, Diette G, Williams D, Yuan J, Bernert JT, Rand C. <u>A randomized trial of air cleaners and a health coach to improve indoor air quality for inner-city children with asthma and secondhand smoke exposure.</u> *Arch Pediatr Adolesc Med* 2011;165(8):741-748.
- 217 Ivey SL, Tseng W, Kurtovich E, Lui B, Weir RC, Liu J, Song H, Wang M, Hubbard A. <u>Evaluating a culturally</u> <u>competent health coach intervention for Chinese</u> <u>American patients with diabetes</u>. *Diabetes Spectr* 2012;25(2).
- 218 Vale MJ, Jelinek MV, Best JD, Santamaria JD. <u>Health</u> coaching patients with coronary heart disease to achieve the target cholesterol: a method to bridge the gap between evidence-based medicine and the "real world"- randomized controlled trial. J Clin Epidemiol 2002;55(3):245-252.
- 219 Quinn CC, Shardell MD, Terrin ML, Barr EA, Ballew SH, Gruber-Baldini AL. <u>Cluster-randomized trial of a mobile</u> <u>phone personalized behavioral intervention for blood</u> <u>glucose control.</u> *Diabetes Care* 2011;34(9):1934-1942.
- 220 Oliver JW, Kravitz RL, Kaplan SH, Meyers FJ. Individualized patient education and health coaching to improve pain control among cancer outpatients. J Clin Oncol 2001;19(8):2206-2212.
- 221 Margolius D, Bodenheimer T, Bennett H, Wong J, Ngo V, Padilla G, Thom DH. <u>Health coaching to improve</u> <u>hypertension treatment in a low-income, minority</u> <u>population.</u> *Ann Fam Med* 2012;10(3):199-205.

- 222 Grey M, Jaser SS, Holl MG, Jefferson V, Dziura J, Northrup V. <u>A multifaceted school-based intervention to</u> reduce risk for type 2 diabetes in at-risk youth. *Prev Med* 2009;49(2-3):122-128.
- 223 Allen JK, Stephens J, Dennison Himmelfarb CR, Stewart KJ, Hauck S. <u>Randomized controlled pilot study testing</u> <u>use of smartphone technology for obesity treatment</u>. J *Obes* (Published online December 2013).
- 224 Navicharern R, Aungsuroch Y, Thanasilp S. <u>Effects of</u> <u>multifaceted nurse-health coaching intervention on</u> <u>diabetic complications and satisfaction of persons with</u> <u>type 2 diabetes.</u> J Med Assoc Thai 2009;92(8):1102-1112.
- 225 Shibayama T, Kobayashi K, Takano A, Kadowaki T, Kazuma K. <u>Effectiveness of lifestyle counseling by</u> <u>certified expert nurse of Japan for non-insulin-treated</u> <u>diabetic outpatients: a 1-year randomized controlled</u> <u>trial.</u> *Diabetes Res Clin Pract* 2007;76(2):265-268.
- 226 Nguyen B, Shrewsbury VA, O'Connor J, Steinbeck KS, Hill AJ, Shah S, Kohn MR, Torvaldsen S, Baur LA. <u>Two-year outcomes of an adjunctive telephone health</u> <u>coaching and electronic contact intervention for</u> <u>adolescent weight-loss maintenance: the Loozit</u> <u>randomized controlled trial</u>. *Int J Obes* 2013;37(3):468-472.
- 227 Blackberry ID, Furler JS, Best JD, Chondros P, Vale M, Walker C, Dunning T, Segal L, Dunbar J, Audehm R, Liew D, Young D. <u>Effectiveness of general practice based</u>, practice nurse led telephone health coaching on glycaemic control of type 2 diabetes: the Patient <u>Engagement and Health coaching for Health (PEACH)</u> pragmatic cluster randomised controlled trial. *BMJ* 2013;347:f5272.
- 228 Pearson ES, Irwin JD, Morrow D, Battram DS, Melling CW. <u>The CHANGE program: comparing an interactive vs.</u> prescriptive approach to self-management among <u>university students with obesity.</u> Can J Diabetes 2013;37(1):4-11.
- 229 Benzo R, Vickers K, Ernst D, Tucker S, McEvoy C, Lorig K. <u>Development and feasibility of a self-management</u> <u>intervention for chronic obstructive pulmonary disease</u> <u>delivered with motivational interviewing strategies.</u> J <u>Cardiopulm Rehabil Prev</u> 2013;33(2):113-123.
- 230 Wentz E, Nydén A, Krevers B. <u>Development of an internet-based support and health coaching model for adolescents and young adults with ADHD and autism spectrum disorders: a pilot study. Fur Child Adolesc Psychiatry 2012;21(11):611-622.</u>
- 231 Park JH, Bae SH, Jung YS, Kim KS. <u>Quality of life and</u> symptom experience in breast cancer survivors after participating in a psychoeducational support program: a <u>pilot study</u>. *Cancer Nurs* 2012;35(1):E34-E41.
- 232 Nguyen HQ, Gill DP, Wolpin S, Steele BG, Benditt JO. <u>Pilot study of a cell phone-based exercise persistence</u> <u>intervention post-rehabilitation for COPD</u>. Int J Chron Obstruct Pulmon Dis 2009;4:301-313.
- 233 Arauz Boudreau AD, Kurowski DS, Gonzalez WI, Dimond MA, Oreskovic NM. Latino families, primary care, and childhood obesity: a randomized controlled trial, Am J Prev Med 2013;44(3 Suppl 3):S247-S257.
- 234 Menon J, Paulet M, Thomas J 3rd. <u>Wellness health</u> coaching and health-related quality of life: a casecontrol difference-in-differences analysis. J Occup Environ Med 2012;54(10):1259-1267.
- 235 Butterworth S, Linden A, McClay W, Leo MC. Effect of motivational interviewing-based health coaching on employees' physical and mental health status. J Occup Health Psychol 2006;11(4):358-365.

- 236 Lawson KL, Jonk Y, O'Connor H, Riise KS, Eisenberg DM, Kreitzer MJ. <u>The impact of telephonic health coaching</u> on health outcomes in a high-risk population. *Glob Adv Health Med* 2013;2(3):40-47.
- 237 Naik AD, White CD, Robertson SM, Armento ME, Lawrence B, Stelljes LA, Cully JA. <u>Behavioral health</u> <u>coaching for rural-living older adults with diabetes and</u> <u>depression: an open pilot of the HOPE study.</u> *BMC Geriatr* 2012;12:37.
- 238 Holland SK, Greenberg J, Tidwell L, Malone J, Mullan J, Newcomer R. <u>Community-based health coaching,</u> <u>exercise, and health service utilization.</u> J Aging Health 2005;17(6):697-716.
- 239 Gabbay RA, Añel-Tiangco RM, Dellasega C, Mauger DT, Adelman A, Van Horn DH. <u>Diabetes nurse case</u> <u>management and motivational interviewing for change</u> (<u>DYNAMIC</u>): results of a 2-year randomized controlled <u>pragmatic trial</u>, J Diabetes 2013;5(3):349-357.
- 240 Galantino ML, Schmid P, Milos A, Leonard S, Botis S, Dagan C, Albert W, Teixeira J, Mao J. Longitudinal benefits of wellness health coaching interventions for cancer survivors. Int J Interdisc Soc Sci 2009;4(10):41-58.
- 241 Bennett JA, Perrin NA, Hanson G, Bennett D, Gaynor W, Flaherty-Robb M, Joseph C, Butterworth S, Potempa K. <u>Healthy aging demonstration project: nurse health</u> <u>coaching for behavior change in older adults.</u> *Res Nurs Health* 2005;28(3):187-197.
- 242 Brook OH, van Hout H, Stalman W, Nieuwenhuyse H, Bakker B, Heerdink E, de Haan M. <u>A pharmacy-based</u> health coaching program to improve adherence to antidepressant treatment among primary care patients. *Psychiatr Serv* 2005;56(4):487-489.
- 243 Brook OH, Van Hout HP, Nieuwenhuysea H, De Haan M. Effects of health coaching by community pharmacists on psychological symptoms of antidepressant users; a randomised controlled trial. Eur Neuropsychopharmacol 2003;13(5):347-354.
- 244 DiDonato KL, May JR, Lindsey CC. <u>Impact of wellness</u> <u>health coaching and monitoring services provided in a</u> <u>community pharmacy.</u> J Am Pharm Assoc 2013;53(1):14-21.
- 245 Hawkes AL, Patrao TA, Green A, Aitken JF. <u>CanPrevent: a</u> telephone-delivered intervention to reduce multiple behavioural risk factors for colorectal cancer. *BMC Cancer* 2012;12:560.
- 246 Danino M, Shechtman Z. <u>Superiority of group</u> counseling to individual health coaching for parents of children with learning disabilities. *Psychother Res* 2012;22(5):592-603.
- 247 Stacey D, Kryworuchko J, Belkora J, Davison B, Durand MA, Eden KB, Hoffman AS, Koerner M, Légaré F, Loiselle MC, Street RL. <u>Health coaching and guidance with patient decision aids: a review of theoretical and empirical evidence.</u> BMC Med Inform Decis Mak 2013;13 Suppl 2:S11.
- 248 Pennington M, Visram S, Donaldson C, White M, Lhussier M, Deane K, Forster N, Carr SM. <u>Cost-effectiveness of health-related lifestyle advice delivered</u> by peer or lay advisors: synthesis of evidence from a systematic review, Cost Eff Resour Alloc 2013;11(1):30.
- 249 Wennberg DE, Marr A, Lang L, O'Malley S, Bennett G. <u>A</u> randomized trial of a telephone care-management strategy, N Engl J Med 2010;363(13):1245-1255.
- 250 Veroff DR, Ochoa-Arvelo T, Venator B. <u>A randomized</u> study of telephonic care support in populations at risk for musculoskeletal preference-sensitive surgeries. *BMC Med Inform Decis Mak* 2013;13:21.

- 251 Wertz D, Hou L, DeVries A, Dupclay L Jr, McGowan F, Malinowski B, Cziraky MJ. <u>Clinical and economic</u> <u>outcomes of the Cincinnati pharmacy health coaching</u> <u>program for diabetes and hypertension</u>. *Manag Care* 2012;21(3):44-54.
- 252 Schwartz J. <u>Wellness health coaching for obesity: a case</u> report. *Glob Adv Health Med* 2013;2(4):68-70.
- 253 Fera T, Bluml BM, Ellis WM. <u>Diabetes ten city challenge:</u> <u>final economic and clinical results.</u> J Am Pharm Assoc 2009;49(3):383-391.
- 254 Naydeck BL, Pearson JA, Ozminkowski RJ, Day BT, Goetzel RZ. <u>The impact of the highmark employee</u> wellness programs on 4-year healthcare costs. J Occup Environ Med 2008;50(2):146-156.
- 255 Long DA, Sheehan P. <u>A case study of population health</u> improvement at a Midwest regional hospital employer. *Popul Health Manag* 2010;13(3):163-173.
- 256 Davis L, Loyo K, Glowka A, Schwertfeger R, Danielson L, Brea C, Easton A, Griffin-Blake S. <u>A comprehensive</u> worksite wellness program in Austin, Texas: partnership between steps to a healthier Austin and capital metropolitan transportation authority. *Prev Chronic Dis* 2009;6(2):A60.
- 257 Saleh SS, Alameddine MS, Hill D, Darney-Beuhler J, Morgan A. <u>The effectiveness and cost-effectiveness of a</u> <u>rural employer-based wellness program.</u> J Rural Health 2010;26(3):259-265.
- 258 Parry C, Min SJ, Chugh A, Chalmers S, Coleman EA. <u>Further application of the care transitions intervention:</u> <u>results of a randomized controlled trial conducted in a</u> <u>fee-for-service setting</u>. Home Health Care Serv Q 2009;28(2-3):84-99.
- 259 Coleman EA, Parry C, Chalmers S, Min SJ. <u>The care</u> <u>transitions intervention: results of a randomized</u> <u>controlled trial</u>. *Arch Intern Med* 2006;166(17):1822-1828.
- 260 Pomerantz JI, Toney SD, Hill ZJ. <u>Care health coaching: an alternative approach to managing comorbid depression</u>. *Prof Case Manag* 2010;15(3):137-142.
- 261 Steventon A, Tunkel S, Blunt I, Bardsley M. <u>Effect of</u> telephone health coaching (Birmingham OwnHealth) on hospital use and associated costs: cohort study with matched controls. *BMJ* 2013;347:f4585.
- 262 Cate H, Bhattacharya D, Clark A, Fordham R, Holland R, Broadway DC. <u>Improving adherence to glaucoma</u> <u>medication: a randomised controlled trial of a patient-</u> <u>centred intervention (The Norwich Adherence Glaucoma</u> <u>Study)</u>, *BMC Ophthalmol* 2014;14(1):32.
- 263 Turkstra E, Hawkes AL, Oldenburg B, Scuffham PA. Costeffectiveness of a coronary heart disease secondary prevention program in patients with myocardial infarction: results from a randomised controlled trial (ProActive Heart). BMC Cardiovasc Disord 2013;13(1):33.
- 264 van Dongen JM, Strijk JE, Proper KI, van Wier MF, van Mechelen W, van Tulder MW, van der Beek AJ. <u>A costeffectiveness and return-on-investment analysis of a</u> worksite vitality intervention among older hospital workers: results of a randomized controlled trial. J Occup Environ Med 2013;55(3):337-346.
- 265 Postmus D, Pari AA, Jaarsma T, Luttik ML, van Veldhuisen DJ, Hillege HL, Buskens E. <u>A trial-based</u> economic evaluation of 2 nurse-led disease management programs in heart failure. *Am Heart J* 2011;162(6):1096-1104.

- 266 Bosmans JE, Brook OH, van Hout HP, de Bruijne MC, Nieuwenhuyse H, Bouter LM, Stalman WA, van Tulder MW. Cost effectiveness of a pharmacy-based health coaching programme to improve adherence to antidepressants. Pharmacoeconomics 2007;25(1):25-37.
- 267 Hersey JC, Khavjou O, Strange LB, Atkinson RL, Blair SN, Campbell S, Hobbs CL, Kelly B, Fitzgerald TM, Kish-Doto J, Koch MA, Munoz B, Peele E, Stockdale J, Augustine C, Mitchell G, Arday D, Kugler J, Dorn P, Ellzy J, Julian R, Grissom J, Britt M. <u>The efficacy and cost-effectiveness of a community weight management intervention: a</u> <u>randomized controlled trial of the health weight</u> <u>management demonstration</u>. *Prev Med* 2012;54(1):42-49.
- 268 Nelson KA, Highstein GR, Garbutt J, Trinkaus K, Fisher EB, Smith SR, Strunk RC. <u>A randomized controlled trial</u> of parental asthma health coaching to improve outcomes among urban minority children. Arch Pediatr Adolesc Med 2011;165(6):520-526.
- 269 Lin WC, Chien HL, Willis G, O'Connell E, Rennie KS, Bottella HM, Ferris TG. <u>The effect of a telephone-based</u> <u>health coaching disease management program on</u> <u>Medicaid members with chronic conditions.</u> *Med Care* 2012;50(1):91-98.
- 270 Abildsnes E, Walseth LT, Flottorp SA, Stensland PS. Power and powerlessness: GPs' narratives about lifestyle counselling. Br J Gen Pract 2012;62(596):e160-e166.
- 271 Manderson B, McMurray J, Piraino E, Stolee P. Navigation roles support chronically ill older adults through healthcare transitions: a systematic review of the literature. Health Soc Care Community 2012;20(2):113-127.
- 272 Sorbi MJ, Mak SB, Houtveen JH, Kleiboer AM, van Doornen LJ. <u>Mobile web-based monitoring and health</u> <u>coaching: feasibility in chronic migraine</u>. J Med Internet Res 2007;9(5):e38.
- 273 Fico G, Fioravanti A, Arredondo MT, Ardigó D, Guillén A. <u>A healthy lifestyle health coaching-persuasive</u> <u>application for patients with type 2 diabetes.</u> Conf Proc IEEE Eng Med Biol Soc 2010;2010:2221-2224.
- 274 Gatti F, Brivio E, Galimberti C. <u>Evaluation of a personal</u> <u>mobile health coaching service for health tracking.</u> *Stud Health Technol Inform* 2013;191:154-157.
- 275 Simkin-Silverman LR, Conroy MB, Bhargava T, McTigue KM. <u>Development of an online diabetes prevention</u> <u>lifestyle intervention health coaching protocol for use in</u> <u>primary care practice.</u> *Diabetes Educ* 2011;37(2):263-268.
- 276 Spassova L, Vittore D, Droste D, Rösch N. <u>Automated</u> <u>lifestyle health coaching for cerebro-cardiovascular</u> <u>disease prevention</u>, *Stud Health Technol Inform* 2013;190:234-236.
- 277 Allen M, Iezzoni LI, Huang A, Huang L, Leveille SG. <u>Improving patient-clinician communication about</u> <u>chronic conditions: description of an internet-based</u> <u>nurse E-coach intervention.</u> Nurs Res 2008;57(2):107-112.
- 278 Patrick K, Marshall SJ, Davila EP, Kolodziejczyk JK, Fowler JH, Calfas KJ, Huang JS, Rock CL, Griswold WG, Gupta A, Merchant G, Norman GJ, Raab F, Donohue MC, Fogg BJ, Robinson TN. <u>Design and implementation of a</u> <u>randomized controlled social and mobile weight loss</u> <u>trial for young adults (project SMART).</u> Contemp Clin Trials 2014;37(1):10-18.
- 279 Skolasky RL, Riley LH 3rd, Maggard AM, Bedi S, Wegener ST. <u>Functional recovery in lumbar spine</u> <u>surgery: a controlled trial of health behavior change</u> <u>counseling to improve outcomes.</u> Contemp Clin Trials 2013;36(1):207-217.

- 280 Donelan K, Rao SR, Rogers RS, Mailhot JR, Galvin R. Experience with health coach-mediated physician referral in an employed insured population. J Gen Intern Med 2010;25(10):1071-1077.
- 281 Adelman AM, Graybill M. Integrating a health coach into primary care: reflections from the Penn state ambulatory research network. Ann Fam Med 2005;3 Suppl 2:S33-S35.
- 282 Appel LJ, Clark JM, Yeh HC, Wang NY, Coughlin JW, Daumit G, Miller ER 3rd, Dalcin A, Jerome GJ, Geller S, Noronha G, Pozefsky T, Charleston J, Reynolds JB, Durkin N, Rubin RR, Louis TA, Brancati FL. <u>Comparative effectiveness of weight-loss interventions in clinical practice. N Engl J Med 2011;365(21):1959-1968.</u>
- 283 Yan T, Wilber KH, Simmons WJ. <u>Motivating high-risk</u> <u>older adults to exercise: does health coaching matter?</u> *Home Health Care Serv Q* 2011;30(2):84-95.
- 284 Long DA. <u>Wellness program satisfaction, sustained</u> <u>health coaching participation, and achievement of</u> <u>health goals</u>. J Occup Environ Med 2012;54(5):592-597.
- 285 van Wier MF, Dekkers JC, Hendriksen IJ, Heymans MW, Ariëns GA, Pronk NP, Smid T, van Mechelen W. Effectiveness of phone and e-mail lifestyle counseling for long term weight control among overweight employees. J Occup Environ Med 2011;53(6):680-686.
- 286 Reinke LF, Griffith RG, Wolpin S, Donesky-Cuenco D, Carrieri-Kohlman V, Nguyen HQ. Feasibility of a webinar for health coaching patients with chronic obstructive pulmonary disease on end-of-life communication. Am J Hosp Palliat Care 2011;28(3):147-152.
- 287 Terry PE, Seaverson EL, Staufacker MJ, Gingerich SB. <u>A</u> comparison of the effectiveness of a telephone health coaching program and a mail-based program. *Health Educ Behav* 2010;37(6):895-912.
- 288 Samarel N, Fawcett J. <u>Enhancing adaptation to breast</u> cancer: the addition of health coaching to support groups. Oncol Nurs Forum 1992;19(4):591-596.
- 289 Nolan RP, Upshur RE, Lynn H, Crichton T, Rukholm E, Stewart DE, Alter DA, Chessex C, Harvey PJ, Grace SL, Picard L, Michel I, Angus J, Corace K, Barry-Bianchi SM, Chen MH. <u>Therapeutic benefit of preventive telehealth</u> <u>counseling in the community outreach heart health and</u> <u>risk reduction trial</u>. *Am J Cardiol* 2011;107(5):690-696.
- 290 Keenan S, King G, Curran CJ, McPherson A. <u>Effectiveness</u> of experiential life skills health coaching for youth with a <u>disability</u>. *Phys Occup Ther Pediatr* (Published online October 2013).
- 291 Befort CA, Donnelly JE, Sullivan DK, Ellerbeck EF, Perri MG. <u>Group versus individual phone-based obesity</u> <u>treatment for rural women.</u> *Eat Behav* 2010;11(1):11-17.
- 292 Danino M, Shechtman Z. <u>Superiority of group</u> <u>counseling to individual health coaching for parents of</u> <u>children with learning disabilities</u>. *Psychother Res* 2012;22(5):592-603.
- 293 Hoddinott P, Chalmers M, Pill R. <u>One-to-one or groupbased peer support for breastfeeding? Women's</u> perceptions of a breastfeeding peer health coaching intervention. *Birth* 2006;33(2):139-146.
- 294 Wolever RQ, Simmons LA, Sforzo GA, Dill D, Kaye M, Bechard EM, Southard ME, Kennedy M, Vosloo J, Yang N. <u>A systematic review of the literature on health and</u> wellness health coaching: defining a key behavioral intervention in healthcare. Glob Adv Health Med 2013;2(4):38-57.
- 295 Holland SK, Greenberg J, Tidwell L, Newcomer R. <u>Preventing disability through community-based health</u> <u>coaching.</u> J Am Geriatr Soc 2003;51(2):265-269.

- 296 Pryor J. <u>Health coaching patients to self-care: a primary responsibility of nursing.</u> Int J Older People Nurs 2009;4(2):79-88.
- 297 Snyder EH, Lawrence CN, Weatherholt TN, Nagy P. <u>The benefits of motivational interviewing and health coaching for improving the practice of comprehensive family assessments in child welfare.</u> Child Welfare 2012;91(5):9-36.
- 298 Kettunen T, Poskiparta M, Kiuru P, Kasila K. <u>Lifestyle</u> counseling in type 2 diabetes prevention: a case study of a nurse's communication activity to produce change talk. Commun Med 2006;3(1):3-14.
- 299 Noordman J, Koopmans B, Korevaar JC, van der Weijden T, van Dulmen S. <u>Exploring lifestyle counselling in</u> routine primary care consultations: the professionals' role, Fam Pract 2013;30(3):332-340.
- 300 Domian EW, Baggett KM, Carta JJ, Mitchell S, Larson E. Factors influencing mothers' abilities to engage in a comprehensive parenting intervention program. Public Health Nurs 2010;27(5):399-407.
- 301 Lok KY, Chan RS, Sea MM, Woo J. <u>Nutritionist's variation</u> in counseling style and the effect on weight change of patients attending a community based lifestyle modification program. Int J Environ Res Public Health 2010;7(2):413-426.
- 302 Parry C, Kramer HM, Coleman EA. <u>A qualitative</u> <u>exploration of a patient-centered health coaching</u> <u>intervention to improve care transitions in chronically ill</u> <u>older adults.</u> *Home Health Care Serv Q* 2006;25(3-4):39-53.
- 303 Walseth LT, Abildsnes E, Schei E. <u>Patients' experiences</u> with lifestyle counselling in general practice: a <u>qualitative study</u>. Scand J Prim Health Care 2011;29(2):99-103.
- 304 Hyman DJ, Pavlik VN, Taylor WC, Goodrick GK, Moye L. <u>Simultaneous vs sequential counseling for multiple</u> <u>behavior change.</u> Arch Intern Med 2007;167(11):1152-1158.
- 305 Shahnazari M, Ceresa C, Foley S, Fong A, Zidaru E, Moody S. <u>Nutrition-focused wellness health coaching</u> <u>promotes a reduction in body weight in overweight US</u> <u>veterans.</u> J Acad Nutr Diet 2013;113(7):928-935.
- 306 Adams SR, Goler NC, Sanna RS, Boccio M, Bellamy DJ, Brown SD, Neugebauer RS, Ferrara A, Wiley DM, Schmittdiel JA. <u>Patient satisfaction and perceived</u> success with a telephonic health coaching program: the <u>Natural Experiments for Translation in Diabetes (NEXT-D) study, Northern California, 2011, Prev Chronic Dis</u> 2013;10:E179.
- 307 Nishita C, Cardazone G, Uehara DL, Tom T. Empowered diabetes management: life health coaching and pharmacist counseling for employed adults with diabetes. Health Educ Behav 2013;40(5):581-591.
- 308 Margolius D, Bodenheimer T, Bennett H, Wong J, Ngo V, Padilla G, Thom DH. <u>Health coaching to improve</u> <u>hypertension treatment in a low-income, minority</u> <u>population</u>, Ann Fam Med 2012;10(3):199-205.
- 309 Coberley CR, McGinnis M, Orr PM, Coberley SS, Hobgood A, Hamar B, Gandy B, Pope J, Hudson L, Hara P, Shurney D, Clarke JL, Crawford A, Goldfarb NI. <u>Association between frequency of telephonic contact</u> and clinical testing for a large, geographically diverse <u>diabetes disease management population</u>. *Dis Manag* 2007;10(2):101-109.
- 310 Samarel N, Fawcett J, Tulman L. <u>The effects of health</u> <u>coaching in breast cancer support groups: a pilot study.</u> *Oncol Nurs Forum* 1993;20(5):795-798.

- 311 Seghers J, Van Hoecke AS, Schotte A, Opdenacker J, Boen F. <u>The added value of a brief self-efficacy health</u> <u>coaching on the effectiveness of a 12-week physical</u> <u>activity program.</u> J Phys Act Health 2014;11(1):18-29.
- 312 Grey M, Whittemore R, Jeon S, Murphy K, Faulkner MS, Delamater A. <u>Internet psycho-education programs</u> <u>improve outcomes in youth with type 1 diabetes</u>. *Diabetes Care* 2013;36(9):2475-2482.
- 313 Tucker LA, Cook AJ, Nokes NR, Adams TB. <u>Telephone-based diet and exercise health coaching and a weight-loss supplement result in weight and fat loss in 120 men and women</u>, Am J Health Promot 2008;23(2):121-129.
- 314 Yen L, Edington MP, McDonald T, Hirschland D, Edington DW. <u>Changes in health risks among the</u> participants in the united auto workers - general motors <u>lifesteps health promotion program</u>. *Am J Health Promot* 2001;16(1):7-15.
- 315 Jefferson V, Jaser SS, Lindemann E, Galasso P, Beale A, Holl MG, Grey M. <u>Coping skills training in a telephone</u> health coaching program for youth at risk for type 2 <u>diabetes</u>. J Pediatr Health Care 2011;25(3):153-161.
- 316 Basak Cinar A, Schou L. <u>The role of self-efficacy in health</u> <u>coaching and health education for patients with type 2</u> <u>diabetes</u>. *Int Dent J* (Published online February 2014).
- 317 van der Wulp I, de Leeuw JR, Gorter KJ, Rutten GE. Effectiveness of peer-led self-management health coaching for patients recently diagnosed with type 2 diabetes mellitus in primary care: a randomized controlled trial, Diabet Med 2012;29(10):e390-e397.
- 318 Iles RA, Taylor NF, Davidson M, O'Halloran P. <u>An</u> effective health coaching intervention for people with low recovery expectations and low back pain: a content analysis. J Back Musculoskelet Rehabil 2014;27(1):93-101
- 319 Terry PE, Seaverson EL, Staufacker MJ, Tanaka A. <u>The</u> <u>effectiveness of a telephone-based tobacco cessation</u> <u>program offered as part of a worksite health promotion</u> <u>program.</u> *Popul Health Manag* 2011;14(3):117-125.
- 320 Bowden DE. Employee weight management through health coaching. Eat Weight Disord 2010;15(1-2):e52e59.
- 321 Berna JS. <u>Wellness health coaching outcomes in a case</u> report of a diabetic native american male. *Glob Adv Health Med* 2013;2(4):62-67.
- 322 Dewing S, Mathews C, Schaay N, Cloete A, Louw J, Simbayi L. <u>Behaviour change counselling for ARV</u> adherence support within primary health care facilities in the Western Cape, South Africa. AIDS Behav 2012;16(5):1286-1294.
- 323 Terry PE, Fowles JB, Harvey L. <u>Employee engagement</u> factors that affect enrollment compared with retention in two health coaching programs - the ACTIVATE study. *Popul Health Manag* 2010;13(3):115-122.
- 324 Basak Cinar A, Schou L. <u>Health promotion for patients</u> with diabetes: health coaching or formal health education? Int Dent J 2014;64(1):20-28.
- 325 Rankin P, Morton DP, Diehl H, Gobble J, Morey P, Chang E. Effectiveness of a volunteer-delivered lifestyle modification program for reducing cardiovascular disease risk factors. Am J Cardiol 2012;109(1):82-86.
- 326 Holland SK, Greenberg J, Tidwell L, Malone J, Mullan J, Newcomer R. <u>Community-based health coaching</u>, <u>exercise</u>, and health service utilization. J Aging Health 2005;17(6):697-716.
- 327 Lawson KL, Jonk Y, O'Connor H, Riise KS, Eisenberg DM, Kreitzer MJ. <u>The impact of telephonic health coaching</u> on health outcomes in a high-risk population. *Glob Adv Health Med* 2013;2(3):40-47.

- 328 Moskowitz D, Thom DH, Hessler D, Ghorob A, Bodenheimer T. <u>Peer health coaching to improve</u> <u>diabetes self-management: which patients benefit</u> <u>most?</u> J Gen Intern Med 2013;28(7):938-942.
- 329 Bowden DE. <u>Employee weight management through health coaching</u>. *Eat Weight Disord* 2010;15(1-2):e52-e59.
- 330 Al-Sinani M, Min Y, Ghebremeskel K, Qazaq HS. Effectiveness of and adherence to dietary and lifestyle counselling: effect on metabolic control in type 2 diabetic Omani patients. Sultan Qaboos Univ Med J 2010;10(3):341-349.
- 331 Coberley CR, McGinnis M, Orr PM, Coberley SS, Hobgood A, Hamar B, Gandy B, Pope J, Hudson L, Hara P, Shurney D, Clarke JL, Crawford A, Goldfarb NI. <u>Association between frequency of telephonic contact</u> and clinical testing for a large, geographically diverse <u>diabetes disease management population</u>. *Dis Manag* 2007;10(2):101-109.
- 332 Rabius V, McAlister AL, Geiger A, Huang P, Todd R. <u>Telephone counseling increases cessation rates among</u> young adult smokers. *Health Psychol* 2004;23(5):539-541.
- 333 Long DA. <u>Wellness program satisfaction, sustained</u> health coaching participation, and achievement of health goals, J Occup Environ Med 2012;54(5):592-597.
- 334 Toth-Capelli KM, Brawer R, Plumb J, Daskalakis C. <u>Stage of change and other predictors of participant retention in a behavioural weight management program in primary care. Health Promot Pract 2013;14(3):441-450.</u>
- 335 Dennis SM, Harris M, Lloyd J, Powell Davies G, Faruqi N, Zwar N. <u>Do people with existing chronic conditions</u> <u>benefit from telephone health coaching?</u> A rapid review. *Aust Health Rev* 2013;37(3):381-388.
- 336 Brook OH, Van Hout HP, Nieuwenhuysea H, De Haan M. Effects of health coaching by community pharmacists on psychological symptoms of antidepressant users; a randomised controlled trial. Eur Neuropsychopharmacol 2003;13(5):347-354.
- 337 Rohrer JE, Naessens JM, Liesinger J, Tulledge-Scheitel S, Vanhouten H. <u>A telephonic health coaching program</u> <u>has more impact when body mass index is over 35.</u> Obes Res Clin Pract 2010;4(1):e1-e82.
- 338 Sjöquist ES, Almqvist L, Asenlöf P, Lampa J, Opava CH. <u>Physical-activity health coaching and health status in</u> <u>rheumatoid arthritis: a person-oriented approach.</u> *Disabil Rehabil* 2010;32(10):816-825.
- 339 Merrill RM, Bowden DE, Aldana SG. Factors associated with attrition and success in a worksite wellness telephonic health coaching program. Educ Health 2010;23(3):385.
- 340 Powell-Wiley TM, Ayers CR, Banks-Richard K, Berry JD, Khera A, Lakoski SG, McGuire DK, de Lemos JA, Das SR. <u>Disparities in counseling for lifestyle modification</u> <u>among obese adults: insights from the Dallas heart</u> <u>study.</u> Obesity 2012;20(4):849-855.
- 341 Anderson KM, Siems LV, Holloway SC, Sultana N, Braund WE, Harris LM. <u>Group counselling improves quality for</u> <u>patients with limited health literacy.</u> *Qual Prim Care* 2012;20(1):5-13.
- 342 Bedrosian RC, Striegel-Moore RH, Wang C. Demographic and clinical characteristics of individuals utilizing an internet-based digital health coaching program for recovering from binge eating. Int J Eat Disord 2011;44(7):639-646.

- 343 Grossmeier J. <u>The influence of worksite and employee</u> variables on employee engagement in telephonic health coaching programs: a retrospective multivariate analysis. *Am J Health Promot* 2013;27(3):e69-e80.
- 344 Balmford J, Borland R, Benda P. Patterns of use of an automated interactive personalized health coaching program for smoking cessation. J Med Internet Res 2008;10(5):e54.
- 345 O'Hara BJ, Phongsavan P, Venugopal K, Bauman AE. <u>Characteristics of participants in Australia's Get Healthy</u> <u>telephone-based lifestyle information and health</u> <u>coaching service: reaching disadvantaged communities</u> <u>and those most at need. Health Educ Res</u> 2011;26(6):1097-1106.
- 346 Venditti EM, Wylie-Rosett J, Delahanty LM, Mele L, Hoskin MA, Edelstein SL. <u>Short and long-term lifestyle</u> <u>health coaching approaches used to address diverse</u> <u>participant barriers to weight loss and physical activity</u> <u>adherence.</u> *Int J Behav Nutr Phys Act* 2014;11:16.
- 347 Shaw R, Gillies M, Barber J, MacIntyre K, Harkins C, Findlay IN, McCloy K, Gillie A, Scoular A, MacIntyre PD. <u>Pre-exercise screening and health coaching in CHD</u> <u>secondary prevention: a qualitative study of the patient</u> <u>experience.</u> *Health Educ Res* 2012;27(3):424-436.
- 348 Noordman J, de Vet E, van der Weijden T, van Dulmen S. <u>Motivational interviewing within the different stages of</u> <u>change: an analysis of practice nurse-patient</u> <u>consultations aimed at promoting a healthier lifestyle.</u> *Soc Sci Med* 2013;87:60-67.
- 349 O'Hara BJ, Bauman AE, Phongsavan P. <u>Using mass-</u> media communications to increase population usage of <u>Australia's get healthy information and health coaching</u> <u>service.</u> BMC Public Health 2012;12:762.
- 350 O'Hara BJ, Bauman AE, King EL, Phongsavan P. <u>Process</u> evaluation of the advertising campaign for the NSW Get <u>Healthy information and health coaching service</u>. *Health Promot J Austr* 2011;22(1):68-71.
- 351 Gingerich SB, Anderson DR, Koland H. Impact of financial incentives on behavior change program participation and risk reduction in worksite health promotion. Am J Health Promot 2012;27(2):119-122.
- 352 Kalauokalani D, Franks P, Oliver JW, Meyers FJ, Kravitz RL. <u>Can patient health coaching reduce racial/ethnic</u> <u>disparities in cancer pain control? Secondary analysis of</u> <u>a randomized controlled trial</u>, *Pain Med* 2007;8(1):17-24.
- 353 Sathe A, Flowers E, Mathur A, Garcia DM, Kotrys J, Gandhi R, Molina C, Mathur A. <u>A culturally specific</u> health coaching program targeting cardiovascular disease risk in South Asians: rationale. design, and baseline data. Ethn Dis 2013;23(3):304-309.
- 354 Ratanawongsa N, Handley MA, Sarkar U, Quan J, Pfeifer K, Soria C, Schillinger D. <u>Diabetes health information</u> <u>technology innovation to improve quality of life for</u> <u>health plan members in urban safety net.</u> J Ambul Care Manage 2014;37(2):127-37.
- 355 Frosch DL, Uy V, Ochoa S, Mangione CM. <u>Evaluation of a behavior support intervention for patients with poorly controlled diabetes</u>. Arch Intern Med 2011;171(22):2011-2017.
- 356 Ivey SL, Tseng W, Kurtovich E, Lui B, Weir RC, Liu J, Song H, Wang M, Hubbard A. <u>Evaluating a culturally</u> <u>competent health coach intervention for Chinese</u> <u>American patients with diabetes.</u> *Diabetes Spectr* 2012;25(2).

- 357 Hohenadel J, Kaegi E, Laidlaw J, Kovacik G, Cortinois A, Kang R, Jadad AR. Leveling the playing field: the personal coach program as an innovative approach to assess and address the supportive care needs of underserved cancer patients. J Support Oncol 2007;5(4):185-193.
- 358 Jordan M. <u>Health coaching for the underserved</u>. *Glob Adv Health Med* 2013;2(3):75-82.
- 359 Al-Sinani M, Min Y, Ghebremeskel K, Qazaq HS. Effectiveness of and adherence to dietary and lifestyle counselling: effect on metabolic control in type 2 diabetic Omani patients. Sultan Qaboos Univ Med J 2010;10(3):341-349.
- 360 Lambe B, Collins C. <u>A qualitative study of lifestyle</u> <u>counselling in general practice in Ireland.</u> Fam Pract 2010;27(2):219-223.
- 361 Tang TS, Nwankwo R, Whiten Y, Oney C. <u>Training peers</u> to deliver a church-based diabetes prevention program. *Diabetes Educ* 2012;38(4):519-525.
- 362 Petersen I, Fairall L, Egbe CO, Bhana A. <u>Optimizing lay</u> <u>counsellor services for chronic care in South Africa: a</u> <u>qualitative systematic review.</u> Patient Educ Couns (Published online February 2014).
- 363 Thom DH, Ghorob A, Hessler D, De Vore D, Chen E, Bodenheimer TA. <u>Impact of peer health coaching on</u> <u>glycemic control in low-income patients with diabetes: a</u> <u>randomized controlled trial.</u> Ann Fam Med 2013;11(2):137-144.
- 364 Goldman ML, Ghorob A, Eyre SL, Bodenheimer T. <u>How</u> <u>do peer coaches improve diabetes care for low-income</u> <u>patients?: A qualitative analysis.</u> *Diabetes Educ* 2013;39(6):800-810.
- 365 Rogers EA, Hessler DM, Bodenheimer TS, Ghorob A, Vittinghoff E, Thom DH. <u>Diabetes peer health coaching:</u> <u>do "better patients" make better coaches?</u> *Diabetes Educ* 2014;40(1):107-115.
- 366 Leahey TM, Wing RR. <u>A randomized controlled pilot</u> study testing three types of health coaches for obesity treatment: professional, peer, and mentor, Obesity 2013;21(5):928-934.
- 367 Noordman J, van der Weijden T, van Dulmen S. <u>Communication-related behavior change techniques</u> <u>used in face-to-face lifestyle interventions in primary</u> <u>care: a systematic review of the literature</u>. *Patient Educ Couns* 2012;89(2):227-244.
- 368 Lanese BS, Dey A, Srivastava P, Figler R. <u>Introducing the health coach at a primary care practice: impact on quality and cost (part 1)</u>. Hosp Top 2011;89(1):16-22.
- 369 Selvaraj FJ, Mohamed M, Omar K, Nanthan S, Kusiar Z, Subramaniam SY, Ali N, Karanakaran K, Ahmad F, Low WH. <u>The impact of a disease management program</u> (COACH) on the attainment of better cardiovascular risk control in dyslipidaemic patients at primary care centres (the DISSEMINATE study): a randomised controlled trial. BMC Fam Pract 2012;13:97.
- 370 Nessen T, Opava CH, Martin C, Demmelmaier I. <u>From</u> <u>clinical expert to guide: experiences from health</u> <u>coaching people with rheumatoid arthritis to increased</u> <u>physical activity.</u> *Phys Ther* (Published online March 2014).
- 371 Vale MJ, Jelinek MV, Best JD, Dart AM, Grigg LE, Hare DL, Ho BP, Newman RW, McNeil JJ. <u>Health coaching</u> patients On Achieving Cardiovascular Health (COACH): a multicenter randomized trial in patients with coronary heart disease. Arch Intern Med 2003;163(22):2775-2783.

- 372 Margolius D, Wong J, Goldman ML, Rouse-Iniguez J, Bodenheimer T. <u>Delegating responsibility from clinicians</u> to nonprofessional personnel: the example of <u>hypertension control.</u> J Am Board Fam Med 2012;25(2):209-215.
- 373 Robling M, McNamara R, Bennert K, Butler CC, Channon S, Cohen D, Crowne E, Hambly H, Hawthorne K, Hood K, Longo M, Lowes L, Pickles T, Playle R, Rollnick S, Thomas-Jones E, Gregory JW. <u>The effect of the Talking Diabetes consulting skills intervention on glycaemic control and quality of life in children with type 1</u> <u>diabetes: cluster randomised controlled trial (DEPICTED study)</u>, *BMJ* 2012;344:e2359.
- 374 Wolever RQ, Caldwell KL, Wakefield JP, Little KJ, Gresko J, Shaw A, Duda LV, Kosey JM, Gaudet T. <u>Integrative</u> <u>health coaching: an organizational case study.</u> *Explore* 2011;7(1):30-36.
- 375 Millenson ML, DiGioia AM 3rd, Greenhouse PK, Swieskowski D. <u>Turning patient-centeredness from ideal</u> to real: lessons from 2 success stories. J Ambul Care Manage 2013;36(4):319-334.
- 376 Dickey LL, Gemson DH, Carney P. <u>Office system</u> interventions supporting primary care-based health behavior change counseling. Am J Prev Med 1999;17(4):299-308.
- 377 Brennan C, Newton M, Wood F, Schug SA, Allsop S, Browne AL. <u>Training general practitioners in remote</u> Western Australia in a method of screening and brief intervention for harmful alcohol use: a pilot study. Aust J Rural Health 2013;21(2):72-79.
- 378 Fitzgerald N, Watson H, McCaig D, Stewart D. <u>Developing and evaluating training for community</u> pharmacists to deliver interventions on alcohol issues. *Pharm World Sci* 2009;31(2):149-153.
- 379 Newman P, McDowell A. Exploring the benefits of health coaching for clinicians and patients. HSJ 2011.
- 380 Wagner PJ, Jester DM, Moseley GC. <u>Medical students as</u> health coaches. Acad Med 2002;77(11):1164-1165.
- 381 Moore GF, Moore L, Murphy S. <u>Integration of</u> motivational interviewing into practice in the national exercise referral scheme in Wales: a mixed methods <u>study.</u> Behav Cogn Psychother 2012;40(3):313-330.
- 382 Ahluwalia S, de Silva D, Kumar S, Viney R, Chana N. <u>Teaching GP trainees to use health coaching in</u> <u>consultations with patients: evaluation of a pilot study.</u> *Educ Prim Care* 2013;24(6):418-426.
- 383 Allicock M, Haynes-Maslow L, Carr C, Orr M, Kahwati LC, Weiner BJ, Kinsinger L. <u>Training veterans to provide peer</u> <u>support in a weight-management program: MOVE!</u> Prev Chronic Dis 2013;10:E185.
- 384 Adelman AM, Graybill M. Integrating a health coach into primary care: reflections from the Penn state ambulatory research network. Ann Fam Med 2005;3 Suppl 2:S33-S35.
- 385 Tang TS, Nwankwo R, Whiten Y, Oney C. <u>Training peers</u> to deliver a church-based diabetes prevention program. *Diabetes Educ* 2012;38(4):519-525.
- 386 Dewing S, Mathews C, Cloete A, Schaay N, Simbayi L, Louw J. <u>Lay counselors' ability to deliver counseling for</u> <u>behavior change.</u> J Consult Clin Psychol 2014;82(1):19-29.
- 387 Doherty Y, Hall D, James PT, Roberts SH, Simpson J. Change counselling in diabetes: the development of a training programme for the diabetes team. Patient Educ Couns 2000;40(3):263-278.

- 388 Burton AM, Agne AA, Lehr SM, Davis NJ, Willett LL, Cherrington AL. <u>Training residents in obesity counseling:</u> incorporating principles of motivational interviewing to enhance patient centeredness. J Grad Med Educ 2011;3(3):408-411.
- 389 Hanks C, Kitzman H, Milligan R. <u>Implementing the COACH relationship model: health promotion for mothers and children</u>. ANS Adv Nurs Sci 1995;18(2):57-66.
- 390 Vickers KS, Kircher KJ, Smith MD, Petersen LR, Rasmussen NH. <u>Health behavior counseling in primary care: provider-reported rate and confidence.</u> Fam Med 2007;39(10):730-735.
- 391 Parker WA, Steyn NP, Levitt NS, Lombard CJ. <u>They think</u> they know but do they? Misalignment of perceptions of lifestyle modification knowledge among health professionals. *Public Health Nutr* 2011;14(8):1429-1438.
- 392 McAvoy BR, Kaner EF, Lock CA, Heather N, Gilvarry E. Our healthier nation: are general practitioners willing and able to deliver? A survey of attitudes to and involvement in health promotion and lifestyle counselling. Br J Gen Pract 1999;49(440):187-190.
- 393 Elwell L, Powell J, Wordsworth S, Cummins C. <u>Health</u> professional perspectives on lifestyle behaviour change in the paediatric hospital setting: a qualitative study. BMC Pediatr 2014;14(1):71.
- 394 Broers S, Smets E, Bindels P, Evertsz' FB, Calff M, de Haes H. <u>Training general practitioners in behavior change</u> <u>counseling to improve asthma medication adherence</u>. *Patient Educ Couns* 2005;58(3):279-287.
- 395 Ostlund AS, Wadensten B, Häggström E, Kristofferzon ML. <u>District nurses' and registered nurses' training in</u> and use of motivational interviewing in primary care <u>settings.</u> J Clin Nurs (Published online December 2013).
- 396 Schumacher JA, Coffey SF, Walitzer KS, Burke RS, Williams DC, Norquist G, Elkin TD. <u>Guidance for new</u> motivational interviewing trainers when training addiction professionals: findings from a survey of experienced trainers. *Motiv Interviewing* 2012;1(1):7-15.
- 397 Madson MB, Loignon AC, Lane C. <u>Training in</u> motivational interviewing: a systematic review. J Subst Abuse Treat 2009;36(1):101-109.
- 398 Baer JS, Rosengren DB, Dunn CW, Wells EA, Ogle RL, Hartzler B. <u>An evaluation of workshop training in</u> motivational interviewing for addiction and mental health clinicians. *Drug Alcohol Depend* 2004;73(1):99-106.
- 399 Brug J, Spikmans F, Aartsen C, Breedveld B, Bes R, Fereira I. <u>Training dietitians in basic motivational</u> interviewing skills results in changes in their counseling style and in lower saturated fat intakes in their patients. J Nutr Educ Behav 2007;39(1):8-12.
- 400 Cook PF, Richardson G, Wilson A. <u>Motivational</u> interviewing training to promote Head Start children's adherence to oral health care recommendations: results of a program evaluation. J Public Health Dent 2013;73(2):147-150.
- 401 Bray KK, Catley D, Voelker MA, Liston R, Williams KB. Motivational interviewing in dental hygiene education: curriculum modification and evaluation. J Dent Educ 2013;77(12):1662-1669.
- 402 Jaffray M, Matheson C, Bond CM, Lee AJ, McLernon DJ, Johnstone A, Skea L, Davidson B. <u>Does training in</u> <u>motivational interviewing for community pharmacists</u> <u>improve outcomes for methadone patients? A cluster</u> <u>randomised controlled trial</u>. *Int J Pharm Pract* 2014;22(1):4-12.

- 403 Hinz JG. <u>Teaching dental students motivational</u> interviewing techniques: analysis of a third-year class assignment. J Dent Educ 2010;74(12):1351-1356.
- 404 Wolfe H, Haller DL, Benoit E, Bolger KW, Cancienne JC, Ingersoll KS, Sharp V. <u>Developing PeerLink to engage</u> <u>out-of-care HIV+ substance users: training peers to</u> <u>deliver a peer-led motivational intervention with fidelity.</u> *AIDS Care* 2013;25(7):888-894.
- 405 Smeerdijk M, Keet R, de Haan L, Barrowclough C, Linszen D, Schippers G. Feasibility of teaching motivational interviewing to parents of young adults with recent-onset schizophrenia and co-occurring cannabis use, J Subst Abuse Treat 2014;46(3):340-345.
- 406 Söderlund LL, Madson MB, Rubak S, Nilsen P. <u>A</u> systematic review of motivational interviewing training for general health care practitioners. *Patient Educ Couns* 2011;84(1):16-26.
- 407 Decker SE, Martino S. <u>Unintended effects of training on clinicians' interest, confidence, and commitment in using motivational interviewing.</u> Drug Alcohol Depend 2013;132(3):681-687.
- 408 Britt E, Blampied NM. <u>Motivational interviewing training:</u> <u>a pilot study of the effects on practitioner and patient</u> <u>behaviour</u>, Behav Cogn Psychother 2010;38(2):239-244.
- 409 Duff AJ, Latchford GJ. <u>Motivational interviewing for</u> <u>adherence problems in cystic fibrosis; evaluation of</u> <u>training healthcare professionals.</u> J Clin Med Res 2013;5(6):475-480.
- 410 Wiley EJ, Irwin JD, Morrow D. <u>Health care practitioners'</u> perceptions of motivational interviewing training for facilitating behaviour change among patients. J Allied Health 2012;41(3):131-139.
- 411 Martino S, Haeseler F, Belitsky R, Pantalon M, Fortin AH 4th. <u>Teaching brief motivational interviewing to year</u> <u>three medical students.</u> *Med Educ* 2007;41(2):160-167.
- 412 Spollen JJ, Thrush CR, Mui DV, Woods MB, Tariq SG, Hicks E. <u>A randomized controlled trial of behavior</u> <u>change counseling education for medical students.</u> *Med Teach* 2010;32(4):e170-e177.
- 413 Daeppen JB, Fortini C, Bertholet N, Bonvin R, Berney A, Michaud PA, Layat C, Gaume J. <u>Training medical</u> <u>students to conduct motivational interviewing: a</u> <u>randomized controlled trial</u>. Patient Educ Couns 2012;87(3):313-318.
- 414 Baer JS, Wells EA, Rosengren DB, Hartzler B, Beadnell B, Dunn C. Agency context and tailored training in technology transfer: a pilot evaluation of motivational interviewing training for community counselors. J Subst Abuse Treat 2009;37(2):191-202.
- 415 Lozano P, McPhillips HA, Hartzler B, Robertson AS, Runkle C, Scholz KA, Stout JW, Kieckhefer GM. <u>Randomized trial of teaching brief motivational</u> <u>interviewing to pediatric trainees to promote healthy</u> <u>behaviors in families</u>. *Arch Pediatr Adolesc Med* 2010;164(6):561-566.
- 416 Haeseler F, Fortin AH 6th, Pfeiffer C, Walters C, Martino S. Assessment of a motivational interviewing curriculum for year 3 medical students using a standardized patient case. Patient Educ Couns 2011;84(1):27-30.
- 417 Childers JW, Bost JE, Kraemer KL, Cluss PA, Spagnoletti CL, Gonzaga AM, Arnold RM. <u>Giving residents tools to</u> talk about behavior change: a motivational interviewing <u>curriculum description and evaluation</u>. *Patient Educ Couns* 2012;89(2):281-287.
- 418 de Roten Y, Zimmermann G, Ortega D, Despland JN. <u>Meta-analysis of the effects of MI training on clinicians'</u> <u>behavior</u>. J Subst Abuse Treat 2013;45(2):155-162.

- 419 Pfister-Minogue KA, Salveson C. <u>Training and</u> experience of public health nurses in using behavior change counseling. Public Health Nurs 2010;27(6):544-551.
- 420 Rubak S, Sandbaek A, Lauritzen T, Borch-Johnsen K, Christensen B. <u>An education and training course in</u> motivational interviewing influence: GPs' professional behaviour - ADDITION Denmark. Br J Gen Pract 2006;56(527):429-436.
- 421 Bohman B, Forsberg L, Ghaderi A, Rasmussen F. <u>An</u> <u>evaluation of training in motivational interviewing for</u> <u>nurses in child health services</u>. *Behav Cogn Psychother* 2013;41(3):329-343.
- 422 Bean MK, Biskobing D, Francis GL, Wickham E 3rd. <u>Motivational interviewing in health care: results of a</u> <u>brief training in endocrinology.</u> J Grad Med Educ 2012;4(3):357-361.
- 423 Evangeli M, Longley M, Swartz L. <u>No reductions and some improvements in South African lay HIV/AIDS counsellors' motivational interviewing competence one year after brief training.</u> *AIDS Care* 2011;23(3):269-273.
- 424 Evangeli M, Engelbrecht SK, Swartz L, Turner K, Forsberg L, Soka N. <u>An evaluation of a brief motivational interviewing training course for HIV/AIDS counsellors in Western Cape Province, South Africa, AIDS Care 2009;21(2):189-196.</u>
- 425 Moser EM, Stagnaro-Green A. <u>Teaching behavior</u> change concepts and skills during the third-year medicine clerkship, Acad Med 2009;84(7):851-858.
- 426 Bell K, Cole BA. <u>Improving medical students' success in</u> promoting health behavior change: a curriculum evaluation. J Gen Intern Med 2008;23(9):1503-1506.
- 427 Pogge E. <u>A team-based learning course on nutrition and lifestyle modification</u>. *Am J Pharm Educ* 2013;77(5):103.
- 428 Triana AC, Olson MM, Trevino DB. <u>A new paradigm for</u> teaching behavior change: implications for residency training in family medicine and psychiatry. *BMC Med Educ* 2012;12:64.
- 429 Roman B, Borges N, Morrison AK. <u>Teaching motivational</u> <u>interviewing skills to third-year psychiatry clerkship</u> <u>students.</u> Acad Psychiatry 2011;35(1):51-53.
- 430 McLaughlin RJ, Fasser CE, Spence LR, Holcomb JD. <u>Development and implementation of a health</u> <u>behavioral counseling curriculum for physician assistant</u> <u>cancer education.</u> J Cancer Educ 2010;25(1):9-15.
- 431 Schoonheim-Klein M, Gresnigt C, van der Velden U. Influence of dental education in motivational interviewing on the efficacy of interventions for smoking cessation, Eur J Dent Educ 2013;17(1):e28-e33.
- 432 Hauer KE, Carney PA, Chang A, Satterfield J. <u>Behavior</u> <u>change counseling curricula for medical trainees: a</u> <u>systematic review.</u> Acad Med 2012;87(7):956-968.
- 433 Jansink R, Braspenning J, Laurant M, Keizer E, Elwyn G, Weijden Tv, Grol R. <u>Minimal improvement of nurses'</u> <u>motivational interviewing skills in routine diabetes care</u> <u>one year after training: a cluster randomized trial.</u> BMC Fam Pract 2013;14:44.
- 434 Cole B, Clark DC, Seale JP, Shellenberger S, Lyme A, Johnson JA, Chhabria A. <u>Reinventing the reel: an</u> <u>innovative approach to resident skill-building in</u> <u>motivational interviewing for brief intervention</u>. *Subst Abus* 2012;33(3):278-281.
- 435 Lupu AM, Stewart AL, O'Neil C. <u>Comparison of active-</u> learning strategies for motivational interviewing skills, knowledge, and confidence in first-year pharmacy students. Am J Pharm Educ 2012;76(2):28.

- 436 Poirier MK, Clark MM, Cerhan JH, Pruthi S, Geda YE, Dale LC. <u>Teaching motivational interviewing to first-year</u> <u>medical students to improve counseling skills in health</u> <u>behavior change</u>, *Mayo Clin Proc* 2004;79(3):327-331.
- 437 Lupu AM, Stewart AL, O'Neil C. <u>Comparison of active-learning strategies for motivational interviewing skills.</u> <u>knowledge, and confidence in first-year pharmacy</u> <u>students.</u> *Am J Pharm Educ* 2012;76(2):28.
- 438 Mounsey AL, Bovbjerg V, White L, Gazewood J. <u>Do</u> <u>students develop better motivational interviewing skills</u> <u>through role-play with standardised patients or with</u> <u>student colleagues?</u> *Med Educ* 2006;40(8):775-780.
- 439 Lane C, Hood K, Rollnick S. <u>Teaching motivational</u> <u>interviewing: using role play is as effective as using</u> <u>simulated patients.</u> *Med Educ* 2008;42(6):637-644.
- 440 Noordman J, van der Weijden T, van Dulmen S. Effects of video-feedback on the communication, clinical competence and motivational interviewing skills of practice nurses: a pre-test posttest control group study. J Adv Nurs (Published online March 2014).
- 441 Knight KM, Parr M, Walker D, Shalhoub J. <u>Web-based</u> <u>training package for HEEADSSS assessment and</u> <u>motivational interviewing techniques: a multi-</u> <u>professional evaluation survey.</u> *Med Teach* 2010;32(9):790.
- 442 Shafer MS, Rhode R, Chong J. <u>Using distance education</u> to promote the transfer of motivational interviewing skills among behavioral health professionals. J Subst Abuse Treat 2004;26(2):141-148.
- 443 Goodie JL, Williams PM, Kurzweil D, Marcellas KB. <u>Can</u> <u>blended classroom and distributed learning approaches</u> <u>be used to teach medical students how to initiate</u> <u>behavior change counseling during a clinical clerkship?</u> J <u>Clin Psychol Med Settings 2011;18(4):353-360.</u>
- 444 Martino S, Canning-Ball M, Carroll KM, Rounsaville BJ. <u>A</u> criterion-based stepwise approach for training counselors in motivational interviewing. J Subst Abuse Treat 2011;40(4):357-365.
- 445 Cucciare MA, Ketroser N, Wilbourne P, Midboe AM, Cronkite R, Berg-Smith SM, Chardos J. <u>Teaching</u> motivational interviewing to primary care staff in the <u>Veterans Health Administration.</u> J Gen Intern Med 2012;27(8):953-961.
- 446 Mitchell S, Heyden R, Heyden N, Schroy P, Andrew S, Sadikova E, Wiecha J. <u>A pilot study of motivational</u> <u>interviewing training in a virtual world.</u> J Med Internet Res 2011;13(3):e77.
- 447 Carpenter KM, Cheng WY, Smith JL, Brooks AC, Amrhein PC, Wain RM, Nunes EV. <u>"Old dogs" and new skills: how clinician characteristics relate to motivational interviewing skills before, during, and after training. J Consult Clin Psychol 2012;80(4):560-573.</u>
- 448 Doran N, Hohman M, Koutsenok I. Linking basic and advanced motivational interviewing training outcomes for juvenile correctional staff in California. J Psychoactive Drugs 2011;Suppl 7:19-26.
- 449 Mitcheson L, Bhavsar K, McCambridge J. <u>Randomized</u> <u>trial of training and supervision in motivational</u> <u>interviewing with adolescent drug treatment</u> <u>practitioners</u>, J Subst Abuse Treat 2009;37(1):73-78.
- 450 Nesbitt BJ, Murray DA, Mensink AR. <u>Teaching</u> motivational interviewing to nurse practitioner students: <u>a pilot study</u>. J Am Assoc Nurse Pract 2014;26(3):131-135.
- 451 Forsberg L, Forsberg LG, Lindqvist H, Helgason AR. <u>Clinician acquisition and retention of motivational</u> <u>interviewing skills: a two-and-a-half-year exploratory</u> <u>study.</u> Subst Abuse Treat Prev Policy 2010;5:8.

- 452 Martino S, Ball SA, Nich C, Canning-Ball M, Rounsaville BJ, Carroll KM. <u>Teaching community program clinicians</u> <u>motivational interviewing using expert and train-the-</u> <u>trainer strategies</u>. *Addiction* 2011;106(2):428-441.
- 453 Miller WR, Yahne CE, Moyers TB, Martinez J, Pirritano M. <u>A randomized trial of methods to help clinicians learn</u> <u>motivational interviewing.</u> J Consult Clin Psychol 2004;72(6):1050-1062.
- 454 Schwalbe CS, Oh HY, Zweben A. <u>Sustaining motivational</u> interviewing: a meta-analysis of training studies. *Addiction* (Published online March 2014).
- 455 Smith JL, Carpenter KM, Amrhein PC, Brooks AC, Levin D, Schreiber EA, Travaglini LA, Hu MC, Nunes EV. <u>Training substance abuse clinicians in motivational</u> <u>interviewing using live supervision via teleconferencing</u>. *J Consult Clin Psychol* 2012;80(3):450-464.
- 456 Smith JL, Amrhein PC, Brooks AC, Carpenter KM, Levin D, Schreiber EA, Travaglini LA, Nunes EV. <u>Providing live</u> <u>supervision via teleconferencing improves acquisition of</u> <u>motivational interviewing skills after workshop</u> <u>attendance</u>. *Am J Drug Alcohol Abuse* 2007;33(1):163-168.
- 457 Bennett GA, Moore J, Vaughan T, Rouse L, Gibbins JA, Thomas P, James K, Gower P. <u>Strengthening</u> motivational interviewing skills following initial training: a randomised trial of workplace-based reflective practice, Addict Behav 2007;32(12):2963-2975.
- 458 Olmstead T, Carroll KM, Canning-Ball M, Martino S. <u>Cost</u> and cost-effectiveness of three strategies for training clinicians in motivational interviewing. *Drug Alcohol Depend* 2011;116(1-3):195-202.